β-GLCANASES, ESSENTIAL KEY ENZYMES FOR THE UTILIZATION OF RENEWABLE RAW MATERIALS

Dissertation submitted for the award of the title of

Doctor rerum naturalium (Dr. rer. nat.)

Fermentation Engineering Group

Faculty of Technology

Bielefeld University

January 2017

Yingfei Shi

Born 02. September 1985 (Taizhou, China)

Acknowledgement

After several turns of white snow and transient water splashing in Bielefeld, now is the time to express my sincere gratitude to dear you, who have supported and helped me so much throughout this period.

I would like to thank Prof. Dr. Erwin Flaschel for accepting my application to join this warm group. By this step from China to Germany, I experienced, gained, and enjoyed a lot. Thanks for your support and help.

My sincere thanks also go to Prof. Dr. Karl Friehs and Dr. Joe Max Risse, who supported me greatly and were always willing to help me. Your doors were always open whenever I ran into a trouble in research or had a personal issue.

I would also like to thank Daniela, Galina, Kirsten, Thomas, Ebson, Usama for your kind support during this period. Your professional skills made my work in the lab much easier, and more importantly your cordial greetings in the mornings gave me warmness like home. Of course, my dear Dominik, Philipp, Ram, Maurice, Gaby, Jan, Lars and Julia, Franzi, Tobias, and my office mate Jakob, I would like to show my heartfelt thanks to all of you not only for your help on the work, but also for all friendly conversations and funny jokes.

My family members at Taizhou and Ningxia, my parents and parents in law, thanks for your support and encouragements throughout these years.

I would also like to thank my closest friend, my beloved wife Yihua for every drop of happiness you brought to my life. I hope I could be the pride of you and our unborn child.

人之垫	也, 勿	故	顺势而	居	上善去
, 如 转	(转木石	、善战者	为。	善地,	水。
圆石于	。 木 石	求 之		心 善 渊	
千仞之	之 性 , 、	于势,		, 与 善	
山者,	安 则 静	不责于		仁,言	
势也。	, 危 则	人故能		善行,	
	动, 方	择人而		止善治	
	则 止,	任势。		, 事 善	
	圆则行。	任势者		能, 动	
	故善战	, 其 战 人		善 时 。	

Abstract	1
Abbreviations	3
1. Introduction	5
1.1 Euglena gracilis	5
1.2 Paramylon	6
1.3 The third-generation bioethanol	8
1.4 Enzymes involved in paramylon degradation	9
1.4.1 β-1,3-Glucanases	9
1.4.2 Lytic polysaccharide monooxygenases (LPMOs)	.16
1.5 Pichia pastoris expression system	.20
1.5.1 A brief history of <i>P. pastoris</i>	.21
1.5.2 Methanol metabolism in <i>P. pastoris</i>	.23
1.5.3 Methanol utilization phenotype in <i>P. pastoris</i>	.25
1.5.4 Promoter in <i>P. pastoris</i>	.26
1.5.5 Selection marker in <i>P. pastoris</i>	.27
1.5.6 Gene integration in <i>P. pastoris</i>	.28
1.5.7 Protein secretion in <i>P. pastoris</i>	.30
1.5.8 Glycosylation in <i>P. pastoris</i>	.32
1.5.9 Conclusion of <i>P. pastoris</i> expression system	.32
1.5.10 β-1,3-glucanases expressed in <i>P. pastoris</i>	.33
2. Materials and Methods	.35
2.1 Basic nucleic acid manipulation	.35
2.1.1 Isolation and cleanup of total ribonucleic acid (RNA)	.35
2.1.2 First strand complementary deoxyribonucleic acid (cDNA) synthe	sis
	.36
2.1.3 Plasmid isolation	. 38
2.1.4 Restriction endonuclease digestion	.38
2.1.5 Agarose gel electrophoresis	.38
2.1.6 Purification and ligation of DNA fragments	.39
2.1.7 Isolation of genomic DNA	.39

2.1.8 Polymerase chain reaction (PCR)	40
2.1.9 Transformation of <i>E. coli</i>	42
2.1.10 Transformation of <i>P. pastoris</i>	43
2.2 Basic protein manipulation	45
2.2.1 Protein concentration assay	45
2.2.2 Sodium dodecyl sulfate polyacrylamide gel electrophoresis	47
2.2.3 SDS-PAGE gel staining	50
2.3 Activity assay and partial purification of <i>E. gracilis</i> intracellular pro	oteins
	51
2.3.1 <i>E. gracilis</i> cultivation	51
2.3.2 Intracellular protein extraction preparation	
2.3.3 Determination of dry biomass and dry paramylon	
2.3.4 Paramylon preparation for enzymatic hydrolysis	
2.3.5 Investigation of intracellular β-1,3-glucanase activity of <i>E. gra</i> cultures	<i>acilis</i> 54
2.3.6 Partial purification of intracellular proteins from E. gracilis	57
2.3.7 Trial of gene amplification of paramylon degrading enzyme in gracilis	<i>۱ E.</i> 59
2.4 Cloning and expression of recombinant β -1,3-glucanase in <i>P. pa</i>	storis63
2.4.1 Gene cloning of recombinant β-1,3-glucanase	63
2.4.2 Expression vector construction	67
2.4.3 Transformation of recombinant <i>P. pastoris</i>	68
2.4.4 Recombinant protein expression in <i>P. pastoris</i>	69
2.4.5 β-1,3-glucanase enzyme activity assay	71
2.5 Cloning and expression of recombinant lytic polysaccharide monooxygenase in <i>P. pastoris</i>	76
2.5.1 Strain and cultivation	76
2.5.3 Synergistic action assay between LPMOs and β -1,3-glucana	se78
3 Results and Discussion	81
3.1 Investigation of β -1,3-glucanase(s) from intracellular <i>E. gracilis</i> p	roteins
3.1.1 Influence of substrate	

3.1.2 Influence of intracellular protein extract concentration	83
3.1.3 Influence of extraction buffer pH	85
3.1.4 Influence of hydrolysis temperature	86
3.1.5 Influence of incubation duration	88
3.1.6 Influence of cultivation duration	89
3.1.7 Partial purification of <i>E. gracilis</i> intracellular proteins	91
3.1.8 Trial of gene amplification of paramylon degrading enzym from intracellular <i>E. gracilis</i> proteins	e purified 96
3.2 Cloning and expression of recombinant β -1,3-glucanase in <i>P</i> .	pastoris97
3.2.1 Selection of β-1,3-glucanase	97
3.2.2 Cloning and expression of <i>Tr</i> GH16 (EGR45305.1)	100
3.2.3 Cloning and expression of <i>Tr</i> GH55 (EGR47521.1)	108
3.2.4 Cloning and expression of <i>Tr</i> GH64 (EGR45483.1)	116
2.3.5 Cloning and expression of <i>Tr</i> GH81 (EGR49603.1)	123
2.3.6 Cloning and expression of <i>Tr</i> GH17 (EGR46171.1)	131
2.3.7 Cloning and expression of <i>Pp</i> GH5 (CAY69081.1)	139
3.3 Cloning and expression of recombinant lytic polysaccharide	
monooxygenase in <i>P. pastoris</i>	146
3.3.1 Extracellular hydrolytic enzyme assay	146
3.3.2 Cloning and expression of <i>Tr</i> AA9 (EGR52697.1)	148
3.3.3 Cloning and expression of <i>Ao</i> AA11 (BAE61530.1)	152
3.3.4 Synergistic Action of LPMO and β -1,3-Glucanase	155
4. Conclusion and Outlook	161
References	165
Appendices	183
Declaration	188

Abstract

During the last decade, the shifting from petroleum based fuel to a greener bio based one was expedited by the increasing concern of global warming. The growing demand oriented firstly to the use of food crops as raw material for producing so called first-generation biofuel. And then converted to the secondgeneration biofuel, using non-food biomass resources for instance, lignocellulosic raw materials. Although these generations of biofuel offered CO₂ emission benefits and improved domestic energy security, they also caused several environmental impacts. Such as high requirement for pesticide and fertilizer and conversion of agricultural land for food to energy crops (Naik et al. 2010; Chaturvedi and Verma 2013). Consequently, a new raw material, algae, which store high amounts of energy in form of carbohydrates contributed to eliminate these drawbacks. The algal biomass based fuel production was known as the third-generation biofuel (Lee and Lavoie 2013).

In general, algal storage polysaccharides are composed of glucose subunits. Paramylon from *Euglena gracilis*, a linear β -1,3-glucan with very high level of polymerization contributes up to 85% of cell dry weight (Sonck et al. 2010). This glucan is supposed to be a superb potential raw material for sustainable production of bioethanol. The production of bioethanol from algal polysaccharides requires efficient hydrolysis in order to generate fermentable monosaccharides. The hydrolysis is generally carried out by chemical or enzymatic methods. The chemical hydrolysis efficiently yields high concentrations of fermentable sugars, however requires environment harmful chemicals and generates byproducts inhibiting fermentation process. While the enzymatic hydrolysis processes under mild conditions without accumulating inhibitory byproducts (Chen et al. 2013).

1

Hence, searching for enzymes which could convert paramylon to glucose became the fundamental purpose of this work. As the first part, *E.gracilis* intracellular proteins were extracted for investigating the hydrolyzing ability on paramylon. The cell free extract prepared from late stationary phase rapidly hydrolyzes alkali treated paramylon. After partial purification, one protein potentially belonging to glycoside hydrolase family 22 was found, and followed with trail of obtaining the gene sequence from *E.gracilis* cDNA pool by degenerate PCR.

In the second part of this work, recombinant enzymes of four endo- β -1,3-glucanases (*Tr*GH16, *Tr*GH55, *Tr*GH64, *Tr*GH81), one exo- β -1,3-glucanase (*Tr*GH17) from *Trichoderma reesei* and one exo- β -1,3-glucanase (*Pp*GH5) from *Pichia pastoris* were successfully expressed in *P.pastoris* GS115. Moreover, their activities towards alkali treated paramylon were confirmed by measuring hydrolysis products of reducing groups and glucose, respectively.

In these two parts, alkali was involved for pretreating paramylon, which was not fully compatible with the purpose of direct hydrolysis of paramylon granule by enzymes without chemical pretreatment. However, no enzymatic method to deconstruct the highly crystallized paramylon granule has been reported by now. The newly discovered lytic polysaccharide monooxygenases (LPMOs), which could break glycosidic linkages of recalcitrant polysaccharides by oxidation and introduce new chain ends for hydrolytic enzymes (Forsberg et al. 2016), might become the key role to overcome the barrier. In the last part of this work, two LPMOs, *Tr*AA9 from *T.reesei* and *Ao*AA11 from *Aspergillus oryzae* were expressed in *P.pastoris*. The synergistic action assay between LPMO and glucanase revealed the possibility that *Tr*AA9 from *T.reesei* enhanced oligosaccharide accumulation in enzymatic hydrolysate. The result may open up a new way for depolymerization of recalcitrant paramylon granule in further research.

2

Abbreviations

CAZy	Carbonhydrate Active Enzyme
GH	Glycoside Hydrolase
AA	Auxiliary Activity
LPMO	Lytic Polysaccharide Monooxygenase
IUBMB	International Union of Biochemistry and Molecular Biology
EC	Enzyme Commission
CDH	Cellobiose Dehydrogenase
pl	Isoelectric Point
SEC	Size Exclusion Chromatography
IEC	Ion Exchange Chromatography
HIC	Hydrophobic Interaction Chromatography
AC	Affinity Chromatography
AOX	Alcohol Oxidase
Mut	Methanol Utilization
HIS4	Histidinol Dehydrogenase Gene
TT	Transcription Termination Region
ER	Endoplasmic Reticulum
Kex	Kexin Protease
α-MF	α-Mating Factor
DNA	Deoxyribonucleic Acid
RNA	Ribonucleic Acid
RNase	Ribonuclease

DNase	Deoxyribonuclease
cDNA	complementary Deoxyribonucleic Acid
PCR	Polymerase Chain Reaction
dNTPs	Deoxynucleotide Triphosphates
OD	Optical Density
rpm	Rounds per Minute
kV	Kilo Volt, Electric Potential
μF	Micro Farad, Electrical Capacitance
Ω	Ohm, Electrical Resistance
bp	Basepare
Kb	Kilo base
kDa	Kilo Dalton

1. Introduction

1.1 Euglena gracilis

Euglena was named by Christian G. Ehrenberg in 1830 (Ehrenberg 1830). The word "Euglena" was formed by "eu" and "glene" which mean "good" and "eyeball" in greek, respectively. The "eyeball" was from the clearly visible stigma with optical microscope (Leedale 1967). In 1900, axenically cultivating of *E.gracilis* was performed by Zumstein for the first time (Zumstein 1900). And most of present studies still use Zumstein's original strain "Z" (Evangelista et al. 2003).

A schematic drawing of Euglena is showed in Fig.1. It lacks a cell wall. Instead, it has a pellicle made up of a protein layer supported by a substructure of microtubules, arranging in strips spiraling around the cell. The action of these pellicle strips sliding over one another gives Euglena its exceptional flexibility and contractility. The highly flexible cell surface allowes them to change shape from thin cells to spheres with lengths between 20 to 100 µm (Schaechter 2012). The motorial system of *E.gracilis* consistes of two flagella, one emerges from the reservoir and the other one is termed non-emergent flagellum. The stigma, which consists of pigment granules, locates beside the reservoir as part of the visual system. Next to the stigma is the contractile vacuole for regulatory functions concerning the cell fluid. Inside the cell fluid, near the center of the cell, the nucleus can be found. The chloroplasts of *E. gracilis* account for a great percentage of cell organelles and for functions concerning the main energy providing facility. The cornered shape of chloroplast is a characteristic to E. gracilis. Due to the presence of chloroplasts, sometimes E. gracilis is placed with green algae and can be feed by photosynthesis or heterotrophy. Other

entities such as paramylon granules can be found spread throughout the cell (Marin et al. 2003).



Figure 1. A schematic drawing of Euglena (Eugene, 2015).

1.2 Paramylon

E. gracilis has been extensively researched for production of vitamins A, C, E (Takeyama et al. 1997), highly nutritious proteins (Schwarz et al. 1995), also as a good source of polyunsaturated fatty acids (Barsanti et al. 2000) and β -glucan (Russo et al. 2016). When grown in the presence of adequate carbon sources under heterotrophic growth conditions, *E. gracilis* can accumulate large amount of insoluble β -1,3-glucan, paramylon, which was firstly discovered by Gottlieb in 1850 (Gottlieb 1850; Bümer et al. 2001). And in 1960s the β -1,3-glucosidic linkages were identified as the structure basis (Clarke and Stone 1960). The structural investigations by X-ray diffraction and NMR spectrum revealed that it is a linear (unbranched, as showing in Fig.2) β -1,3-glucan type polysaccharide with a very high level of crystallinity about 90% in the native state (Kreger and Meeuse 1952; Kiss et al. 1988).



Fig.2 Schematic representation primary structure of paramylon (β -1,3-glucan) (Ramesh and Tharanathan 2003).



Fig.3 Schematic representation of microfibrils of a paramylon granule (Marchessault and Deslandes 1979).



Fig.4 Electron micrograph of a freeze-etched paramylon granule (Holt and Stern 1970).

Paramylon has high molecular mass estimated to be over 500 kDa and a high degree of polymerization about 3000 in form of membrane-bound granule, contributing up to 85% of cell dry weight (Barsanti et al. 2001; Sonck et al. 2010). Microfibrils which are composed of triple helices of β -1,3-glucan chains (as showed in Fig.3) traverse the inner granule and forms the unique high crystallinity (Kiss et al. 1987). The electron micrograph (Fig.4) revealed the concentric layers of the inner structure. In Fig.4, the segmentation of the inner material and outer layer are indicated by the arrows (Holt and Stern 1970).

Paramylon, as well as lentinan and pachyman, belong to a group of naturally occurring bioactive polysaccharides, and possesses antitumor and cytokinerelated immunopotentiating activity (Kondo et al. 1992). Sulfated derivatives of paramylon significantly inhibit the cytopathic effect of human immunodeficiency virus (HIV-1, HIV-2) and the expression of HIV antigen in human peripheral blood mononuclear cells (Koizumi et al. 1993). When incorporated in human or animal diets, paramylon supports cholesterol lowering and moderates the postprandial blood glucose and insulin response (Wang et al. 1997).

1.3 The third-generation bioethanol

For mitigating climate change, the demand of decreasing fossil energy resources and explorations for fossil energy is boosting in worldwide. The strategy of substitute fossil by biomass is the key strategy to achieve this goal, and predominate the next generation biofuels (Aditiya et al. 2016). By now, the production of fuels from biomass is categorized as three generations: the firstgeneration, which uses good crops as raw material; the second-generation converting also lignocellulosic material into biofuel; and the third-generation, which utilizes algae biomass (Abdallah et al. 2016). In algae, carbohydrates are stored in form of storage sugars and structural material. Beyond varying in the glycosidic bond between monomers, algae energy storage polysaccharides are composed of glucose subunits (Kim 2015). The process of bioethanol production from algae polysaccharides consists of three major steps: biomass pretreatment, enzymatic hydrolysis of algae polysaccharides, and fermentation of sugar monomers to ethanol. The pretreatment step disrupts algal cell and releases intracellular sugars. Subsequently, the pretreated biomass is degraded by lytic enzymes into simple sugars for fermentation. In the last step, simple sugars are used for the microorganism's growth, while ethanol is produced as a byproduct of the fermentation process (Harun et al. 2014). Studies proved the suitability of algae as feedstock for bioethanol years ago. Microalgae, such as Chlorococcum infusionum, Chlamydomonas reinhardtii, and Chlorella vulgaris, were exploited for the third-generation bioethanol production (Daroch et al. 2013). Mircoalgae were estimated being able to produce 5000-15,000 gal of ethanol per acre annually, which was more reliable than the first- and second-generation bioethanol feedstock (Nguyen and Vu 2012). Meanwhile, macroalgae (seaweeds) with a world average yield of 73 kg/m² per year, were also reported to be one of the best candidates employed as the bioethanol feedstock in 2014 (Noraini et al. 2014). Paramylon, as the storage polysaccharide of Euglena, was reported contributing up to 85% of cell dry weight (Sonck et al. 2010), making it an outstanding candidate of renewable resource for the third-generation bioethanol production. However, the Paramylon granule was extremely resistant to physical and chemical attacks. In vitro, hydrolysis of native granules hasn't been performed successfully. The structure of paramylon granule could be destroyed by treatment with alkaline, but without being hydrolyzed. It was believed that the treatment conducts not only a removal of the membrane, but also a restructuring as exposing enzymatic action sites (Monfils et al. 2011). Until now, limited scientific study investigated this type of recalcitrant raw material (Abdallah et al. 2016). And no bioethanol produced from paramylon has been reported to our knowledge.

1.4 Enzymes involved in paramylon degradation

In nature, carbohydrates are degraded through enzymatic cleavage of the glycosidic bonds between monomers. The diversity of polysaccharides and their complex structures results in the existence of huge diversity of enzymes for degradation of these biomolecules. These enzymes are mainly glycoside hydrolases, but can also involve oxidase, such as lytic polysaccharide monooxygenases (LPMOs) (Martinez 2016).

1.4.1 β-1,3-Glucanases

Enzymatic degradation of β -glucans involves a variety of enzymes that hydrolyze glycosidic bonds naming β -glucanases. The enzymatic hydrolysis take place via general acid catalysis requiring a proton donor and a

9

nucleophile/base. Two mechanisms are involved in the hydrolysis resulting in either a retention or an inversion of the anomeric configuration (See Fig.5) (McCarter and Withers 1994). In retaining mechanism, the glycosidic oxygen is protonated by the acid catalyst (AH) and the base B⁻ conducts nucleophilic assistance to aglycon departure. Then the glycosyl enzyme interacts with a water molecule, generating a product with the same stereochemistry as the substrate at the anomeric carbon. In the inverting mechanism, protonation of the glycosidic oxygen and aglycon departure are accompanied by a concomitant attack of a water molecule which is activated by the base residue B⁻. This reaction results in a product having opposite stereochemistry to the substrate.



Fig.5 Schematic representation of two enzymatic glycosidic bond hydrolysis mechanisms. (a) The retaining mechanism; (b) The inverting mechanism (McCarter and Withers 1994).

In both mechanisms, positions of the proton donor are identical. But positions of the nucleophile relative to the sugar anomeric carbon are different. In inverting enzymes accommodating a water molecule between the base and the sugar are more distant than in retaining enzymes (Davies and Henrissat 1995).

β-glucanases are systematically classified by the International Union of Biochemistry and Molecular Biology (IUBMB) depending on the selectivity of glucosidic bond and the hydrolytic reaction patterns against specified substrates. The IUBMB sorts these enzymes by Enzyme Commission (EC) number 3.2.1.x. Degradation of β -1,3-glucan, such as paramylon, involves the actions of two groups of glucanases: exo-β-1,3-glucanase (E.C.3.2.1.58) and endo- β -1,3-glucanase (E.C.3.2.1.39). The exo- β -1,3-glucanases are expected to produce glucose by sequential hydrolysis of β-1,3-glucan from the nonreducing end (Yamamoto and Nevins 1983). Endo- β -1,3-glucanases cleave at random sites along the β -1,3-glucan chain, generating oligosaccharides (Pitson et al. 1993). Quite different to IUBMB, the Carbonhydrate Active Enzyme (CAZy) database provides a sequence based family classification of the enzymes which could assemble, modify, or breakdown oligo- and polysaccharides (Lombard et al. 2014). These carbonhydrate active enzymes are classified into different glycoside hydrolase (GH) families by amino acid sequence similarity, members of each family share a common ancestor, conserved catalytic residues and a similar tertiary structure. Each clan in IUBMB contains enzymes from several GH families. For instance, the exo- β -1,3-glucanase (E.C.3.2.1.58) is classified in families 3, 5, 17, and 55, while the endo- β -1,3-glucanase (E.C.3.2.1.39) exists in families 16, 17, 55, 64, 81, and 128 (Sakamoto et al. 2011).

A GH family 5 exo- β -1,3-glucanase from the human pathogen *Candida albicans* naming *Exg* is implicated in cell wall β -glucan remodelling through its glucosyl hydrolase and/or transglucosylase activities. The tertiary structure of *Exg* displays an expected irregular (β/α)₈-barrel structure and the distinctive "pocket" comprises eight conserved residues including Arg92, His135, Asn191, Glu192, His253, Tyr255, Glu292, Trp363. Structure of the "pocket" characterizes the active sited of GH family 5 glycoside hydrolases. A glucose from the non-reducing end of β -1,3-glucan chain is hydrolyzed through retaining

mechanism (Fig.5-a). The Glu292 acts as the nucleophile, and the Glu192 as the proton donor to conduct the so called double displacement by forming a covalent glycosyl-enzyme intermediate (See Fig.6). (Cutfield et al. 1999; Patrick et al. 2010).



Fig.6 Schematic representation of double displacement reaction.

Formation of the covalent glycosyl-enzyme intermediate is presumed to proceed through an oxo-carbenium ion-like transition state and involve nucleophile Glu292 and proton donor Glu192, which act on the glycosidic bond at the nonreducing end of a β -1,3-glucan chain (Cutfield et al. 1999).

In 2015, a high-resolution crystal structure of an exo- β -1,3-glucanase belonging to GH family 55, namely *SacteLam*55A, from *Streptomyces* sp. was presented (Bianchetti et al. 2015). The enzyme contains two right handed β helical domains in a single polypeptide chain. These two domains are separated by a long linker region but positioned side by side with a substrate binding cleft on the interface. The enzyme acts in exo-mode with inversion of anomeric configuration and produces glucose from nonreducing end of β -1,3glucan. The acidic and basic residues activate water as nucleophile to cleave glycosidic bond (As showed in Fig.5-b) (Vuong and Wilson 2010). In the predicted catalytic region, three residues Asp449, Glu480 and Glu502 are strictly conserved among the GH family 55 members. The Glu502 interacting with the anomeric oxygen as the catalytic acid. The residue Glu480 also has important role in catalysis acting as a general base in the catalytic mechanism by interaction with Gln174, Ser198, and the active site water to be the nucleophile. The conserved residue Asp499 has a primary role in positioning the substrate (Bianchetti et al. 2015).

Although the GH familiy 3 and 17 contain a number of exo- β -1,3-glucanases from different prokaryotes and eukaryotes, no crystallographic data is available for any of these enzymes.

Endo- β -1,3-glucanases are classified into six glycoside hydrolase families. To date, crystallographic studies have been reported for three of these families except GH family 55 and 128. Endo-β-1,3-glucanases in GH family 16 share a β -jelly roll fold and conduct the hydrolysis reaction in a retaining mechanism. At the catalytic region, two glutamate residues are highly conserved acting as nucleophile and proton donor to complete the double displacement action (See Fig.6) (Fibriansah et al. 2007). The crytal structure of a laminarinase (Lam, endo-β-1,3-glucanases) from Thermotaga maritima MSB8 was reported in 2011 (Jeng et al. 2011). The enyzme has the classical sandwich-like β -jelly roll fold forming by two antiparallel β -sheets against each other, and forms a concavity for locating the substrates. In the catalytic groove, Glu132 is the base/nucleophile to attack the C1 atom of the sugar ring to promote the cleavage of β-1,3-linkage in the substrate chain. Glu137 acts as the acid to receive the electron and transfer it to adjacent water molecule for attacking the same C1 atom in the β -position, releasing interaction between Glu132 base and the product to complete the retaining catalysis.

Endo- β -1,3-glucanases in GH family 17 possess the (β/α)⁸ barrel fold and two catalytic glutamate residues as proton donor and nucleophile respectively. The hydrolysis of the glycosidic bond in β -1,3-glucan chain is catalyzed by retention of the stereochemistry of anomeric carbon at cleavage point through a double displacement mechanism with covalent glycosyl-enzyme intermediate (Jenkins et al. 1995). For instance, in the endo- β -1,3-glucanases from *Solanum tuberosum*, Glu118 (proton donor) and Glu259 (nucleophile) are strictly conserved as catalytic residues (Wojtkowiak et al. 2012). Moreover, residues Tyr58, Tyr201, Phe204, Phe305 and Phe322 inside of the catalytic cleft are also strictly conserved among members of GH family 17, which may involve in stacking interactions with the substrate for positioning (Varghese et al. 1994).

Some endo- β -1,3-glucanases are members of GH family 64 cleaving a long chain polysaccharide β -1,3-glucan into specific pentasaccharide oligomer (Nishimura et al. 2001). In 2009, the structure, catalysis mechanism and essential residues of *LPHase* from *Streptomyces matensis* DIC-108 was reported (Wu et al. 2009). The structure of this enzyme consists of a barrel domain and a mixed (α/β) domain, forming a wide-open groove. Within the wide groove, several conserved residues including Glu154 and Asp170 act as acid and base catalysts, respectively, and residues Thr156, Asn158, Trp163 and Thr167 act as substrate binding residues. It cleaves the substrate chain from the reducing end through direct displacement mechanism yielding a pentaose product with the inverted anomeric configuration.



Fig.7 Schematic representation of the proposed catalysis of *LPHase*. When a long chain polysaccharide β -1,3-glucan diffuses into the wide groove. The substrate binding residues Thr156, Asn158 and Trp163 positioned the

reducing end at the +5 subsite, and then bound by Glu154 and Thr167. Hydrolysis of a β -glucosidic bond by Glu154 acting as a proton donor to the glycosidic oxygen and the Asp170 as a base-assisted nucleophilic from the opposite side of the sugar ring (Wu et al. 2009).

Members of GH family 81 are mainly β -1,3-glucanases from fungi, plant and bacteria. The first crystal structure of this family member was reported in 2013, which is *RmLam*81A from *Rhizomucor miehei* (Zhou et al. 2013). The enzyme has three distinct domains: domain A, B and C. The domain A is comprised of two eight-stranded antiparallel β -sheets, while domain C has a core of (α/α)₆barrel. These two domains compact with each other forming a long catalysis cleft. Domain B exists on the reverse side of the structure to stabilize the whole molecule. Two residues Glu553 and Glu557 are deduced to be a proton donor and a basic catalyst. This agrees with the inversion mechanism of the hydrolytic catalysis.

Enzymatic hydrolyzing paramylon by endo- and exo-1,3- β -glucanase was concerned decades ago. In 1969, Barras and Stone reported that pyramylon was enzymatically degraded by the action of endo- and exo-1,3- β -glucanases in *E.gracilis* cell-extract (Barras and Stone 1969). Vogel and Barber succeeded in characterizing enzymes degrading the treated paramylon in 1986. They found an enzyme act on the substrate yielding glucose as the sole product, and concluded the enzyme an exo-glucanase. This enzyme had optimum conditions of pH 5.0 in 0.5 mol/L acetate buffer and temperature of 60 °C. The highest enzyme activity appeared when the stationary phase *E. gracilis* cells were disruped by French press at about 207 bar. Additionally, an endo-glucanase was also characterized. The optimum pH was at pH 5.2. But in contrast to the exo-glucanase, the endo-glucanase lost its activity by heating to 55 °C. Another difference between the two glucanases was their solubility: the

15

exo-glucanase was mostly exited in the supernatant after centrifugation. But the endo-glucanase remained in the pellet of percentage about 15-39%. The pellet contained large amounts of insoluble paramylon, indicating that the endoglucanase was associated with the membrane or the surface of the paramylon granule (Vogel and Barber 1968).

1.4.2 Lytic polysaccharide monooxygenases (LPMOs)

For the derivation of biofuels from non-edible biomass in a more environment friendly transition and breaking the bottleneck of enzymatic conversion, studies are stimulated. Desires on enzymes capable of breaking down recalcitrant polysaccharides, such as paramylon in this work, are boosted. In the traditional model, degradation of such polysaccharides is performed by glycoside hydrolase cocktails, consisting of endo-acting enzymes that cut randomly in the polymer chain and exo-acting enzymes which degrade the substrate from free chain ends. Despite the well acceptance of this model, efficient enzymatic conversion of highly crystalline polysaccharide remains a major bottleneck. Difficulties for the glycoside hydrolases directly deconstruct crystalline substrate into fermentable sugars remains. An unidentified component which may overcome these difficulties was firstly speculated in 1950 (Reese et al. 1950). Researchers proposed that two components were involved in microbial cellulose degradation: One component broke the barrier of polysaccharide crystallinity and the second one corresponded to glycoside hydrolase. Early studies in 1974 firstly reported oxidation as an important enzymatic reaction in cellulose degradation (Eriksson et al. 1974). Decades have passed between the neglect and discovery of lytic polysaccharide monooxygenase (LPMO) activity against recalcitrant polysaccharide such as cellulose (Forsberg et al. 2011) and chitin (Vaaje et al. 2010). The discovery of the LPMOs spured introspection of the classical recalcitrant polysaccharide degradation by hydrolytic enzymes.

LPMOs catalyze the oxidative cleavage of glycosidic linkages of polysaccharides for introducing new chain ends in present of molecular oxygen and an external electron donor, which boosts the activities of hydrolytic enzymes (Forsberg et al. 2016). Chain cleavages are formed by oxidation at either C1 or C4 carbon of a sugar ring (Beeson et al. 2012; Isaksen et al. 2014). Enzymes with mixed C1/C4 activitiy have also been reported (Forsberg et al. 2014a). As showed in Fig.8, oxidation at C1 results in the formation of a lactone, then be further hydrated to a reducing-end aldonic acid, while oxidation at C4 formed a ketoaldose at the non-reducing end (Hemsworth et al. 2015). LPMOs can accept electrons from variety of donors, for instance cellobiose dehydrogenase (CDH) (Courtade et al. 2016), small molecule reducing agents such as ascorbate and gallic acid (Vaaje et al. 2010), and lignin derived redox mediators (Westereng et al. 2015).



Fig.8 Schematic representation of the LPMO reactions.

LPMOs catalyze oxidation within a polysaccharide chain leading to chain cleavage. Oxidation at C1 results in the formation of a lactone, which is

hydrated to become a reducing-end aldonic acid. C4 oxidation leads to the formation of a ketoaldose at the non-reducing end (Hemsworth et al. 2015).

Originally, LPMOs were classified mistakenly in two families in the carbohydrate active enzymes (CAZy) database: glycoside hydrolase family 61 (GH61), which mainly consisting of fungal enzymes having weak endoglucanase activity; and carbohydrage binding module family 33 (CBM33) in which were mostly bacterial proteins. In 2010, CBM33s were found deconstructing chitin in an oxidative mechanism (Vaaje et al. 2010), and subsequently GH61s were found able to boost the activity of glycoside hyrolase enzymes on lignocellulose (Harris et al. 2010b). In 2011, CBM33s (Forsberg et al. 2011) and GH61s (Westereng et al. 2011) were then demonstrated as oxidative enzymes and cooper was identified as the correct metal cofactor in the oxidation. These enzymes were first described as "lytic oxidases" (Mba et al. 2012), and then changed to the term "lytic polysaccharide monooxygenase" (LPMO) in 2012 (Horn et al. 2012), reflecting the ability to break and lossen polysaccharide chains from recalcitrant substrates.

Due to the potentially key role of breaking the bottle neck of enzymatic biomass conversion, there was a great interest in unraveling the ability of LPMOs to boost the conversion. Since the discovery of LPMOs, high diversity in sequence, modularity and substrate preference has been explored. Currently, LPMOs are classified in auxiliary activity (AA) families 9, 10, 11, and 13 in CAZy database (Levasseur et al. 2013). The fungal LPMOs belong to families AA9, AA11, and AA13, whereas bacterial LPMOs are sorted in family AA10. In the family of fungal LPMOs, 328 genes related to cellulose cleaving, 66 genes of chitin cleaving, 14 genes of starch cleaving, while 1840 genes of chitin and cellulose cleaving bacterial LPMOs are identified (Martinez 2016). The enzyme has been demonstrated acting on 1,4-glycosidic bonds for deconstructing

cellulose, chitin, starch, and hemicelluloses. In Table 1, some of the characterized LPMOs and their activities and substrate specificities are listed.

CAZy family	Enzyme	Organism	Substrate	Position	Reference
AA9	Cel61A	Trichoderma reesei	cellulose	N/D	(Harris et al. 2010)
	TaAA9	Thermoascus aurantiacus	cellulose	C1, C4	(Quinlan et al. 2011)
	<i>Pc</i> GH61D	Phanerochaete chrysosoporium	cellulose	C1	(Wu et al. 2013)
	<i>Nc</i> LPMO9D	Neurospora crassa	cellulose	C4	(Beeson et al. 2012)
AA10	<i>BI</i> AA10A	Bacillus licheniformis	chitin	C1	(Forsberg et al. 2014b)
	CjLPMO10B	Cellvibrio japonicus	cellulose	C1	(Forsberg et al. 2014a)
	<i>Ef</i> AA10A	Enterococcus faecalis	chitin	C1	(Gudmundsson et al. 2014)
	SgLPMO10F	Streptomyces gresius	chitin	C1	(Nakagawa et al. 2015)

	Table 1	Characterized	LPMOs
--	---------	---------------	-------

Introduction					
	<i>Tf</i> AA10B	Thermobifida fusca	cellulose	C1	(Forsberg et al. 2014a)
AA11	AoAA11	Aspergillus oryzae	chitin	C1	(Forsberg et al. 2014b)
AA13	AnAA13	Aspergillus nidulans	starch	C1	(Leggio et al. 2015)
	NcAA13	Neurospora crassa	starch	C1	(Van et al. 2014)

Since 2012, LPMOs have been involved in several commercial cellulose preparations, resulting in efficiently depolymerzation by combined action of LPMOs and glycoside hydrolases (Harris et al. 2014). Considering both the benefit of improving recalcitrant substrate utilization and lowering glycoside hydrolases dosage, the cocktails including LPMOs need to be formulated for every raw material. But by now, no activity has been demonstrated against β -1,3-glucan, such as paramylon. This situation stimulated the trial in this work of searching LPMOs which could act directly on paramylon granule and depolymerize it with synergy of β -1,3-glucanases.

1.5 Pichia pastoris expression system

Pichia pastoris is the most cost-effective eukaryotic protein expression system that results in high protein expression yield and achieves a high success rate for a variety of recombinant proteins. As a yeast, *P. pastoris* has many advantages of higher eukaryotic expression systems, such as protein processing, protein folding, and posttranslational modification, straightforward molecular and genetic manipulations and has the added advantage of secreting

the highly expressed protein into the medium (Ahmad et al. 2014). Therefore, the *P.pastoris* was used as recombinant enzyme expression system in this work. Basic knowledges, such as methanol metabolism, gene integration, and protein secretion of this system were introduced in following sections (1.5.1-1.5.10).

1.5.1 A brief history of P. pastoris

Methylotroph is the type of cultivation of organism grow on reduced one-carbon (C1) compounds, such as methanol, as sole source of carbon and energy. Methylotrophs are confined to a few prokaryote and eukaryote microorganisms. To prokaryotic methylotrophs, the C1 compound could be methanol, methylamine or methane, whereas in eukaryotic methylotrophs this is limited to methanol (Van et al. 2006).

In yeasts, only a limited number of species belonging to the genera *Pichia*, *Candida*, or *Torulopsis* are able to grow on methanol as sole carbon source, (Gellissen 2000). Among them, *P. pastoris* is well studied. It has unicellular oval shape, 1-5 μ m wide and 5-30 μ m long (as showing in Fig.9), with a typical eukaryotic cell wall structure consisting of 30-60% polycacharides (mainly β -glucan and mannan sugar polymers), 15-30% proteins, 5-20% lipids, and small amount of chitin (Klis 1994). It has broad pH and temperature range for living between pH of 3-7 (Cregg et al. 2000) and 25-35 °C (Patrick et al. 2005).



Fig.9 Scanning electron micrograph of Pichia pastoris (Schutter et al. 2009).

P. pastoris is taxonomically classified under the kingdom Fungi, division Eumycota, subdivision Ascomycotina, class Hemoascomycetes, order Endomycetales, family Saccharomycetaceae, and subfamily Saccharomycetoideae. Since 1940s, Herman Phaff isolated several yeasts from the Yosemite region of California and established a new *Pichia* species, *Pichia pastoris* (Phaff 1986). In 1969, the ability of yeast utilizing methanol as the sole carbon source was firstly described by Koichi Ogata (Koichi Ogata 1969). Immediate attention was attracted in the following decade, because of P. pastoris' potential ability of generating yeast biomass and protein for animal feed. In 1970s, mediums and protocols for *P. pastoris* growing on methanol at high cell densities were developed by Phillips Petroleum Company. In early 1980s, Phillips Petroleum contracted with Salk Institute Biotechnology/ Industrial Associate Inc.(SIBIA), developed the P. pastoris system as a heterologous protein expression system (Higgins and Cregg 1998). The enzyme alcohol oxidase I (AOX1) gene and its promoter were isolated (Roggenkamp et al. 1984; Roa and Blobel 1983), vectors, strains and methods for molecular genetic manipulation were also developed for P. pastoris expression system (Cregg et al. 1985). High levels of heterologous protein could be expressed by this new system, due to the combination of the strong and tightly regulated AOX1 promoter, and by optimized fermentation methods (Tschopp et al. 1987; Sreekrishna et al. 1988). In 1993, Phillips Petroleum sold the patent to the current holder Research Corporation Technologies (RCT) and licensed Invitrogen Corporation to sell the commercial products of this system which were widely used in molecular biology studies and industries until now (Cereghino and Cregg 2000).

Since 1984, the *P. pastoris* expression system has been selected for cloning and producing hundreds of heterologous proteins due to advantages over other microorganisms. Properties that make the *P. pastoris* an outstanding system were summarized as ability to reach high cell densities in uncomplex growth medium, easy manipulations in genetic terms, using strong and tightly regulated eukaryotic promoters such as alcohol oxidase (*AOX*) promoter and post translational modification performances as glycosylation, disulfide bond formation, proteolytic processing, secretion of foreign proteins by signal peptides into extracellular medium and therefore simplifying downstream processes (Cregg et al. 2000; Daly and Hearn 2005; Ahmad et al. 2014).

1.5.2 Methanol metabolism in *P. pastoris*

The general mode of methylotrophs to utilize carbon is converting C1 molecules into C3 compounds by a cyclic pathway. In *P. pastoris* and other methylotrophic yeasts, the C1 assimilate uniquely via the xylulose monophosphate cycle. The initial oxidation of methanol takes place in specialized microbodies, peroxisomes, and followed by subsequent metabolic steps in cytoplasm (Sibirny et al. 1990).



Fig.10 Schematic representation of methanol metabolism in *P.pastoris* (Van et al. 2006).

1: alcohol oxidase (*AOX*), 2: catalase (*CAT*), 3: formaldehyde dehydrogenase (*FLD*), 4: S-formylglutathione hydrolase (*FGH*), 5: formate dehydrogenase (*FDH*), 6: dihydroxyacetone synthase (*DHAS*), 7: dihydroxyacetone kinase, 8: formaldehyde reductase.

In peroxisome, oxidation of the sole carbon source methanol is catalyzed by the alcohol oxidase (*AOX*, EC 1.1.3.13), an enzyme belongs to the family of glucose methanol choline oxidoreductase. The oxidation of methanol by *AOX* results in the generation of hydrogen peroxide and formaldehyde. Enzyme catalase (*CAT*) decomposes hydrogen peroxide into water and oxygen (Ozimek et al. 2005). Formaldehyde is assimilated in the peroxisome or dissimilated in the cytosol, the proper partitioning over these two pathways is fully depending on whether sufficient xylulose-5-phosphate (Xu₅P) is present within the peroxisome (Yurimoto et al. 2005).

Dihydroxyacetone synthase (*DHAS*, EC 2.2.1.3) catalyzes the transketolase reaction between xylulose-5-phosphate (Xu₅P) and formaldehyde yielding dihydroxyacetone (DHA) and glyceraldehyde-3-phosphate (GAP) (Sakai et al. 1998). These C3 compounds are diffused to the cytosol. DHA is phosphorylated by dihydroxyacetone kinase forming dihydroxyacetone phosphate (DHAP). In the following aldolase reaction with GAP, it forms fructose-1,6-phosphate (FBP) which is converted to fructose-6-phosphate (F₆P) catalyzed by phosphatase. Xu₅P is regenerated by the subsequent pentose phosphate cycle and transferred into the peroxisome. One third of GAP is exploited for biomass formation via gluconeogenesis (Yurimoto et al. 2011).

Within the cytosolic dissimilation pathway, formaldehyde is trapped by the reduced form of glutathione (GSH) to generate S-hydroxymethyl glutathione (GS-CH₂OH) which is a suitable substrate for formaldehyde dehydrogenase (*FLD*, EC 1.2.1.1) yielding S-formylglutathione (GS-CHO) and NADH. GS-CHO is than hydrolyzed by S-formylglutathione hydrolase (*FGH*, EC 3.1.2.12) to formate and GSH. Formate oxidation is catalyzed by formate dehydrogenase (*FDH*, EC 1.2.1.2) yielding CO₂ and NADH. Dissimilation of formaldehyde results on one hand in the generation of NADH, on the other hand in production of ATP via mitochondrial oxidative phosphorylation. The enhancement of AP

will stimulate the formation of Xu₅P and import it into the peroxisome for assimilatory pathway (Van et al. 2006).

1.5.3 Methanol utilization phenotype in *P. pastoris*

Two alcohol oxidase genes participate the methanol metabolism of *P. pastoris* : AOX1 which was isolated in 1985 by Ellis *et al.* (Koshland 1953) for the first time and AOX2 which was characterized in 1989 by Cregg *et al.* (Cregg et al. 1989). The major alcohol oxidase activity detected in *P. pastoris* fermentation using methanol as carbon source is contributed by AOX1 (responsible for 85% of alcohol oxidase activity), which is more active than AOX2 (responsible for 15% of alcohol oxidase activity) (Cereghino and Cregg 2000). This activity difference is caused by the difference in nucleotide sequence located at 5' end of the protein coding portion, although the AOX2 is about 97% homologous to AOX1. The presence of methanol as the only carbon source in the environment can induce the AOX1 promoter, and be repressed when other carbon source exists, such as glucose or glycerol (Paulova et al. 2012).

Because of the presence of two alcohol oxidases in the system, *P. pastoris* strains can be divided into three phenotypes respecting to methanol utilizing abilities (Patrick et al. 2005).

- a) Methanol utilization plus (Mut⁺) phenotype: Both AOX genes are active and the wildtype cells have efficient growth on methanol (Jungo et al. 2006).
- b) Methanol utilization slow (Mut^s) phenotype: Only the weak AOX2 gene is active, thus cells grow slower using methanol as only carbon source comparing to wild type (Jungo et al. 2006), but can avoid oxygen limitations and difficulties associated with high cell density cultivation (Cos et al. 2006).

c) Methanol utilization minus (Mut⁻) phenotype: This type of *P.pastoris* are unable to utilize methanol for growth, as both *AOX*1 and *AOX*2 genes are defective.

The most commonly used expression host of *P. pastoris* is GS115, the wild type with both the *AOX*1 and *AOX*2 genes, and has a phenotype of Mut⁺ (Cregg et al. 1985). The frequently used Mut^s strain is KM71, in which the chromosomal *AOX*1 gene is largely replaced with the *S.cerevisiae ARG*4 gene, resulting in slow growth on methanol relying on the weak *AOX*2 gene for the *AOX* activity. The Mut⁻ type of *P.pastoris* host, for instance MC100-3, has deleted both *AOX* genes and is unable to grow on methanol (Martín-Cuadrado et al. 2008).

1.5.4 Promoter in *P. pastoris*

The promoter AOX1 (alcohol oxidase 1, P_{AOX1}) is the most used inducible promoter, which was firstly employed in 1987 for β -galactosidase expression (Tschopp et al. 1987). As reported, the P_{AOX1} is induced only when methanol is the only carbon source, and is strongly repressed by glucose, glycerol or ethanol until these carbon source be depleted (Inan and Meagher 2001). Based on the analysis of the AOX1 promoter sequence, a consensus sequence of core motif consisting 5'-CYCCNY-3' was found as specific binding sites for transcription factors (Kranthi et al. 2010). The zinc finger protein Mxr1p (methanol expression regulator 1) (Cereghino et al. 2006) and ROP (repressor of phosphoenolpyruvate carboxykinase) (Kumar and Rangarajan 2012) share this DNA binding specificity, but regulate the PAOX1 antagonistically in *P.pastoris*. In the promoter of AOX1, Mxr1p binds at least six regions with the core motif, and deletions of these regions resulting a significant activity reduction of the promoter (Kranthi et al. 2009). Contrarily, deletion of the gene encoding ROP results in increasing activity of PAOX1 in a nutrient rich medium (Kumar and Rangarajan 2012).

Although the *P. pastoris* with strong AOX promoter is an effective system for protein expression, there are disadvantages due to the usage of methanol. These can be summarized as not suitable for food and therapeutic products manufacturing, as well as the explosive and flammable danger when storing large quantities of methanol (Patrick et al. 2005). Taking these limitations into consideration, alternative promoters for heterologous protein expression in P. pastoris, either constitutive or inducible have been developed. For constitutive promoters, promoter GAP (Glyceraldehyde-3-P dehydrogenase) (Waterham et al. 1997), promoter TEF1 (Translation elongation factor 1) (Adamian et al. 2011), promoter PGK1 (3-Phosphoglycerate kinase) (Almeida et al. 2005), and promoter Glycosyl GCW14 (Phosphatidyl inositol anchored protein) (Liang et al. 2013) are often used. Beside promoter AOX1, the most used and recently established inducible promoters in *P. pastoris* for heterologous expression are: promoter DAS (Dihydroxyacetone synthase) (Ellis et al. 1985), promoter FLD1 (Formaldehyde dehydrogenase 1) (Shen et al. 1998) and promoter ADH1 (Alcohol dehydrogenase), promoter ENO1 (Enolase), and promoter GUT1 (Glycerol kinase) (Cregg 2012). Among the constitutive promoters, promoter GAP (Glyceraldehyde-3-phosphate) is the commonly used for constitutive protein expression (Waterham et al. 1997).

1.5.5 Selection marker in *P. pastoris*

The standard setup of selection marker used in *P. pastoris* are either auxotrophy markers or genes conferring resistance to antibiotics. For the former type, several auxotrophic strains together with vectors having the respective genes have been developed. For instance, *HIS*4 (Histidinol dehydrogenase gene) (Cregg et al. 1985), *ARG*4 (Aregininosuccinate lyase gene), *ADE*1 (PR-aminoimidazole succino carboxamide synthase gene), *URA*3 (Orotidine-5'-phosphate decarboxylase gene) (Creghino et al. 2001) and *MET*2 (Methyl transferase gene) (Thor et al. 2005). In the latter strategy, zeocin resistance

gene (Drocourt et al. 1990), blasticidin S deaminase gene (Kimura et al. 1994), and kanamycin resisitance gene (G418) (Scorer et al. 1994) are commonly used. In this work, as the *P. pastoris* GS115 is a well studied *HIS*4 auxotrophic mutant, a vector with wild type *HIS*4 gene was used for transformation and protein expression.

1.5.6 Gene integration in *P. pastoris*

Stable gene integration in *P. pastoris* is performed via homologous recombination between linearized vector and genome in homology regions. The integration can be conducted in two ways, gene insertion and gene replacement (Ahmad et al. 2014).

The gene insertion commonly events at the *AOX*1 locus or *HIS*4 locus. Single or multiple gene insertion occurs at both loci. The multiple insertion happens at a low frequency less than 10%. Gene insertion at the *AOX*1 locus conduct via a crossover mechanism at one of these regions in the vector: the *AOX*1 promoter, the *AOX*1 transcription termination region (TT), or 3'-end of *AOX*1 expression cassette. The Fig.11 shows the insertion of linearized recombinant vector at 3'-end *AOX*1 region, gaining an expression cassette of promoter *AOX*1, gene of interest, and *HIS*4.

Gene insertion at the *HIS*4 locus occurs by crossover event between the *HIS*4 locus in the *P. pastoris* chromosome and the *HIS*4 gene in the recombinant plasmid as shown in Fig.12, by linearizing the vector at a restriction enzyme site located in *HIS*4 gene.

28



Fig.11 Insertion of linearized plasmid to the AOX1 locus (Life technologies

2014).



Fig.12 Insertion of linearized plasmid to the *HIS*4 locus (Life technologies 2014).

Gene replacement conducts via double crossover event between the *AOX*1 promoter and 3' *AOX*1 terminal regions between the recombinant vector and *P. pastoris* genome, resulting in *AOX*1 gene deletion and replacement by the expression cassette containing promoter *AOX*1, gene of interest and selection marker. Because of the disruption of the *AOX*1 gene, these recombinant strains
generated are Mut^s phenotype and rely weaker enzyme *AOX*² for growth on methanol. In this work, all the recombinant vectors were integrated into *P. pastoris* strain GS115 by gene replacement at *AOX*1 locus. The Fig.13 below shows the gene replacement at the *AOX*1 locus.



Fig.13 Gene replacement at the *AOX*1 coding region (Life technologies 2014).

1.5.7 Protein secretion in P. pastoris

P. pastoris expression system eliminates the costs of expensive downstream purification, as the secreted heterologous protein constitutes the majority of total protein in the medium (Weidner et al. 2010). Protein secretory pathway in *P. pastoris* involves several key steps as showing in Fig.14 (Lambertz et al. 2014). The peptide with an endoplasmic reticulum (ER) secretion signal is synthesized by the ribosome, then targeted to the ER where N-glycosylation, folding and cleavage occurs. The peptide exits the ER and enters the golgi apparatus where their pro-region is removed by dibasic endo-peptidases, such as *Kex2*. Afterward, the recombinant mature protein is packed into a secretory vesicle that will bud from the golgi apparatus and fuse with the plasma membrane. It is then delivered to the periplasm, and transported into the extracellular medium via the cell wall afterwards. For secretion of the

recombinant protein, a secretion signal leader peptide is needed. The commonly used secretion signals in *P. pastoris* are α -mating factor (α -MF), and the invertase signal sequence (*SUC*2) from *S.cerevisiae*, or the *P. pastoris* endogenous acid phosphatase signal sequence (*PHO*1) (Daly and Hearn 2005). In this work, the secretion signal α -MF was involved for protein secretion (See Fig.15).



Fig.14 Schematic representation of protein secretion in *P. pastoris* (Lambertz et al. 2014). Arrows indicate the secretory pathway, from the ER (1) the protein is transferred via the golgi apparatus (2) to the medium (3) in secretory vesicles.



Fig.15 Primary structure of α -MF secretion signal leader.

The α -MF leader consists of two regions, a pre-sequence signal peptide of 19 amino acids and a pro-region of 67 amino acids. The pre-region is responsible for directing the nascent protein post translationally into the ER and is cleaved off subsequently by signal peptidase (Waters et al. 1988). The pro-

region is needed for transferring the protein from the ER to the golgi apparatus compartment and is cleaved at the Lysine-Arginine (K-R) site by the protease *Kex2* to release the mature protein.

1.5.8 Glycosylation in *P. pastoris*

Glycosylation is the most common post-translational modification to proteins. *P. pastoris* is capable of both *O*-linked and *N*-linked glycosylation of expressed proteins (Pourcq et al. 2010).

Eukaryotic cells assemble *O*-linked saccharide onto the hydroxyl groups of serine or threonine residues in proteins and mainly occur in the golgi apparatus. In yeast such as *P.pastoris*, the *O*-oligosaccharides usually comprise one to five mannose α -1,2 linked residues in arrangement (Duman et al. 1998).

In all eukaryotes, the initial step of *N*-linked glycosylation takes place at the cytoplasmic side of the ER, begins with the sequential addition of *N*-acetylglucosamine (GlcNAc) and mannose (Man) residues, and synthesizes a dolichol-linked glycan precursor. The further elongating occurs after the intermediate Man₅GlcNAc₂-PP-Dol structure is flipped to the lumen of the ER until Glc₃Man₉GlcNAc₂ is formed. The resultant oligosaccharide is transferred to asparagine residues of the nascent polypeptide chain trimming to Man₈GlcNAc₂. At this point, glycosylation pattern of yeast and higher eukaryote differs. In yeast, processing after the glycoconjugate is transported to the golgi apparatus starts with the addition of α -1,6-mannose residues. This branch is elongated by stepwise addition of mannose residues leading to a very complicated hypermannosylated glycoform mixture (Dean 1999).

1.5.9 Conclusion of *P. pastoris* expression system

The use of *P.pastoris* as host for recombinant protein expression has become increasingly popular in recent years. *P.pastoris* is easy to manipulate genetically. The linearized foreign DNA can be inserted in high efficiency via

homologous recombination to generate stable transformats. The benefit of the strong promoters drive promising high-level expression of target proteins. More importantly, *P. pastoris* is an eukaryote provides the potential of producing soluble, correctly folded recombinant proteins with post-translational modifications. A further benefit of *P. pastoris* is that the ability of growing in high cell density enabling large scale fermentations at lower cost than most other eukaryotic systems (Ahmad et al. 2014).

Although many basic elements of *P. pastoris* expression system are well studied and many industrial enzymes have been successfully expressed, further work could still be conducted for optimizing the system. Improving protein secretion by employing different secretion signals is one of the foremost goal in researches (Vadhana et al. 2013). Another general interest is to find effective alternatives to induce protein expression instead of toxic methanol for industrial scale fermentation (Delic et al. 2013). Moreover, by engineering the glycosylation pathways, *P. pastoris* may produce proper post-translational modifications of therapeutic proteins for better pharmacokinetic behaviors (Pourcq et al. 2010).

1.5.10 β-1,3-glucanases expressed in *P. pastoris*

To produce β -1,3-glucanase, the heterologous expression of the enzyme has been tried in various host organisms, including *Escherichia coli* (Kutty et al. 2015), *Bacillus subtilis* (Chuen and Lynch 1998), *Streptomyces lydicus* (Wu et al. 2015), *Saccharomyces cerevisiae* (Giczey et al. 2001), and higher plants (Zhang et al. 2015). The popular yeast *P.pastoris* was also used as expression host. For instance, endo- β -1,3-glucanases from *Lentinula edodes* (Sakamoto et al. 2011), *Aspergillus fumigatus* (Hartl et al. 2011), and exo- β -1,3-glucanases from *Phanerochaete chrysosporium* (Ishida et al. 2009), *Candida albicans* (Patrick et al. 2010) have been successfully expressed in *P. pastoris*. In 2015, the β -1,3-glucananse from thermophilic fungus *Chaetomium thermophilum* which belonging to GH family 55 was expressed in *P. pastoris* (Papageorgiou and Li 2015). Two exo- β -glucanases of GH family 3 from Saccharomycopsis fibuligera and Trichoderma reesei were also cloned and expressed in P. pastoris. These two enzymes both exhibit significant hydrolysis activity against substrate p-nitrophenyl-beta-d-glucopyranoside, but no evidence of glycosidc bond specificity was reported (Ma et al. 2015). An endo-β-1,3-glucanases of GH family 17 from Ash pollen was produced in P. pastoris. Meanwhile, the Nterminal catalytic domain of this enzyme was expressed as independent protein in *P. pastoris*. Although full-length enzyme, showed low enzymatic activity, the recombinant N-terminal catalytic domain had 200-fold higher activity on laminarin as substrate (Torres et al. 2015). As a common enzyme used in biomass conversion, it's believed that more work about β -1,3-glucanase expression in the *P. pastoris* system will be reported, such as optimizations of codon usage, protein structure, medium and cultivation conditions, fermentation engineering and so on.

2. Materials and Methods

2.1 Basic nucleic acid manipulation

2.1.1 Isolation and cleanup of total ribonucleic acid (RNA)

Total ribonucleic acid (RNA) isolation was performed using RNeasy Plant Mini Kit (Qiagen) according to the standard protocol.

RNA purity and integrity are essential for RNA-related experiments. Ribonuclease (RNase) is a highly stable contaminant in laboratory environment which degrades RNA. To prevent prepared samples from RNase degradation, following guidelines should be abided. A set of separate instruments such as pipettes, centrifuge, votexer and disposable nuclease-free plastic-wares are needed exclusively for work with RNA. For glasswares: incubate in 0.1% (v/v) diethylpyrocarbonate (DEPC, Sigma-Aldrich, USA) solution overnight at 37 °C following by autoclaving to ensure complete decomposition of DEPC to CO₂ and ethanol. The work surfaces are cleaned with RNaseZap solution (Thermo Scientific, USA) before use. And wear gloves and change frequently when handling RNA.

The isolated total RNA was cleaned up using deoxyribonuclease I (DNase I, Sigma-Aldrich, USA) and RNeasy MinElute Cleanup Kit (Thermo Scientific, USA). This work was conducted according to the protocol bellow.

1. Add the following reagents into an RNase-free PCR tube:

Table 2 Total RNA clean-up reaction mixture.

Reagent	Volume (µL)
Isolated total RNA	8

10× Reaction buffer	1
DNase I	1

- 2. Mix gently and incubate at room temperature for 15 minutes;
- Add 1 μL of stop solution to bind calcium and magnesium ions for inactivating DNase I;
- Incubate the mixture at 70 °C for 10 minutes to denature both the DNase I and RNA, then chill on ice;
- 5. Collect RNA according to the concentration protocol provided by the kit;
- 6. The concentration and purity of RNA isolated was determined by NanoDrop[®] spectrophotometer (Thermo Scientific, USA). Absorbance readings at 260 nm should be between 0.05 and 1.00. The ratio between the readings at 260 and 280 nm (A260/A280) provides an estimate of the purity of the RNA. Pure RNA preparations have A260/A280 values of 2.0. RNA samples displaying A260/A280 values in the range of 1.8 to 2.0 could be regarded as highly pure and samples with values below 1.0 should not be used;
- 7. The pure RNA prepared was used for first strand cDNA synthesis immediately.

2.1.2 First strand complementary deoxyribonucleic acid (cDNA) synthesis

The first strand complementary deoxyribonucleic acid (cDNA) synthesis was performed with RevertAid H Minus First Strand cDNA Synthesis Kit (Thermo Scientific, USA) using the total RNA as template. The synthesis reaction was procced on ice according to the following protocol. 1. Add the following reagents into a sterile, nuclease-free PCR tube and add nuclease-free water to the final volume of 12 μ L;

Component	Amount
Template RNA	2.5 µg
10 µmol/L Primers	1 µL
Nuclease-free water	Το 12 μL

Table 3 cDNA synthesis reaction mixture.

- Mix gently, centrifuge briefly and incubate at 65 °C for 5 min. Chill on ice, afterwards;
- 3. Add the following reagents to the PCR tube and incubate for 60 min at 42 °C;

Component	Volume (µL)
5× Reaction buffer	4
20 U/ µL RNase inhibitor	1
10 mM dNTP Mix	2
200 U/µL Reverse transcriptase	1

Table 3 Continued

4. Terminate the reaction by heating at 70 °C for 5 min. The reverse transcription was used for cloning work immediately or stored at -80 °C for further use.

2.1.3 Plasmid isolation

E. coli transformant was picked to inoculate 5 mL LB medium per well with respective antibiotic in 24 square-well plate (HJ-bioanalytik, Germany). The culture was allowed to grow overnight at 37 °C with shaking at 120 rpm, shaking orbital of 50 mm (Kuhner LS-X, Germany). Plasmid isolation was prepared from *E. coli* in small scale using innuPREP Plasmid Mini Kit (Analytikjena, Germany). The concentration and purity of isolated plasmid was determined by NanoDrop[®] spectrophotometer. The plasmid was used immediately or stored at -20 °C.

2.1.4 Restriction endonuclease digestion

All restriction endonucleases were FastDigest[™] enzymes (Thermo Scientific, USA). The reaction components were mixed according to the standard reaction setup provided by the manufacture and incubated for 2 hours in a 37 °C water bath.

2.1.5 Agarose gel electrophoresis

Agarose gel electrophoresis was used to separate, identify, and purify deoxyribonucleic acid (DNA) fragments. A 10 g/L gel was prepared by mixing agarose in 0.5×TAE buffer. 5 µL Roti [®]-Safe Gel Stain (Carl Roth, Germany) was added to every 100 mL gel solution and mix thoroughly before solidifying.

To run electrophoresis, the gel was submerged with $0.5 \times TAE$ buffer in a chamber (Mupid-One, Nippon Genetics, Japan). 5-15 µL DNA sample, which was previously mixed with 1/10 volumes of 10 × DNA loading buffer, as well as 8 µL 1 Kb DNA Ladder (Plasmid Factory, Germany) was loading into each well, respectively. The electrophoresis was carried out at 150 volts until dye markers have migrated an appropriate distance depending on the size of the DNA fragment of interest. An BioDocAnalyze Systeme (Analyticjena, Germany) was used for gel visualization.

2.1.6 Purification and ligation of DNA fragments

DNA fragments from reaction mixtures such as restriction endonuclease digestions, PCR *etc.*, were purified using Wizard [®] SV PCR Clean-up Kit (Promega, USA) following the manufacturer's instructions.

DNA fragment separated by agarose gel electrophoresis was excised from the gel and then extracted by Wizard [®] SV Gel Clean-up Kit (Promega, USA) according to the standard procedure.

T4 DNA Ligase LC (Thermo Scientific, USA) was used for DNA ligations. Inserting DNA and linearized vector were mixed in a molar ratio of 3:1. The reaction mixture was prepared following the standard sticky end ligation protocol provided by the manufacturer. And the reaction was performed by incubating in a water bath at 22 °C for 1 hour. Before transformation, the reaction mixture was further incubated at 65 °C for 10 min, then chilled on ice to inactivate T4 DNA ligase.

2.1.7 Isolation of genomic DNA

To prepare the *P. pastoris* genomic DNA, single colony appeared on YPD plate after incubating at 30 °C was selected. After cultivating overnight in 5 mL/well of YPD medium using 24 square-well plate (HJ-bioanalytik, Germany) at 30 °C with constant shaking (180 rpm, shaking orbital 20 mm, IKA-KS 4000ic control, Germany), yeast cells were harvested by centrifugation. Genomic DNA isolation was performed using MasterPure[™] Yeast DNA Purification Kit (Epicentre, USA) following the protocol provided.

To isolate genomic DNA from *Asperfillus oryzae*, a glycerol stock was used to prepare a seed culture by inoculating 5 mL mL/well YPD medium in 24 square-well plate and cultivated for 16 h at 30 °C with shaking frequency of 180 rpm, shaking orbital of 20 mm. The entire seed culture was transferred into 50 mL of the same growth medium in 250 mL baffled shake flask and cultivated for 24 h at the same conditions. A 10 mL portion was centrifuged for harvesting mycelium. The harvested pellet was washed twice with demineralized water before lyophilization. Then the freeze-dried mycelium pellet was pulverized into fine powder using mortar and pestle in the presence of liquid nitrogen. Genomic DNA isolation from *A. oryzae* was then performed using Wizard[®] Genomic DNA Purification Kit (Promega, USA) according to plant tissue protocol. The genomic DNA was used immediately or stored at -20 °C.

2.1.8 Polymerase chain reaction (PCR)

 PCR was performed for the specific amplification of DNA, using plasmid, genomic DNA, or cDNA as template. The reaction mixture was set up following the recommendation provided with the Phusion[®] High-Fidelity DNA Polymerase (New England Biolabs, USA);

Component	Volume (µL)	Final Concentration
5× Phusion HF Buffer	4.0	1×
10 mM dNTPs	0.4	200 µM each
Forward primer	1.0	0.5 µM
Reverse primer	1.0	0.5 µM
Template DNA	**	< 250 ng
*DMSO (optional)	(0.6 µL)	3% (v/v)
Phusion DNA Polymerase	0.2	0.02 U/µL

Table 4 PCR reaction mixture.

H ₂ O Add to 20.0	-
------------------------------	---

*: For amplification of difficult targets, such as GC-rich sequences, or genomic DNA, a final concentration of 3% (v/v) dimethyl sulfoxide (DMSO) was added.

**: Recommend doses of DNA template are showed in the following table.

Template DNA	Amount for 20 µL Reaction
Genomic DNA	20 ng – 100 ng
Plasmid	0.4 pg – 4 ng
cDNA	≤ 2 µL

Table 5 Recommend doses of DNA template.

2. The reaction was performed by Mastercycler[®] Pro (Eppendorf, Germany) and the thermocycling conditions below were abided.

step	Temperature (°C)	Duration	Cycles
Initial Denaturation	98	30 s	1
Denaturation	98	10 s	
Annealing	*	20 s	35
Extension	72	** 15-40 s/Kb	

Table 6 PCR thermocycling conditions.

	Materials and Methods		
Final Extension	72	10 min	1
Hold	4	×	1

*: The annealing temperature depends on primers used.

**: Extension time is dependent on amplicon length and complexity. The recommend extension durations were showed below.

Table 7 Recommend extension duration.		
Amplicon	Extension Duration (s / kb)	
Genomic DNA	30	
Plasmid	15	
cDNA	40	

2.1.9 Transformation of *E. coli*

Transformation of E. coli was carried out according to the protocol from Chung et al. (Chung et al. 1989). And the rapid E. coli transformation procedure was optimized as following.

- 1. Cultivate *E. coli* DH5α overnight at 37 °C with shaking (120 rpm, orbital 50 mm, Kuhner LS-X, Germany) in 50 mL LB medium in 250 mL shake flask;
- 2. Dilute the overnight culture by the ratio of 1:50 with LB medium and grow at the same condition until the cell density reach $OD_{600} = 0.25 \cdot 0.4$;

- Harvest the cells by centrifugation at 4 °C, and re-suspend the pellet by 1/10 volume ice-cold TSS buffer (Appendices, solutions and mediums) to prepare competent cells;
- Add 100 pg-10 ng of DNA to 100 μL competent cells. Gently mix the cells and DNA by flicking the tube and incubate on ice for 10 minutes;
- 5. Transfer the tube to room temperature and incubate for 10 minutes;
- 6. Re-incubate the tube on ice for another 10 minutes;
- Add 800 µL LB medium and incubate the cells at 37 °C with shaking (120 rpm, orbital 50 mm, Kuhner LS-X, Germany) for 1 hour;
- Spread 100 μL transformed *E. coli* on LB agar plate with 100 μL/mL ampicillin, and incubate overnight at 37 °C.

2.1.10 Transformation of *P. pastoris*

The method for *P. pastoris* transformation was developed from the high efficiency protocol reported by Wu and Letchworth (Wu and Letchworth 2004). The transformation was carried out by electroporation using Gene Pulser[®] and Pulse Controller[®] (Bio-Rad, USA). Procedure of competent cell preparation and transformation was optimized below.

- A single colony of *P. pastoris* strain GS115 was grown overnight in 50 mL YPD medium using 250 mL baffled shake flask at 30 °C with shaking (180 rpm, shaking orbital 50 mm, Kuhner LS-X, Germany);
- 20 µL of the seed culture was used to inoculate 100 mL YPD medium and grew at the same conditions until the cell density reached OD₆₀₀= 1.0-2.0;
- 3. The cells were harvested by centrifugation and re-suspend with the same volume of *P. pastoris* transformation pretreat buffer (0.1 mol/L

lithium acetate, 10.0 mmol/L dithiothreitol) at room temperature for 30 minutes;

- 4. The cells were washed with ice-cold 1.0 mol/L sorbitol solution thrice;
- The cells were re-suspended with 0.5 mL sorbitol solution (1.0 mol/L) to prepare competent cells;
- 100 μL competent cells were mixed with 1.5-2.0 μg DNA gently. Transfer to a 0.2 cm gap electroporation cuvette (Bio-Budget, Germany) and incubate on ice for 5 minutes;
- 7. The electroporating pulse was applied at 1.5 kV, 25 μ F, 200 Ω ;
- The cuvette was incubated on ice for 5 minutes, and the electroporated cells were mixed with 800 µL ice-cold 1.0 mol/L sorbitol solution afterwards;
- 100 μL aliquot was spread on RDB agar plate and incubate at 30 °C for 3-4 days until yeast colonies appear.

2.2 Basic protein manipulation

2.2.1 Protein concentration assay

The protein concentration assay was performed using Protein Assay Kit (Bio-Rad, USA) based on the Bradford method. Two procedures were developed for different ranges of measurement. The standard procedure was for samples with protein concentration in the range of 0.1 g/L to 0.8 g/L. The more sensitive microassay procedure was for samples in the range up to 80 mg/L.

2.2.1.1 Standard procedure for protein concentration assay

- Dye reagent was prepared by diluting one part dye reagent concentrate with four parts of demineralized water. Then filtered through a Whatman 1 filter (GE Healthcare, USA) to remove particulates;
- Bovine serum albumin (BSA) protein standard samples were prepared in following concentrates (0 g/L; 0.1 g/L; 0.2 g/L; 0.3 g/L; 0.5 g/L; 0.8 g/L);
- 10 μL of each standard or sample solution was pipetted into separate wells in a 96 well micro titer plate (BrandTech Scientific, Thermo Fisher Scientific, USA);
- 200 µL of diluted dye reagent was added to each well. The sample and reagent were mixed thoroughly using multi-channel pipet to dispense the reagent;
- The mixture was incubated at room temperature for at least 5 minutes.
 Samples should incubate at room temperature for less than 1 hour;
- Absorbance at 595 nm was measured by SpectraMax M Microplate Readers (Molecular Devices, USA);
- Standard curve for protein concentration from 0 to 0.8 g/l could be seen in Fig.16.



Fig.16 Standard curve for the standard protein concentration assay

(0-0.8 g/L).

Fitting formula: $y = 1.068x, R^2 = 0.9887$

2.2.1.2 Micorassay procedure for the more sensitive protein concentration

- Protein standard samples were prepared in following concentrates (0 mg/L; 10 mg/L; 20 mg/L; 30 mg/L; 50 mg/L; 80 mg/L;
- 160 μL of each standard or sample solution was pipetted into separate well in 96 well micro titer plate (BrandTech Scientific, Thermo Fisher Scientific, USA);
- 40 µL of dye reagent concentrate was added to each well. The sample and reagent were mixed thoroughly;
- The mixture was incubated at room temperature for at least 5 minutes.
 Samples should incubate at room temperature for less than 1 hour;

- Absorbance at 595 nm was measured by SpectraMax M Microplate Readers (Molecular Devices, USA);
- Standard curve for protein concentration from 0 to 80 mg/L could be seen in Fig.17.





(0-80 mg/L).

Fitting formula: y = 83.899x, $R^2 = 0.9901$

2.2.2 Sodium dodecyl sulfate polyacrylamide gel electrophoresis (SDS-PAGE)

SDS-PAGE was applied to separate and analysis proteins, which was carried out using 12.5% (w/w) resolving gel and 5.0% (w/w) stacking gel in tris glycine buffer. The electrophoresis was conducted following to the protocol below.

 Assemble the glass plates and prepare the 12.5% (w/w) resolving gel solution, add APS and TEMED together at last as described in Table 8 and mix carefully to avoid the formation of bubbles;

Component	Volume per 15 mL
ddH ₂ O	5.0 mL
30% (w/w) Acrylamide/bisacrylamide solution	6.25 mL
1.5 mol/L Tris-HCl buffer (pH=8.8)	3.75 mL
50 g/L SDS solution	300 µL
10 g/L APS solution	75 μL
TEMED	10 µL

Table 8 Resolving gel preparation.

- Pour the gel solution between the glass plates, leave free space for the stacking gel. And carefully cover the top of resolving gel with 100 % isopropanol. Discard the isopropanol until the resolving gel was polymerized;
- Prepare the stacking gel as described in Table 9 and pour into the glass plate sandwich. Insert combs and wait the gel to polymerize;

Component	Volume per 5 ml
ddH ₂ O	2.77 mL
30% (w/w) Acrylamide/Bisacrylamide solution	0.83 mL
0.5 mol/L Tris-HCl buffer (pH=6.8)	1.25 mL

Table 9 Stacking gel preparation.

50 g/L SDS solution	100 µL
10 g/L APS solution	50 µL
TEMED	6 µL

- Mix three parts of protein sample with one part of 4x sample buffer (Life Technologies, USA) before incubating in 98 °C for 15 minutes;
- Remove combs. Put the glass plate sandwich into electrophoresis tank and fill with 1x Tris-glycine-SDS buffer. Load 8 μL PageRuler® Prestained Protein Ladder (Life Technologies, USA) and samples respectively into wells;
- Set an appropriate voltage and current depending on the number of gels refer to Table 10. Increase power when the dye front reaches the resolving gel;

Table 10 Pe	ower management	of protein	electrophoresis.
-------------	-----------------	------------	------------------

Gel	1 gel	2 gels
Stacking gel	13 mA	25 mA
Resolving gel	25 mA	50 A

7. Stop the electrophoresis when protein marker reached appropriate positions and process gel staining.

2.2.3 SDS-PAGE gel staining

2.2.3.1 Coomassie brilliant blue staining

After the electrophoresis, wash the SDS-PAGE gel with water twice to remove running buffer left on the surfaces. Stain the gel with coomassie brilliant blue G-250 staining buffer (appendices, solutions and mediums) with low speed shaking at room temperature overnight. Destain the gel by soaking in water for at least 2 hours.

2.2.3.2 Silver staining

The more sensitive silver staining method was applied for SDS-PAGE gel visualization and detection of protein having low concentration. The staining protocol is shown below, and all manipulations were carried out at room temperature with low speed shaking (appendices, solutions and mediums).

- 1. Fix the protein by incubate it in fixer solution for 1 hour.
- 2. Wash the gel twice by soaking in deionized water for 2 minutes;
- 3. Sensitize by soaking in sensitizing solution for 30 minutes;
- 4. Wash the gel twice by deionized water for 20 seconds each time;
- 5. Incubate gel for 20 minutes in silver nitrate solution;
- 6. Wash the gel twice by deionized water for 1 minutes each time;
- 7. Develop the gel in visualizing solution until protein bands appear;
- Terminate the staining by changing to stop solution and incubate for 5 minutes.

2.3 Activity assay and partial purification of *E. gracilis* intracellular proteins

2.3.1 E. gracilis cultivation

The *Euglena gracilis* Z wild strain was obtained from the culture collection of Fermentation Technology Research Group, Bielefeld University and cultivated in a basal medium (Table 11), in 300 mL shake flask at 37 °C with shaking (120 rpm, shaking orbital 50 mm, Kuhner LS-X, Germany) under light exclusion. Each 100 mL medium was inoculated with 5 mL culture of exponential phase in the same medium.

Component	Concentration (g/L)
MgSO ₄ × 7 H ₂ O	0.23
(NH4)2HPO4	1.00
CaCl ₂ × 2 H ₂ O	0.03
KH ₂ PO ₄	1.00
NaC6H5O7 × 5.5 H2O	0.36
Glucose × H ₂ O	20.00
Trace-element	Concentration (mg/L)
Co(NO ₃) ₂	1.30
MnCl ₂ × 4 H ₂ O	1.80

Table 11 Medium for *E. gracilis* cultivation.

Materials and Methods		
CuSO4 × 5 H2O	0.02	
H ₂ MoO ₄	0.20	
ZnSO4 × 7 H2O	0.40	
Fe ₂ (SO ₄) ₃ × 6 H ₂ O	3.00	
Vitamine	Concentration (mg/L)	
Thiamin × HCI	0.01	
Cyanocobalamin	0.05	

2.3.2 Intracellular protein extraction preparation

E. gracilis cells were harvested by centrifugation at 3,500 g for 10 min and washed in demineralized water. The pellet was resuspended in 0.1 mol/L sodium acetate buffer (pH=6.0). The cell lysis was prepared by three cycles of sonication at the highest level for 30 s, and cooled in ice bath for 1 min between each cycle. The supernatant obtained by centrifugation at 10,000 g for 20 min was used for the further enzyme activity assay.

2.3.3 Determination of dry biomass and dry paramylon

For determining the dry biomass, a defined volume of culture was centrifuged at 3,500 g for 5 min. The cells were harvested, washed by demineralized water three times, and dried in 37 °C for 72 hours before weighting.

To weight dry paramylon, cells were harvested by centrifugation for 5 min at 3,500 g and the cell pellet was re-suspended in the same volume of demineralized water. The cells were disrupted by three cycles of sonication for 30 s at the highest level. After a further centrifugation at 3,500 g for 5 minutes

the insoluble substances were washed twice with the same volume of 100% ethanol. The insoluble proteins were removed by incubation with 1 g/L trypsin in 0.1 mol/L sodium phosphate buffer (pH 7.6) overnight (37-40 °C) on a shaker (about 20 rpm, shaking orbital 50 mm, Kuhner LS-X, Germany). After the incubation, the pellet was obtained by centrifugation at 3,500 g for 5 min, and followed by washing with demineralized water for twice. The paramylon pellet was dried in 37 °C for 72 hours before weighting.

2.3.4 Paramylon preparation for enzymatic hydrolysis

The native paramylon granule was prepared as the method mentioned in determination of dry paramylon in section 2.3.2.

To prepare 2× lyophilized paramylon, paramylon granule was dissolved in 0.1 mol/L KOH solution (in concentrations of 1.0 g/L, 2.5 g/L, and 5.0 g/L) and transferred to dialysis tubing (MWCO: 8.000 to 10.000, Zellu Trans Roth, Germany), then incubated for at least 24 hours in demineralized water and followed by lyophilization. The procedure was performed twice to produce the so called 2× lyophilized paramylon.

To prepare the alkali treated paramylon, the native granule was mixed with 1.0 mol/L KOH to concentration of 50.0 g/L, the mixture was adjusted to pH 6.0 by 0.1 mol/L HCl forming a paramylon gel. The so-called paramylon gel was diluted with demineralized water in concentrations of 1.0 g/L, 2.0 g/L, 2.5 g/L, and 5.0 g/L. The alkali treated paramylon and 2× lyophilized paramylon solutions were stored for long period at 4 °C and used as substrates for enzyme hydrolysis.

2.3.5 Investigation of intracellular β-1,3-glucanase activity of *E. gracilis* cultures

2.3.5.1 β-1,3-Glucanase activity assay

Glucanases hydrolyze substrate and generate reducing sugars and reducing chain ends with free carbonyl groups. These enzymes' activities are commonly assayed by quantifying the amount of the reducing groups in the product. Typically, by some colorimetric procedures, such as 3,5-dinitrosalicylic acid (DNS) method. The reducing groups react with 3,5-dinitrosalicylic acid (DNS) in alkaline solution forming 3-amino-5-nitro salicylic acid which is orange-red colored and could be detected spectrophotometrically at 540 nm (Fig.18).



Fig.18 Reaction of DNS with reducing sugar (Kyoto 2011).

The measurement of intracellular β -1,3-glucanase activity of *E.gracilis* cultures were performed according to a standard DNS assay procedure below.

- Following reagents were prepared: 600 g/L sodium potassium tartrate solution, and 20 g/L 3,5-DNS solution. Prepare DNS work solution freshly by mixing the two solutions mentioned above with equal volume.
- 180 μL glucose solution of different concentrations or sample was mixed with 60 μL DNS work solution in 96 well micro titer plate (BrandTech Scientific, Thermo Fisher Scientific, USA). Carefully float the plate on 98 °C water bath for 15 min. Measure the absorbance at

540 nm by SpectraMax M Microplate Readers (Molecular Devices, USA) after cooling to room temperature. Absorbance versus concentration of the standards were plotted to construct the standard curve (Fig.19).



Fig.19 Standard curve of DNS assay.

Fitting formula: y = 1503.8x, $R^2 = 0.9915$

3. Calculate the reducing sugar equivalent released by glucanase according to the standard sugar curve in term of mmol/L.

Reducing sugar equivalent $=\frac{y}{M} = 8.35x$ (mmol/L)

Where:

y= reducing sugar equivalent, mg/L;

M = molar mass of glucose, 180.16 g/mol.

2.3.5.2 Influence of substrate

For analyzing the influence of substrates to enzyme activity, native paramylon, $2 \times$ lyophilized paramylon or alkali treated paramylon in different concentrations were used as substrates for 100 µL of 10 mg/L laminarinase (endo- β -1,3-glucanase, Sigma-Aldrich, USA), respectively. After 8 hours incubation at 37 °C the enzyme activity was measured by DNS method.

2.3.5.3 Influence of intracellular protein concentration

To study the effect of *E.gracilis* intracellular protein concentration on the degree of conversion towards alkali treated paramylon, concentrated extracts (5-,10-,20-, and 50-fold) from a 7 days culture were prepared. 100 μ L extracts were used as crude enzymes and incubated with 1 mL, 1 g/L alkali treated paramylon at 37 °C for 8 hours.

2.3.5.4 Influence of the extraction buffer pH value

pH value of the extraction buffer have influence on hydrolases activity of *E.gracilis* extracts. After cultivation for 7 days, cell free extracts were prepared in 0.1 mol/L sodium acetate buffer at different pH values (6.0, 5.5, 5.0, 4.5) as crude enzymes. 100 μ L enzyme samples were incubated with 1 mL of 1 g/L 2× lyophilized paramylon at 37 °C for 8 hours and followed by reducing group measurement.

2.3.5.5 Influence of hydrolysis temperature

1 mL of 1 g/L alkali treated paramylon (pH=5.5) with 100 μ L of 50× extract was incubated for 8 hours at 30, 37, 45, 50, and 55 °C, followed by reducing group measurement.

2.3.5.6 Influence of incubation duration

100 μ L of 50× extract of the 7-day-old culture was incubated at 37 °C in 1 mL of 1 g/L alkali treated paramylon (pH 5,5) for 16 hours with hourly measurement.

2.3.5.7 Influence of cultivation duration

During cultivation time of 14 days, samples were taken daily and produced a $50 \times$ extract. 100 µL of the extraction was incubated in 1 mL of 1 g/L alkali treated paramylon (pH 5,5) for 16 hours at 37 °C and followed by reducing group assay.

2.3.6 Partial purification of intracellular proteins from *E.* gracilis

2.3.6.1 (NH₄)₂SO₄ Precipitation

E. gracilis intracellular protein extract was prepared as section 2.3.2. After centrifugation at 22,000 g for 20 min, the supernatant was precipitated to a final (NH₄)₂SO₄ concentration of 0-20%, 20-60% and 60-80% saturation respectively. Each precipitate was dialyzed (MWCO: 8.000 to 10.000, Zellu Trans Roth, Germany) against 20 mmol/L Tris-HCl buffer (pH=5.5) at 4 °C for 4 hours. Hydrolytic activity was found in the precipitate of 20-60% (NH₄)₂SO₄ saturation.

2.3.6.2 Anion exchange chromatography

The 20–60% saturation precipitate was dialyzed (MWCO: 8.000 to 10.000, Zellu Trans Roth, Germany) against 20 mmol/L Tris-HCl buffer (pH=5.0) at 4 °C for 4 hours. The enzyme extraction was applied to a weak anion exchange chromatography column (Toyopearl DEAE 650, 1.6 cm × 10 cm), using 20 mmol/L Tris-HCl buffer (pH=5.0) as loading buffer at flow rate of 1.0 mL/min. Proteins with glucanase activity against alkali alkali treated paramylon were

found in flow through fraction under pH=5.0 and the flow through peak was collected.

The collected proteins were dialyzed (MWCO: 8.000 to 10.000, Zellu Trans Roth, Germany) for changing buffer to 20 mmol/L Tris-HCI (pH=7.0) at 4 °C for 4 hours and then applied to the same anion exchange chromatography column using 20 mmol/L Tris-HCI buffer (pH=7.0) as loading buffer at flow rate of 1.0 mL/min. Proteins were eluted by stepwise method. The fraction having glucanase activity was eluted by 20% elution buffer (20 mmol/L Tris-HCI, 1 mol/L NaCI, (pH=7.0), flow rate 1.0 mL/min). And this peak was collected for future purification.

2.3.6.3 Hydrophobic chromatography

Enzyme sample obtained from ion exchange chromatography was mixed with the same volume of solution consisting of 40 mmol/L Tris-HCI and 1.8 mol/L ammonium sulfate. Then applied to a hydrophobic chromatography column (Fractogel TSK Butyl 650 (s), 1.6 cm × 6 cm; loading solution: 20 mmol/L Tris-HCI and 0.9 mol/L ammonium sulfate, pH=5.5; eluting solution: 20 mmol/L Tris-HCI, pH=5.5). Proteins were eluted stepwise and the fractions were dialyzed (MWCO: 8.000 to 10.000, Zellu Trans Roth, Germany) against 20 mmol/L Tris-HCI buffer (pH=5.5) at 4 °C for 4 hours followed by hydrolytic assay toward alkali alkali treated paramylon. Paramylon degrading activity was obtained from 45 % to 80% eluting fraction.

2.3.6.4 Size exclusion chromatography

The fraction with glucanase enzyme activity collected from hydrophobic chromatography was concentrated by viva-spin (MWCO 5000, GE Healthcare, USA) to approximately 1.0 mL and then applied to a size exclusion chromatography column (Superdex 200, 1.6 cm × 120 cm) using 20 mmol/L

Tris-HCl buffer (pH=5.5). Fractions were collected by the size of 1.0 mL at a flow rate of 1.0 mL/min for enzyme activity and protein concentration assay.

2.3.7 Trial of gene amplification of paramylon degrading enzyme in *E. gracilis*

2.3.7.1 Total RNA isolation and cDNA synthesis

Total RNA of *E. gracilis* isolation was performed as described in section 2.1.1. 50 mg lyophilized *E. gracilis* cells was used for each mini column according to the standard procedure. After total RNA clean-up, cDNA synthesis was conducted using $oligo(dT)_{16}$ primer with adapter (See Table 12).

2.3.7.2 Design of degenerate primer

According to the same peptide sequence identified by mass spectrum, four degenerate primers (in Table 12) was designed in order to lower the degeneration level.

Table 12 Primers for degenerate PCR.

Primer	Sequence 5'-3'
<i>EUG</i> LysoF-a	CGC GGI ATH YTI CAA ATH AAC
<i>EUG</i> LysoF-b	CGC GGI ATH YTI CAA ATH AAT
<i>EUG</i> LysoF-c	CGC GGI ATH YTI CAG ATH AAC
<i>EUG</i> LysoF-d	CGC GGI ATH YTI CAG ATH AAT
Adapter oligo	GGC CAC GCG TCG ACT AGT ACT TTT TTT TTT TTT
(dT) ₁₆ primer	ттт т

AUAP GGC CAC GCG TCG ACT AGT AC

I=A+T+G+C, H=A+T+C, Y=C+T.

2.3.7.3 Degenerate PCR

The summary of target protein gene of *E. gracilis* amplification was shown in Fig.20.



Fig.20 Summary of degenerate amplification.

First strand cDNA synthesis was synthesized from the poly A tail of mRNA using an adapted poly T primer. Amplification of target gene was performed using two primers, the degenerate primer that prospectively annealing to the target region in the cDNA and the other was a primer targeting the adapter of the poly T end generated from cDNA synthesis. The PCR was conducted as followed.

1. Mix the following reagent into a PCR tube as Table 13;

Reagents	Volume (µL)
Template cDNA	1.0
10 μ mol/L Oligo (dT) ₁₆ primer with adapter	1.0
10 µmol/L AUAP primer	1.0
5× Reaction buffer	2.0
10 mmol/L dNTPs	0.2
DNA polymerase	0.1
ddH2O	to total volume of 10 μ L

Table 13 Degenerate PCR reaction mixture.

2. Start the amplification program as described in Table 14;

Table 14 Degenerate PCR thermocycling conditions.

Temperature (°C)	Duration	Cycle
98	30 s	1
98	10 s	
55	20 s	45
72	80 s	
72	10 min	1

4	Hold

3. Analysis of degenerate PCR products by 10.0 g/L agarose gel electrophoresis.

2.4 Cloning and expression of recombinant β -1,3-glucanase in *P. pastoris*

2.4.1 Gene cloning of recombinant β-1,3-glucanase

2.4.1.1 Trichoderma reesei cultivation

Spore suspension of *T. reesei* QM 6a (ATCC[®] 13631[™]) was routinely maintained from potato dextrose agar plate (PDA) at 28 °C. Suspension containing 1×10^6 spores/mL was washed from the PDA plate with sterile 0.05% (v/v) Tween 80 solution and used to inoculate 100 mL of glucose-yeast extract medium (Table 15, pH=7.0) in a 500 mL baffled shake flask. The spores were grown at 28 °C for 24 h with shaking frequence of 180 rpm, shaking orbital of 50 mm (Kuhner LS-X, Germany).

Component	Concentration
Glucose	10.0 g/L
Yeast extract	10.0 g/L
(NH4)2SO4	1.4 g/L
KH ₂ PO ₄	2.0 g/L
CaCl ₂	0.4 g/L
MgSO ₄	0.3 g/L

Table 15 Glucose-yeast extract medium.

0.5 g/L FeSO ₄ *	1 mL/L
3.7 g/L CoCl ₂ *	1 mL/L
1.6 g/L MnSO4*	1 mL/L
1.4 g/L ZnSO4*	1 mL/L

*: Autoclaved separately and added into medium before use.

The myceliums obtained afterward were harvest by centrifugation and washed twice by sterile demineralized water before changing to inducing medium. For induction, lactose-yeast extract medium (substitute lactose for glucose in glucose-yeast extract medium, pH=7.0) was used and incubated for 48 h under 28 °C on a shaker at 180 rpm (shaking orbital 50 mm, Kuhner LS-X, Germany).

2.4.1.2 Total RNA isolation and cDNA synthesis of T. reesei

The myceliums harvested after induction were freeze dried and saved at -80 °C until total RNA isolation. The RNA isolation and cDNA synthesis were carried out as described in section 2.1.1 and 2.1.2. The cDNA produced was kept in - 80 °C until use.

2.4.1.3 Gene amplification of β-1,3-glucanase from *T. reesei*

Primers were designed specifically to amplify the β -1,3-glucanase enzyme genes (Table 16). The fragment of *Kex2* processing site (5´-AAAAGA-3´) for secretion signal α -factor cleavage downstream of *Xho* I site was added into the 5´ end of forward primers. *Sal* I was used instead when amplified gene containing *Xho* I cleavage site. Meanwhile gene of purification tag (Hexa-

histidine) and stop codon followed by restriction cleavage site *Not* I were designed at the 3['] end of reverse primers.

 β -1,3-glucanase enzyme genes from *T. reesei* were amplified by PCR using cDNA as template. The products were analyzed by agarose gel electrophoresis and fragments with the right size were cut and cleaned up before ligated into the pJET 1.2/blunt Cloning Vector (Thermo Scientific, USA) according to the standard procedure provided, then transformed into *E. coli* DH5 α (Procedures see section 2.1). Recombinant vectors were screened by restriction endonuclease digestion and the positive ones containing right size insertion were used for recombinant expression vector construction.
Table 16 Primers for gene amplification from *T. reesei*.

Enzyme	Primer	Sequence 5'-3'
TrGH16	P.p45305ForwSal	<u>GTCGAC</u> AAAAGATGGGCACCTCCGTCCTATG
EGR45305.1	P.p45305Rev	<u>GCGGCCGC</u> TTAGTGATGATGATGGTGATGGGTAGAGTACTGGGCAACGTAG
TrGH55	P.p47521Forw	TATA <u>CTCGAG</u> AAAAGAACGCCCACCCCG
EGR47521.1	P.p47521Rev	<u>GCGGCCGC</u> CTAATGGTGATGATGATGATGGACGGTATAACGACCGACAT
TrGH64	P.p45483Forw	CTCGAG AAAAGAGCCCCCTCTCTCATCACAAGAGTTGC
EGR 45483.1	P.p45483Rev	<u>GCGGCCGC</u> TTAATGATGATGATGATGATGAGGAGGAGGCGCGCCGATG
TrGH81	P.p49603Frow	CTCGAG AAAAGACTGCCCAATGGCGCAAA
EGR49603.1	P.p49603Rev	<u>GCGGCCGC</u> CTAGTGGTGATGATGATGCGACTGGACAAACTTGAACC
TrGH17	P.p46171Forw	TTTG <u>CTCGAG</u> AAAAGACAGAACAACTACCTCGGCTTCAAC
EGR46171.1	P.p46171Rev	AAA <u>GCGGCCGC</u> TTAGTGATGGTGATGGTGATGGGTGGCGGTGGGCTTG

2.4.1.4 Gene amplification of β -1,3-glucanase from *P. pastoris*

For obtaining gene of one exo- β -1,3-glucanase from *P. pastoris* GS115, the intronless gene was cloned by PCR-based strategy using genomic DNA as template (section 2.4.1.7). The specific primer designing (In Table 17), and gene amplification were conducted the same as described in section 2.4.1.3.

Table 17	Primers f	for exo-β	-1,3-gluca	nase gene	amplification	n from <i>P.</i>	patoris.

Enzyme	Primer	Sequence 5'-3'
<i>Pp</i> GH5	P.p69081Forw	CCGG <u>CTCGAG</u> AAAAGATACTCCAGT GAAAAAATCATGGG
CAY69081.1	P.p69081Rev	ATT <u>GCGGCCGC</u> TTAGTGATGGTGATGG TGATGGAACCCACATTGATTTGGATA

2.4.2 Expression vector construction

The secretory expression vector pAaHBgI (collected by our group; Fig.21) was selected for recombinant vector construction. The *Kex2* processing site for secretion signal α -factor cleavage was added into forward primers upstream of the gene of interest, furthermore the purification tag (Hexa-histidine) and stop codon were added by the end of reverse primers. The gene insertion was performed between restriction cleavage sites *Xho* I (*Sal* I was used when the gene contains *Xho* I site) and *Not* I. Constructed expression vectors were transformed into *E.coli* DH5 α and colonies appeared on the selective agar plate were cultivated in order to proof correct clones via plasmid isolation. All related manipulations are described in section 2.1.



Fig.21 Map of *P. pastoris* expression vector pAaHBgl (Ahmad et al. 2014).

2.4.3 Transformation of recombinant P. pastoris

The constructed expression vectors were linearized at two *Bgl* II sites (bases 6153 and 4161 in Fig.21) located upstream of *AOX*1 promoter and downstream of *3'* UTR *AOX*1. The long fragment with a length of about 8 kb was isolated and cleaned up before concentrating with Vacuubrand[®] vacuum system (Sigma-Aldrich, USA).

Exceptionally, when the gene of interest contains *Bgl* II site, the DNA fragment between two *Bgl* II site (bases 6153 and 4161 in Fig.21) was amplified by PCR using primers given in Table 18. The amplification product was cleaned up directly after PCR reaction.

Table 18 Primers for *P. patoris* transformation.

Primer	Sequence 5'-3'
pAaHBgI-PAOX1-Forw	ATCTAACATCCAAAGACGAAAGGTTGAATGA
pAaHBgl-UTRAOX1-Rev	TCTTGAGATAAATTTCACGTTTAAAATCAGCGTA CC

Transformation of *P. pastoris* was carried out by electroporation. The protocol in detail was described in section 2.1.10. Recombinant *P. patoris* were screened for recombinant enzyme expression by SDS-PAGE and enzyme activity assay.

2.4.4 Recombinant protein expression in *P. pastoris*

2.4.4.1 Recombinant expression in 24 square-well plate

For selecting the positive expression recombinant, the pilot expression was performed involving picking colonies from selective plates, cultivation in a small scale culture, methanol inducing, SDS-PAGE and enzyme activity assay. For small scale expression, the 24 square-well plate (HJ-bioanalytik, Germany) was used following the procedure below.

- A single colony was picked from selective agar plate, and used to inoculate 5 mL YPD medium in one well of the plate. Cultivation was performed for 48 hours at 30 °C, 180 rpm, shaking orbital 20 mm (IKA-KS 4000ic control, Germany);
- Harvest the cells by centrifuging at 1500g for 5 minutes at 4 °C and dispose the supernatant. Wash the cells with 5 mL BMMY medium twice, then resuspend cell pellet in 3 mL BMMY medium for protein expression;

- Cultivate under the same conditions and add to each well 150 µL of 10% (v/v) methanol to the final concentration of 0.5% every 24 hours;
- 4. Stop inducing after 120 hours. The cells and supernatant were separated by centrifugation at 1500g for 5 minutes at 4 °C and stored at -20 °C.
- 5. Use the supernatant as crude sample for enzyme activity assay and protein electrophoresis.

2.4.4.2 Recombinant expression in shake flask

Positive results from 2.4.4.1 were confirmed by expression in 250 mL baffled flasks following the method below.

- 100 µL of glycerol stork of positive results from 2.4.4.1 was used to inoculate 5 mL of YPD in 24 square-well plate (HJ-bioanalytik, Germany), and grow at 30 °C in a shaking incubator (180 rpm, shaking orbital 20 mm, IKA-KS 4000ic control, Germany) for 24 hours;
- The 5 mL seed culture was mixed with 50 mL BMGY medium in a 250 mL baffled flask and grow at 30 °C in a shaking incubator (180 rpm, shaking orbital 50 mm, Kuhner LS-X, Germany) for 24 hours;
- Harvest the cells by centrifugation at 1,500g for 5 minutes at 4 °C, and decant the supernatant. The cells were washed twice with 50 mL BMMY medium, then resuspended in 50 mL BMMY medium for protein expression;
- 100% methanol was added to a final concentration of 0.5% (v/v) every
 24 hours until the inducting time reaches 120 hours;
- Samples taken every 24 hours were centrifugated at 1,500g for 5 minutes at 4 °C. The cell pellet and supernatant were stored separately in -20 °C.

2.4.5 β-1,3-glucanase enzyme activity assay

2.4.5.1 Preparation of substrate

Alkali treated paramylon (2.0 g/L, pH=6.5) was prepared as described in section 2.3.4. The alkali treated paramylon could be stored for long period at 4 °C and used as substrate for activity assay.

Laminarin was dissolved in 50 mmol/L sodium acetate buffer (pH=6.5) to a final concentration of 2.0 g/L.

2.4.5.2 Activity assay of recombinant endo-β-1,3-glucanase

Glucanase hydrolyzes glucans to produce reducing sugars and reducing ends with the presence of free carbonyl groups. These enzymes are commonly assayed by quantifying the amount of the reducing groups during the assay. For determining reducing groups in low concentration, a sensitive 3-methyl-2benzothiazolinone hydrazine (MBTH) assay was used (Anthon and Barrett 2002). This method was developed to measure aldehydes originally (Hauser and Cummins 1964), and had been adapted to determine the quantification of reducing sugars. It had been confirmed for the determination of polygalacturonase activity (Anthon and Barrett 2002) and saccharification assay (Gomez et al. 2010).

In this work, the highly sensitive MBTH method was optimized for standard 96-well microtiter plate with a final volume of 200 µL.

- Prepare MBTH reagent by freshly mixing equal volumes of 0.43 g/L MBTH and 0.14 g/L DL-dithiothreitol (DTT) before use. The MBTH and DTT stock solutions should be stored at 4 °C;
- Prepare oxidizing reagent solution containing 5.0 g/L NH₄Fe(SO₄)₂ 12H₂O, 0.5 % (v/v) H₂SO₄ and 0.8 % (v/v) HCl. The oxidizing reagent was stable at room temperature for at least one week;

- Mix 90 μL enzyme sample with 90 μL substrate solution in 96 well micro titer plate (BrandTech Scientific, Thermo Fisher Scientific, USA). The mixture was incubated at 37 °C for 60 min afterwards;
- Prepare glucose standard samples with concentration from 0 to 500 µmol/L for standard curve;
- Pipet 25 µL of standard solution or sample into each well of the microtiter plate;
- Add 25 μL of 1.0 mol/L NaOH solution followed by 50 μL MBTH reagent in each well and mix thoroughly;
- Cover the multiwall plate by sealing tape and incubate at 80 °C for 20 min;
- Mix 100 µL oxidizing reagent to each well. The samples were cooled to room temperature before determining the absorbance at 620 nm by SpectraMax M Microplate Readers (Molecular Devices, USA). Absorbance versus concentration of the standards were plotted to construct the standard curve (Fig.22);



Fig.22 Standard curve of MBTH assay.

Fitting formula: y = 670.69x, $R^2 = 0.9952$

9. Calculate the reducing suger equivalents released by glucanase according to the standard sugar curve and convert the enzyme activity in term of IU/L, where one unit of enzyme activity (IU) is defined as 1 µmol of reducing sugar equivalent released in 1 min.

Enzyme activity = $\frac{y}{t} \times 8 = 89.43x$, (IU/L)

Where:

y= 670.69x, μmol/L x= absorbance at 620 nm t= enzyme reaction duration, 60 min 8= dilution factor.

2.4.5.3 Activity assay of recombinant exo-β-1,3-glucanase

To determine exo- β -1,3-glucanase activity, the hydrolysis product glucose was determined by D-glucose assay kit (R-Biopharm, Germany). D-glucose was phosphorylated to D-glucose-phosphate (G-6-P) in the presence of the enzyme hexokinase and adenpsine-5'-triphosphate (ATP) with the simultaneous formation of adenosine-5'-diphosphate. In the presence of the enzyme glucose-6-phosphate dehydrogenase, G-6-P was oxidized by nicotinamide-adenine dinucleotide phosphate (NADP) to D-gluconate-6-phosphate with the formation of reduced nicotinamide-adenine dinucleotide phosphate (NADP) to D-gluconate-6-phosphate (NADPH). The amount of NADPH formed in the rection was stoichiometric to the amount of D-glucose. The increase in NADPH was measured by means of absorbance at 340 nm (Kunst et al. 1984). The experiments were conducted by the following procedure.

In this work, the assay was optimized for 96-well quartz micro titer plate.

- 1. Mix 100 μ L enzyme sample with 100 μ L substrate solution and incubate the mixture at 37 °C for 15 min;
- 2. Terminating the reaction by inactivating the enzyme at 98 °C for 2 min;
- 3. Determine the produced glucose by D-glucose assay kit;

The assay reagents triethanolamine buffer (TAE buffer) and standard solution were prepared according to the original procedure. Components of reaction mixture were pipetted into quartz microtiter plate wells as shown in Table 19.

Component	Volume (µL)			
	Blank	Standard	Sample	
TAE buffer	80	80	80	
Sample solution	-	-	8	
Standard solution	-	8	-	
Water	160	152	152	
Read the absorbance at 340 nm (A1) approximately 3 min after mixing				
HK/G6P-DH suspension	1.6	1.6	1.6	
Read the absorbance at 340 nm (A ₂) approximately 15 min after mixing				

Table 19 Components of glucose assay reaction.

4. The concentration of glucose in the sample solution was calculated as:

$$c = \frac{V \times \Delta A}{\varepsilon \times d \times v \times 1000} = 0.4794 \times \Delta A$$
, (mol/L)

Where:

c = glucose concentration in the sample solution, mol/L

v = 0.2416 mL, assay volume

 $\Delta A = (A2-A1)$ sample-(A2-A1) blank

- ε = 6.3 L/(mmol×cm), molecular absorption coefficient of NADPH at 340 nm
- d = 0.1 cm, light path from well botton to liquid surface

v = 0.0008 ml, sample volume

1000 $\frac{\text{mol}}{\text{mmol}}$ = conversion from mmol to mol

 Convert exo-β-1,3-glucanase enzyme activity in term of IU/L, where one unit of enzyme activity (IU) was defined as the amount of enzyme required to liberate 1 µmol reducing sugar equivalent in 1 minute under the assay condition;

Enzyme activity
$$=\frac{c \times 10^6}{t} \times 2 = 6.392 \times 10^4 \times \Delta A$$
, (IU/L)

Where:

 $c = 0.4794 \times \Delta A$, mol/L (see step 4.)

t = 15 min, incubate duration

 $10^6 \frac{\mu M}{M}$ = conversion from mol/L to μ mol/L

2 = dilution factor.

2.5 Cloning and expression of recombinant lytic polysaccharide monooxygenase in *P. pastoris*

2.5.1 Strain and cultivation

2.5.1.1 Strain and cultivation

Trichoderma reesei QM6a and *Aspergillus oryzae* (ATCC[®] 42149[™]) were grown on potato dextrose agar (PDA, BD, Germany) at 28 °C to obtain spores. Spore suspensions were diluted to 1×10^6 spores/mL with sterile 0.05% (v/v) Tween 80 solution and used to inoculate 100 mL pre-culivation medium (pH=7.0, Table 20) in a 500 mL baffled shake flask. The pre-cultivation was performed at 28 °C, 180 rpm, shaking orbital of 50 mm (Kuhner LS-X, Germany) for 24 hours.

Biomass obtained from pre-cultivation was harvest by centrifugation at 10,000 g for 15 minutes at 4 °C and washed twice by sterile demineralized water before changing to production medium. In production medium, components were the same as in pre-cultivation medium, except paramylon granule was used at the concentration of 2.5 g/L as the only carbon source instead of glucose.

Component	Concentration
Glucose	10.0 g/L
(NH4)2SO4	1.4 g/L

Table 20 Pre-cultivation medium for T. reesei and A. oryzae

Materials and M	lethods
KH ₂ PO ₄	2.0 g/L
CaCl ₂	0.4 g/L
MgSO ₄	0.3 g/L
0.5 g/L FeSO₄*	1.0 mL/L
3.7 g/L CoCl ₂ *	1.0 mL/L
1.6 g/L MnSO4*	1.0 mL/L
1.4 g/L ZnSO4*	1.0 mL/L

*: Autoclaved separately and added into medium before use.

2.5.1.2 Enzyme activity assay of β -1,3-glucanase induced by paramylon

The supernatant was collected as crude enzyme by centrifugation at 10,000 g, 4 °C for 15 min. The activity assay of β -1,3-glucanase was conducted as described in section 2.3.5.1, using alkali treated paramylon or laminarin as substrate. Enzyme activity was estimated as hydrolysis rate by determining reducing groups by DNS method.

2.5.2 Gene cloning and expression of LPMOs

Lytic polysaccharide mono-oxygenases (LPMOs) are a recently discovered class of enzymes capable of oxidizing recalcitrant polysaccharides. LPMOs from *T. reesei* and *A.oryzae* were selected and the genes were obtained for heteroexpression in *P. pastoris*.

Gene of *Tr*AA9 (GeneBank: EGR52697.1) from *T. reesei* was cloned using cDNA as template. Gene of *Ao*AA11 (GeneBank: BAE61530.1) was obtained by PCR-based strategy using genomic DNA as template. The detailed preparation of cDNA and gnomic DNA are described in section 2.1.

Gene specific primers of LPMOs (See Table 21) were designed and PCR amplification followed the same strategy as described in section 2.4.1.3.

Recombinant expression of LPMOs in the yeast *P. pastoris* were performed following the β -1,3-glucanase expression described in section 2.4.3-2.4.5.

Enzyme	Primer	Sequence 5'-3'
TrAA9	<i>P.p</i> 52697Forw	<u>CTCGAG</u> AAAAGACATGGACATATTAAT GACATTGTCATC
EGR52697.1	<i>P.p</i> 52697Rev	<u>GCGGCCGC</u> TTAGTGATGATGATGGTG ATGGTTAAGGCACTGGGCGTAG
AoAA11	<i>P.p</i> 61530Forw	TTCC <u>CTCGAG</u> AAAAGACACATGATGAT GGCGCAGCCC
BAE61530.1	<i>P.p</i> 61530Rev	ATT <u>GCGGCCGC</u> TTAGTGGTGATGATG GTGGTGGCAGGCCTGGCTGCCC

Table 21 Primers for LPMO gene amplification.

2.5.3 Synergistic action assay between LPMOs and β -1,3-

glucanase

The copper-dependent lytic polysaccharide monooxygenases (LPMOs) have been shown to play a central role in oxidative degradation of cellulose. But determination of the activity of LPMO directly remains difficult, therefore the strategy of detecting the soluble product after synergistic action of LMPO and exo- or endo- β -1,3-glucanase was used. The synergistic degrading insoluble paramylon were conducted as described below.

- As the recombinant enzymes were secreted, the supernatants were used as crude enzyme solution. These colutions were dialyzed against 50 mM sodium acetate (pH 6.5) by dialysis tubular membrane (MWCO 6000-8000, GE Healthcare, USA) for 4 hours at 4 °C followed by filtering sterilization using 0.22 µm filter (Millex-GP, Merck, Germany);
- 2. Crystalline paramylon granules were autoclaved and dried before use;
- 3. Enzymes were mixed in the ratio of 1:1 by volume and paramylon was added to the final concentration of 1.0 g/L;
- The mixtures were incubated at 37 °C with shakeing (30 rpm, shaking orbital 50 mm, Kuhner LS-X, Germany) for 72 hours;
- The supernatants were centrifuged using Vivaspins (MWCO 5000, Zellu Trans Roth, Germany) at 5000 g for 5 minutes. The filtrate was collected for total carbohydrate assay;
- Phenol sulfuric acid method was used to assay for total carbohydrate in the product.

Standard procedure of phenol sulfuric acid assay for 96 well mocrotiterplate was performed as described below.

- Pipet 50 μL sample or glucose standard solution (0-1.5 mmol/L) into 96 well micro titer plate (BrandTech Scientific, Thermo Fisher Scientific, USA). Add 150 μL of 95% (v/v) sulphuric acid into each sample and mix thoroughly;
- 2. Add 30 µL of 0.5 mol/L phenol into each sample and mix well;

 Incubate the plate at 90 °C for 5 min and measure the absorbance at 490 nm by SpectraMax M Microplate Readers (Molecular Devices, USA). Absorbance versus concentration of the standards were plotted to construct the standard curve (Fig.23);



Fig.23 Standard curve of phenol sulfuric acid assay.

Fitting formula: y = 0.4614x, $R^2 = 0.9901$

Reducing sugar equivalent = $y \times M = 83.13x$ (mg/L)

Where:

$$y = 0.4614x, mmol/L$$

- x = absorbance at 490 nm
- M = molar mass of glucose, 180.16 g/mol.

3 Results and Discussion

In this chapter, results from three parts of this work were described and further discussed. The first part (section 3.1) comprised of results from the investigation and partial purification of β -1,3-glucanase from intracellular *E. gracilis* proteins. The results of cloning and expression of recombinant β -1,3-glucanases formed the second part (section 3.2). In this part, four endo- β -1,3-glucanases (*Tr*GH16, *Tr*GH55, *Tr*GH64, *Tr*GH81) and two exo- β -1,3-glucanase (*Tr*GH17, *Pp*GH5) were successfully expressed in *P. pastoris*, and their activities toward alkali treated paramylon were confirmed. In the last part of this work (section 3.3), results from trail of the newly discovered LPMOs deconstructing paramylon granule were presented, including cloning and expression of two LPMOs (*Tr*AA9, *Ao*AA11) in *P. pastoris* and their synergistic actions with β -1,3-glucanase on paramylon granule.

3.1 Investigation of β -1,3-glucanase(s) from intracellular *E. gracilis* proteins

Paramylon acts as storage polysaccharide in *E. gracilis*, therefore, intracellular enzymes should be expressed to deconstruct paramylon for autotrophic nutrition. For this reason, *E. gracilis* intracellular proteins were extracted for investigating the hydrolyzing ability on paramylon in this part. Following the characterizations of enzyme action (section 3.1.1-3.1.6), results from protein purification and gene amplification of paramylon degrading enzyme from intracellular *E. gracilis* proteins were presented in sections 3.1.7 and 3.1.8.

3.1.1 Influence of substrate

The native paramylon granule is extremely resistant to enzymatic attacks. For a proper pretreat which modifies paramylon more accessible, the native granule, $2 \times$ lyophilized and alkali treated paramylon were prepared. These three substrates at different concentrations (1.0, 2.5, 5.0 g/L) were hydrolyzed by 10 mg/L laminarinase (1,3- β -D-glucan glucanohydrolase, EC 3.2.1.39; dissolved in 0.1 mol/L sodium acetate buffer, pH=6.0), and the conversions were compared by reducing sugar equivalents as enzymatic hydrolysate.



Fig.24 Influence of substrate on hydrolytic activity of laminarinase. 100 μ L 10 mg/L laminarinase was incubated with 1 mL substrates respectively at 37 °C for 8 hours. The reducing sugar equivalents were measured by 3,5-dinitrosalicylic acid (DNS) method (section 2.3.5.1).

Because of the high crystallinity (approaching 90%), the paramylon granules were rather difficult to be hydrolyzed at all concentrations tested, the recalcitrant physical structure blocked the interaction between enzyme and substrate resulting no reducing power of the reaction mixture containing native granule as substrate. By now there is no successful hydrolysis of native granules *in vitro*.

In comparison, enzymatic hydrolysis became possible after pretreatment. Both pretreated substrates showed increasing in conversion. By providing easier access to substrate chain, which were previously hidden within the tightly packed granule, pretreatment became essential for paramylon hydrolyzation. The laminarin activities toward 2x lyophilized and alkali treated paramylon calculated as reducing sugar equivalent were averagely 1.5 mmol/L and 1.3 mmol/L. The 2x lyophilized paramylon was easier for enzymatic hydrolysis than alkali treated paramylon, and it presented in powder form making the determination more accurately. However, 2x lyophilized paramylon was prepared by two cycles of dialysis and freeze drying costing more than 96 hours. Comparing to the time-consuming method, dilution and neutralization of concentrated paramylon-gel dissolved in 0.5 mol/L KOH could be quickly accomplished. When this preparation (alkali treated paramylon) was used as substrate, an obvious high yield hydrolysate was observed. It demonstrated that alkali treated paramylon was also a good substrate for enzyme hydrolysis. Therefore, alkali treated paramylon was used as substrate for further investigating in following sections.

3.1.2 Influence of intracellular protein extract concentration

To study the effect of *E. gracilis* intracellular extract concentration on the degree of conversion towards alkali treated paramylon, concentrated extracts from a 7 days culture in 5, 10, 20 and 50 times were prepared in 0.1 mol/L sodium acetate buffer (pH=6.0). These extracts were used as crude enzymes and incubated with alkali treated paramylon (adjusted to pH=6.0) at 37 °C for 8 hours. Laminarinase (10 mg/L, dissolved in 0.1 mol/L sodium acetate buffer, pH=6.0) and deionized water were used as controls.



Fig.25 Influence of intracellular extract concentration on hydrolytic activity towards alkali treated paramylon. 100 μ L concentrated extracts (5-,10-,20-, and 50-fold) from a 7 days culture were incubated with 1 mL, 1 g/L alkali treated paramylon at 37 °C, pH=6.0, for 8 hours. As controls, 100 μ L of 10 mg/L laminarinase and deionized water were used instead of intracellular extracts. The reducing sugar equivalents were measured by 3,5-dinitrosalicylic acid (DNS) method (section 2.3.5.1).

After incubation, enzymes in 50× extract showed stronger hydrolysis activity comparing to lower concentrated extracts resulting a reducing sugar equivalent of 1.94 mmol/L. And the 50× extract was the only crude enzyme sample generated more reducing sugars than the positive control 10 mg/L laminarinase. Relatively, extract at 5, 10, and 20 folds, contributed lower enzymatic degrading power due to smaller quantity of enzymes. Laminarinase contained only endo- β -1,3-glucanase which catalyzed the hydrolysis of internal β -1,3 linkages in paramylon, different from which, the cell-free extract contained various paramylon degrading enzymes acting synergistically. The presence of active β -

1,3-glucan exo-, endo-hydrolases was reported in Barras and Stone's research (Barras and Stone 1969). Additionally, other enzymes' actions accelerated the degradation of paramylon, for instance, phosphorylases which have been characterized in *E. gracilis* were active toward short oligosaccharides released from paramylon (Kitaoka et al. 2012). Enzymes acted synergistically utilizing paramylon as energy source for cell growth and demonstrated combined hydrolysis power in the extract.

3.1.3 Influence of extraction buffer pH

The influence of pH on activity of the hydrolases in *E. gracilis* was determined with 50 times concentrated cell extract after 7 days cultivation. These extracts and laminarinase solutions were prepared in 0.1 mol/L sodium acetate buffer at different pH values (pH=6.0, 5.5, 5.0, 4.5) as crude enzymes and positive controls. The samples were incubated with 2× lyophilized paramylon at 37 °C for 8 hours and followed by reducing group measurement.



Fig.26 Influence of pH on hydrolytic activity. 100 μ L 50-fold concentrated extracts at different pH values (6.0, 5.5, 5.0, 4.5) from a 7 days culture were

incubated with 1 mL of 1 g/L 2× lyophilized paramylon at 37 °C for 8 hours. As controls, 100 μ L of 10 mg/L laminarinase and 0.1 mol/L sodium acetate buffer were used instead of intracellular extracts. The reducing sugar equivalents were measured by 3,5-dinitrosalicylic acid (DNS) method (section 2.3.5.1).

The hydrolases from cell extract had optimal activity at pH 5.5, and contributed a reducing sugar equivalent of 2.05 mmol/L. At pH 4.5, laminarinanse generated 1.77 mmol/L hydrolysate, which indicated the highest activity. It was reported that the intracellular extract contained both endo- and exo-hydrolase, and an endo- β -1,3-glucanase from *E. gracilis* extract rapidly reduced the viscosity of CM-pachyman at pH 5.2 (Barras and Stone 1969). In this work, a significant increasing of reducing sugar equivalent after variation from pH 6.0 to 5.5 and descending of hydrolase activity as the pH values variation from pH 5.5 to 4.5 were oberserved. The highest activity at pH 5.5 suggested that the endo-hydrolase acted on internal glycosidic linkages of paramylon making non-reducing chain ends available for attack by the exo-hydrolase, and might achieve the highest synergy efficiency.

3.1.4 Influence of hydrolysis temperature

In addition to the optimal pH value, a temperature optimum was determined. In comparison of 50 times concentrated extract at different temperature from 30 to 55 °C against alkali treated paramylon, the best result was achieved at 37 °C.



Fig.27 Influence of hydrolysis temperature on hydrolytic activity. 1 mL of 1 g/L alkali treated paramylon (pH=5.5) with 100 μ L of 50× extract prepared from a 7-day-old culture in 0.1 mol/L sodium acetate buffer (pH=5.5) was incubated for 8 hours at 30, 37, 45, 50, and 55 °C, followed by reducing group measurement. As controls, 100 μ L of 10 mg/L laminarinase and 0.1 mol/L sodium acetate buffer were used instead of intracellular extracts. The reducing sugar equivalents were measured by 3,5-dinitrosalicylic acid (DNS) method (section 2.3.5.1).

The optimum temperature found in this study was different from reported researches. In works from Vogel and Barber, the optimum temperature of exoglucanase was 60 °C (Vogel and Barber 1968) and was further confirmed by Barras and Stone that the exo-hydrolase retained most activity at 55 °C (Barras and Stone 1969). The fundamental cause of this difference could lie in the combination of different enzymes in paramylon degradation. The crude enzyme had optimal activity at 37 °C, contributing a reducing sugar equivalent of 1.96 mmol/L, indicated the optimum temperature for paramylon hydrolases synergistic action in the extract. Results in Fig.27 showed that the laminarinase was active over a broad temperature range and obtained the best at 45 °C. At 45 °C, laminarinanse generated 1.92 mmol/L hydrolysate, which indicated the highest activity.

3.1.5 Influence of incubation duration

To analyze the effect of incubation duration on the degree of paramylon degradation, the $50\times$ extract was incubated with 1.0 g/L alkali treated paramylon as substrate. Over a period of 16 hours, an aliquot was measured every hour concerning the reducing group content.



Fig.28 Influence of incubation duration on hydrolytic activity. 1 mL of 1 g/L alkali treated paramylon (pH=5.5) with 100 μ L of 50× extract prepared from a 7-day-old culture in 0.1 mol/L sodium acetate buffer (pH=5.5) was incubated for 8 hours at 37 °C for 16 hours with reducing sugar equivalent measurement hourly by 3,5-dinitrosalicylic acid (DNS) method (section 2.3.5.1).

Under the optimized conditions, the reducing sugar equivalent accumulated continuously during the first 11 hours to 2.59 mmol/L. The amount of degrading product kept in a steady increasing when the incubation time exceeded 11 hours. At 16 h when the experience stopped, the *E. gracilis* intracellular hydrolases contributed a reducing sugar equivalent of 2.70 mmol/L. Similar result was reported by by Barras and Stone that significant amount of reducing sugars were released from untreated laminarin and dispersed paramylon in 12 hours, and continued tardily up to 24 hours (Barras and Stone 1969).

3.1.6 Influence of cultivation duration

The paramylon degrading enzymes were produced to deconstruct it when the cells need to utilize the energy reserve as carbon source. Activity of *E. gracilis* hydrolases from cell-free extract varied with the stage of growth. To invest the hydrolase activities in different growth stage, extract samples were prepared daily and incubated with 1 g/L alkali treated paramylon as substrate at 37 °C for 16 h followed by reducing product assay. Residual glucose and dry cell weight were also showed in the Fig.29.



Fig.29 Influence of cultivation duration on hydrolytic activity. During cultivation time of 14 days, samples were taken daily and produced a 50×

extract in 0.1 mol/L sodium acetate buffer (pH=5.5). 100 μ L of the extraction was incubated in 1 mL of 1 g/L alkali treated paramylon (pH 5,5) for 16 hours at 37 °C and followed by reducing group assay. The reducing sugar equivalents and residual glucose in the medium were measured by 3,5-dinitrosalicylic acid (DNS) method (section 2.3.5.1). The dry cell weight was measured by method in section 2.3.3.

Paramylon hydrolase activity increased sharply within the first 8 days of cultivation, revealing the initiation of paramylon utilization. As glucose in medium was consumed within 6 days, paramylon played a role as the only energy storage in this carbon-deprived medium. The amount of reducing product revealed that hydrolase activity in cell-free extract kept stable during the last 6 days. Meanwhile, the dry cell weight decreased as paramlon was degraded for metabolism. The similar result was found by Vogel and Barber that the specific glucanase activity in early stationary phase was at least twice as in the logarithmic growth phase whether calculated by protein content or cell number (Vogel and Barber 1968). Indicated by the study of *E. gracilis* cultivation time effect on hydrolase activity, a practical point that crude enzyme extract should be prepared from a culture in late stationary phase after consumption of the carbon source in the medium.

3.1.7 Partial purification of *E. gracilis* intracellular proteins

50× extract of E. gracilis was prepared using 12 day-old culture by the method described in section 2.3.2. After centrifugation at 22,000g for 20 min, 50 mL supernatant was precipitated at a final (NH₄)₂SO₄ concentration of 0-20, 20–60 and 60-80% saturation, respectively. Each precipitate was dialyzed against 20 mmol/L Tris-HCl buffer (pH=5.5) and adjusted to a volume of 50 mL by the same buffer afterwards. Fraction of 20-60% (NH₄)₂SO₄ precipitate showed activity towards alkali treated paramylon. Proteins obtained by (NH₄)₂SO₄ precipitation (20-60%) were partially purified by a weak anion exchange chromatography. Fractions collected by stepwise elution were adjust to 50 mL before determination of hydrolytic activity towards alkali treated paramylon. This fraction exhibited high activity towards alkali treated paramylon was mixed with 50 mL of buffer consisting 40 mmol/L Tris-HCl, 1.8 mol/L ammonium sulfate to a total volume of 100 mL, and applied for hydrophobic chromatography. Fractions collected by stepwise elution were dialyzed against 20 mmol/L Tris-HCI buffer (pH=5.5) and fraction eluted by 45-80% elution buffer showed high hydrolytic activity. The results for the partial purification of paramylon degrading enzyme were summarized in Table 22.

Step	Protein (g)	Reducing sugar	Specific reducing	Yield	Purification
		equivalent *	sugar equivalent **	(%)	level
Extraction	11.0	2.47	0.23	100	1
Salt precipitation	2.6	1.85	0.71	74.9	3.1

Table 22 Results of E	. gracilis intracellular	protein	purification.
-----------------------	--------------------------	---------	---------------

Results and Discussion						
IEC	1.4	0.77	0.55	31.2	2.4	
HIC	0.15	0.14	0.93	5.7	4.0	
	0110	0111	0100	011		

*: Reducing sugar equivalent measured by DNS method (section 2.3.5.1), mmol/L;

**: Reducing sugar equivalent /Total protein, mmol/(L·g).

As showed in table 21, the reducing sugars generated by unpurified extract from alkali treated paramylon was 2.47 mmol/L. The results of section 3.1.6 suggest that reducing sugars contributed by 50× cell free extract at 12 d was 2.95 mmol/L. For protein purification, 50 mL *E. gracilis* extract was prepared. While in section 3.1.6, the extraction volume was 1 mL using the same ultrasonication. The decreasing in activity was probably caused by an incomplete cell disruption. The first purification step, salt fractionation, led to an increase in purity of 3 folds, and recovered almost 75% of all the target protein in the original extract. After changing buffer by dialysis, the enzyme sample was purified through an ion-exchange column. The purification level increased to 2.4 folds comparing with the original extract, whereas the yield fell to 31.2 %. The followed hydrophobic chromatography resulted in a purification level of 4.0 folds, while recovering 5.7 % target protein form the extract. The SDS-PAGE (Fig.30) showed that the number of bands decreased after each step.



Fig.30 SDS-PAGE analysis of *E.gracilis* intracellular protein purification. Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); lane 1-5, proteins after partial purification. Lane 1: sell free extraction; lane 2: salt precipitation; lane 3 and 4: IEC elution; lane 5: HIC elution.

The electrophoresis revealed the significant reduction of protein components after each purification step. As showing in Fig.30, Lane 5, elution from hydrophobic chromatography consisted approximately 10 proteins, while generated 0.14 mmol/L reducing product, providing an applicable protein mixture for size exclusion chromatography.

As the last step of purification, the fraction with target activity collected from the hydrophobic chromatography was concentrated by viva-spin before applying a size exclusion chromatography. Fractions were collected by the size of 1.0 mL at the flow rate of 1.0 mL/min of 20 mmol/L Tris-HCl buffer (pH=5.5) for enzyme activity assay. 71 fractions from SEC were collected and their activities responsible for paramylon degradation were analyzed. The results were showed in the following Fig.31.

93



Fig.31 Activity and protein assay of SEC fractions. 100 μ L of the collected fraction was incubated in 1 mL of 1 g/L alkali treated paramylon (pH 5,5) for 16 hours at 37 °C and followed by reducing sugar assay by 3,5-dinitrosalicylic acid (DNS) method (section 2.3.5.1).



Fig.32 SDS-PAGE analysis of SEC fractions.

Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); lane 1-7, SEC fractions.

As showed in Fig.31, the pick of reducing sugar equivalent curve indicated obviously that proteins in these fractions had paramylon degrading activities. The seven fractions generating reducing sugar equivalent higher than 0.3 mmol/L were analyzed by SDS-PAGE. After Coomassie brilliant blue staining, different bands were detected in each fraction as showing in Fig.32. Protein bands in the active fraction obtained after separation by SDS-PAGE were sent to central lab for protein analytic of the Ludwig Maxmilians university of Munich and analyzed by liquid chromatography tandem mass spectrometry (LC-MS/MS). For the protein C in Fig.32, a string of 16 amino acids (NTDGSTDYGILQINSR) were found matching 1,4- β -N-acetylmuramidase (EC: 3.2.1.17) with 12% sequence coverage. This GH family 22 enzyme was confirmed having hydrolytic activity against chitin, which was a insoluble long-chain polymer of N-acetylglucosamine with high degree of polymerization (Aam et al. 2010).

Until now, limited enzymes relating with paramylon degradation were indentified. In 1969, an *E.gracilis* intracellular exo-1,3-glucanase was purified and characterized (Barras and Stone 1969). And recently, an endo- β -1,3-glucanase from *E.gracilis* was indentified, which had a molecular weight of about 40 kDa and showed maximum hydrolytic activity towards laminarin at pH 4.0-5.5, 60 °C after 1 h incubation or at 50 °C after 20 h incubation (Takeda et al. 2015). The protein C in this work was purified for the first which might be another enzyme related with paramylon degrading from *E.gracilis*.

95

3.1.8 Trial of gene amplification of paramylon degrading enzyme purified from intracellular *E. gracilis* proteins

cDNA of *E. gracilis* was synthesized using poly A primer and mRNA as template according to the procedure showed in section 2.3.7. Amplification of the target gene responsible for paramylon degradation protein was performed using degenerate primer initiating from the 3' end of cDNA and the adapter primer from the 5' end. For keeping a low degeneracy, four degenerate primers were designed according to the same protein sequence (-GILQIN-) identified by LC-MS/MS (section 2.3.7.2). However, no clear band was detected by electrophoresis, demonstrating an unsuccessful trial of target gene amplification.

Searching genes in organisms with no available genomic resources could be a low success rate task. Degenerate PCR was one of the methods commonly used for accomplishing this task, either using total genomic DNA or a cDNA library as template. This technique had a low probability but high cost, and its effectiveness depends on various factors. One direct influence might be the quality of cDNA library. In this work, when the total RNA of E. gracilis was isolated for cDNA synthesis, abundant polysaccharide, paramylon, remained in the cell seriously impact the quality of isolated total RNA. The commercial RNA isolation kit used in this work was based on an acid guanidinium thiocyanatephenol-chloroform extraction reagent firstly reported by Chomczynki and Sacchi (Chomczynski and Sacchi 1987). This regent was widely used as lysis buffer before the separation of RNA containing aqueous phase and subsequent RNA precipitation. Polysaccharide in the aqueous phase could physically entrap nucleic acids and entrained into discarded phase during centrifugation. Meanwhile, olligosaccharide precipitate with RNA might interfere with downstream applications (Dellacorte 1994). Both impacts caused by polysaccharide could encumber and fail the subsequent amplification work. Another direct influence was the degenerate primers used in amplifying. In this work, the limited known of the protein sequence restricted the primer designing. In order to keep a low degeneracy, a pool of four degenerate primers for the same site were designed. However, each of the primer had 288-fold degeneracy, which was still a high one. Moreover, length of the part between the identified peptide coding region and poly T tail in cDNA was unknown. All the uncertainties could completely disrupt the PCR amplification.

Although more exploration need to be conducted before confirming protein C (In Fig.32) be paramylon degrading related enzyme, the purification and identification of protein C for the first time may open a new horizon for paramylon depolymerizer searching for the future work.

3.2 Cloning and expression of recombinant β -1,3-glucanase in *P. pastoris*

In this part, recombinant enzymes including four endo- β -1,3-glucanases (*Tr*GH16, *Tr*GH55, *Tr*GH64, *Tr*GH81), one exo- β -1,3-glucanase (*Tr*GH17) from *T. reesei* and one exo- β -1,3-glucanase (*Pp*GH5) from *P. pastoris* were cloned and expressed in *P. pastoris*, Their activities on laminarin and alkali alkali treated paramylon were confirmed by measuring hydrolysis products of reducing groups and glucose, respectively.

3.2.1 Selection of β-1,3-glucanase

The Carbohydrate Active Enzymes database (CAZy) classified glycoside hydrolases according to significant amino acid sequence similarity, reflecting the functional and structural relationships between classification families (Lombard et al. 2014). This is different to the International Union of Biochemistry and Molecular Biology (IUBMB), by which the glycoside hydrolases are grouped on their substrate specificity and occasionally molecular mechanism. Consequently, the β -1,3-glucanases could be placed in different groups. For instance, the exo- β -1,3-glucanases (E.C.3.2.1.58) are grouped in glycoside hydrolase (GH) families 3, 5, 17, 55 and 132, while the endo- β -1,3-glucanases (E.C.3.2.1.39) are located in GH families 16, 17, 55, 64 and 81 (http://www.cazy.org/Glycoside-Hydrolases.html).

The filamentous fungus *Trichoderma reesei* attracted considerable attention of the biofuel industry as a workhorse of plant cell wall degrader. The components of plant cell wall polysaccharide were mainly cellulose, hemicellulose and pectin, which were tightly packed and evolved the ability to resist enzymatic attack from microbes (Brink and Vries 2011). *T. reesei* was able to utilize these polysaccharides due to the strong ability of producing wide range of extracellular hydrolytic enzymes. The increasing knowledge of *T. reesei* genome sequence made it a suitable source for β-1,3-glucan degrading enzymes (Martinez et al. 2008).

In yeasts, glucan degrading enzymes were implicated in glucan metabolism and involved in many processes, such as formation and release of ascospores (Larriba et al. 1995), because of their regid cell walls mainly constructed by β -1,3-glucan and other polysaccharides (Jae 1997). Recombinant expression has been conducted for producing β -1,3-glucanases from yeast strains like *Saccharomyces cerevisiae* (Larriba et al. 1995) and *P. pastoris* (Xu et al. 2006). In the following work, one gene coding for exo- β -1,3-glucanase from *P. pastoris* was also selected for cloning and overexpression.

Selected β -1,3-glucanases for recombinant expression in this work were showed in Table 23.

98

Organism	GenBank ID	Enzyme	GH family
T. reesei	EGR45305.1	Endo-β-1,3-glucanase, <i>Tr</i> GH16	16
T. reesei	EGR47521.1	Endo-β-1,3-glucanase, <i>Tr</i> GH55	55
T. reesei	EGR45483.1	Endo-β-1,3-glucanase, <i>Tr</i> GH64	64
T. reesei	EGR49603.1	Endo-β-1,3-glucanase, <i>Tr</i> GH81	81
T. reesei	EGR46171.1	Exo-β-1,3-glucanase, <i>Tr</i> GH17	17
P. pastoris	CAY69081.1	Exo-β-1,3-glucanase, <i>Pp</i> GH5	5

Table 23 Selected β -1,3-glucanases from *T.reesei* and *P. pastpris*.

3.2.2 Cloning and expression of *Tr*GH16 (EGR45305.1)

Nucleotide and amino acid sequences of *Tr*GH16 are shown in Fig.33. The open reading frame had a size of 864 bp terminated by TAA stop codon, which corresponds to a translation product with 287 amino acids. Predicting by SignalP 3.0, a N-terminal signal peptide existed and the cleavage site lay between residues 19 (a) and 20 (w). No propetide cleavage site was predicted by ProP1.0 Server in downstream of the signal peptide.

Fig.33 Nucleotide sequence and amino acid sequence of *Tr*GH16.

In amino acid sequence, the putative signal peptide was boxed.

By aligning amino acid sequence of *Tr*GH16 with available protein sequences from the GenBank database, it had high identity of 34% with the endo- β -1,3glucanase from *Streptomyces sioyaensis* (GenBank: AAF31438.1) (Hong et al. 2002). Additionally, *Tr*GH16 was identical to enzymes belonging to Glycoside Hydrolase Family 16 including the putative catalytic residues (Glu, Asp and Glu) and jelly roll fold related active sites (Mertz et al. 2009) as showing in Fig.34. Within the GH16 family, two conserved glutamates in the pattern E*X*D*X*(*X*)E played the role of the catalytic domain (Viladot et al. 1998), and catalyzed the glycosylic hydrolysis reaction in a retaining mechanism. At active site, the "R1" glutamate residue acted as a nucleophile to attack the C1 atom of the sugar ring to promote the cleavage of β -1,3-linkage in the absence of water molecules. The "R2" glutamate residue served as the acid to receive the electron and then transfer it back to adjacent water molecule to attack the same C1 atom to complete the hydrolysis (Jeng et al. 2011).

EGR45305	1	MWSPKVAAAILAFVG	LSNAWAPPSYGGFNLVWNDAFAGNggtSPNTG-NWNIITGNL -NVNNELETYSSSTK-	70
AAF31438	1	[33]LAAAVATAATG	PADASAPAPPSGWSQVFLDDFDGAagsSVNTA-NWQFDTGTS[6]NWGTGEVESMTSSTS-	106
BAE02683	1	M-RVVIACCL	VSTALAGVIWQEDFNSLDTS-KWNFETGNN[1]GWGNSELECYTTSSN-	54
ACD93221	1	M-NAFTFPLLLAFCA	FAHGAWVLDWEDEFNGGNLAdRWNFELGCN GWGNNELQCYTDNRGa	60
BAH84971	1	[17]VdGVFTIPA	-ASSLSPRLYRRGNTVFEDSFNSHQLNPK-HWHHEITCW GGGNGEFQMYTPEAA-	78
EGR45305	71	NVQLSgGNTLQLVPWKDS-	S TFGGWTSGRL-ESKYTFTPQAGKVtraEASIRMGSNAqanKQGIWPAFWMLG	140
AAF31438	107	NVSLDgNGDLLITPRRDA-	SGNWTSGRIeTTRTDFQPPAGGKlrvEARLQMPNVTgdaAAGYWPAFWMLG	175
BAE02683	55	NVRVE-NGNLVIEARRE	NVD -GCSFTSGRI-HTRGKFDFKYGTIEARIKLPNLANGLWPAFWMLG	1 16
ACD93221	61	NARQE-DGKLVISAVREWW	rgdGVN PDKEFTSARM-TTKANWLHGKFEMRARLPK-GKHLWPAFWMMP	124
BAH84971	79	NTYIK-NGVLYLKPTFTAd	kfgDD[29]QIPPIMSSKVFSVASITHGRVEVVAKIPKGDWIWPAIWLLP	171
		R1 R2	R3	
EGR45305	141	DVLR-HGGSWPSCGEIDIL	ËQV -NGQPTGHGTLHCDVYPGGICNEGNGIGGPVNFNNPNDFHTWRVE[7]	212
AAF31438	176	APFRgNYQNWPGVGELDIM	IENV -QGLNKTWATMHCGTSPGGPCNETSGIGNLTacpNTTCHSGLHTYTME[5]	249
BAE02683	117	A-ESNTWPDQGEMDIM	IEAG[7]GNVNKEILGTFHWSnnGQHAQYGTNYVAPNDLTTDFHVYKLT	183
ACD93221	125	QN-S-EYGGWPRSGEIDIT	EYR GQRPQQILGTLHFGAAPdnkGDVGTGERDFPIDFSADFHTFGLD	188
BAH84971	172	PGWPwKYGAWPASGEIDIM	ESR[11]TQGVDRVLSTIHYGASPSQHRQQGDSKTSKTGTTWADSFHTYSVD	249
EGR45305	213	WQTETLTWSLDGTNFFQIT	GSRIGNQNVWN NIAHSPLFFILNVAVGGSWPGNPNSATLDGYGAM-MEV	279
AAF31438	250	-SPEAIRFSVDGVTYQTVT	ANHMDAVTWTNATNHGFFVILNVAMGGGFPGAFGGGPTGATEPGHpMVV	316
BAE02683	184	WSQSEIKMFIDDIQYMVFD	DITPL PVFQKNFYVLLNLAVGGNFPNIHDAGAVTAPLPGQ-MLV	244
ACD93221	189	WSPDSMQWLLDDQVYHTES	LQRNFWDGVYNQN[1]SPFDKNFFIILNLAVGGNFFGGEPFDPSESDGWAKNT-FEV	261
BAH84971	250	WTAGHIRMDIDNQPVMAWT	TPSQGYWSYSHQS[11]APFDGKMSLILNVAVGatnGYFQDSWHnTPHAKPWKNNSPtAMM	336
EGR45305	280	GYVAQ YST-	287	
AAF31438	317	DYVQV[115]WQSW[38]	478	
BAE02683	245	DYIKV TQ	251	
ACD93221	262	EYVKK WTWN	270	
BAH84971	337	DFWKS KQQW[21]	366	

Fig.34 Alignment of the *Tr*GH16 sequence.

EGR45305: TrGH16 from T. reesei;

AAF31438: endo-β-1,3-glucanase from *Streptomyces sioyaensis*;

BAE02683: endo-β-1,3-glucanase from *Bursaphelenchus xylophilus*;

ACD93221: endo-β-1,3-glucanase from *Cryptopygus antarcticus*;

BAH84971: endo-β-1,3-glucanase from *Haliotis discus hannai*.

Strictly conserved residues were high-lighted in red, residues predicted as jelly roll fold related active sites were high-lighted with blue background,
conserved catalytic residues Glu (R1), Asp (R2) and Glu (R3) were marked with "#".

As described in the method (section 2.4.1.3), the enzyme gene was cloned by a PCR-based strategy using cDNA as template. A PCR product of 845 bp was obtained (see Fig.35A, Lane 4) containing 825 nucleotide ORF with TAA as stop codon, which encoding 274 amino acids of mature *Tr*GH16 with a predicted molecular mass of 29.84 kDa. The bright band shown in Fig.35A, Lane 4 was ligated into pJET 1.2 cloning vector for further plasmid construction.





A: Analysis of PCR products from cDNA of *T. reesei*. Lane 1: Gene of *Tr*GH55a (EGR 47521.1, 2223 bp); lane 2: Gene of *Tr*GH55b (EGR 48699.1, 2259 bp); lane 3: Gene of glucanase (EGR 47490.1, 798 bp, no further study in this work); lane 4: Gene of *Tr*GH16 (EGR 45305.1, 845 bp); lane 5: Gene of *Tr*GH64 (EGR 45483.1, 1316 bp); lane 6: Gene of *Tr*GH81 (EGR 49603.1, 2576 bp); M: 1 kb DNA ladder (Plasmid Factory, Germany).

B: Recombinant vector screening by restriction endonucleases. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1 to 4, results of four recombinant pAaHBgl-*Tr*GH16 vectors digested by Hind III and Not I.

As in the gene sequence of *Tr*GH16 contained the restriction site of *Xho* I (C^TCGAG), the endonuclease of its isocaudomer *Sal* I (G^TCGAC) was introduced instead for constructing. Consequently, after the fragment being inserted into the vector pAaHBgl between sites of *Xho* I and *Not* I, the recombinant vector could not be linearized by either *Sal* I or *Xho* I. Restriction nuclease *Hind* III site existed in the recombinant vector, locating at 312 bp upstream of the replaced *Xho* I. Two fragments from the reconstructed expressing vector pAaHBgl with the sizes of 7047 bp and 1157 bp should be produced after cleavage with *Not* I when the gene of interested was inserted correctly. As shown in Fig.35B, lane 1 to 4, one band about 1200 bp and one band with the size between 8000 and 6000 bp were generated. The faint bands about 10 kb represented uncut plasmid.

The recombinant vector represented by Fig.35B, lane 1 was named as pAaHBgl-*Tr*GH16 and selected as expression vector for transformation into *P. pastoris* GS115. The transformation was performed by the method described in section 2.4.3. Colonies appeared on the RDB agar plate were cultivated in small scale using 24 square-well plate for protein expression. After methanol induction for 120 hours, supernatants were analyzed by SDS-PAGE and MBTH assay toward laminarin as well as alkali treated paramylon respectively. Fig.36 revealed the results of SDS-PAGE analysis of secreted proteins from six recombinant *P. pastoris* GS115 transformants after methanol induction. In lane 5 and 6, the secreted *Tr*GH16 displayed apparent molecular mass about 35 kDa (highlighted by red line), which was higher than the theoretical mass of 29.84 kDa due to glycosylation predicted by NetNGlyc 1.0 Server.



Fig.36 SDS-PAGE analysis secreting expression of recombinant *Tr*GH16.

Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); lane 1-6, 15 μ L culture supernatant of recombinant *P. pastoris* GS115 after methanol induction for 120 hours in small scale using 24 square-well plate (section 2.4.4.1).

To investigate the enzyme activity, laminarin and alkali treated paramylon cleavage assays were performed. The reducing groups generated by endo- β -1,3-glucanase *Tr*GH16 were determined using MBTH method (section 2.4.5.2). One unit of enzyme activity (IU) was defined as 1 µmol of reducing sugar equivalent released in 1 min. As shown in Fig.37, enzyme activities of sample 5 and 6 (corresponding to lane 5 and 6 in Fig.36) were obviously higher. When alkali treated paramylon was used as substrate, enzyme activities of sample 5 and 6 were 46.6 and 22.9 IU/L respectively. While acting against laminarin, their activities were 49.4 and 38.3 IU/L. This phenomenon was also observed in other recombinant enzymes in the following work, as the better dissolved laminarin making substrate approaching much easier comparing to the alkali treated paramylon. The other samples had activities lower than 5.0 IU/L toward both substrates.



Fig.37 Enzyme activity assay from culture supernatants of selected transformants (*Tr*GH16). Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction in small scale using 24 square-well plate (section 2.4.4.1) were used as crude enzymes for endo- β -1,3-glucanase activity assay. 90 µL enzyme sample was mixed with 90 µL laminarin (2.0 g/L, pH=6.5, indicated by black column) and alkali treated paramylon (2.0 g/L, pH=6.5, indicated by gray column), respectively, and incubated at 37 °C for 60 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

Results of SDS-PAGE and activity assay both indicated that the endo- β -1,3glucanase *Tr*GH16 was expressed and secreted by transforms NO. 5 and 6. The recombinant strain NO.5 was named as *P. pastoris* GS115-*Tr*GH16 and selected for protein expression. Recombinant *Tr*GH16 expression was conducted in 250 mL baffled flask as described in (section 2.4.4.2). Supernatant collected in every 24 h after changing to BMMY medium was assayed for endo- β -1,3-glucanase activity using laminarin and alkali treated paramylon as substrate, respectively.



Fig.38 Recombinant *Tr*GH16 expression in *P. pastoris*. Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction using 24 shake flask (section 2.4.4.2) were used as crude enzymes for endo- β -1,3-glucanase activity assay. 90 µL enzyme sample was mixed with 90 µL laminarin (2.0 g/L, pH=6.5) and alkali treated paramylon (2.0 g/L, pH=6.5), respectively, and incubated at 37 °C for 60 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

Cells were harvested and changed to BMMY medium for methanol induction after precultivation in shake flask (methord in section 2.4.4.2). In this way, no activity was detected at the time point of 0 h. The enzyme activity was firstly measured after 24 hours and increased along the inducting time. In general, recombinant *Tr*GH16 maintained higher activity when acting on laminarin, as paramylon existed in the state of colloidal form, providing less free polysaccharide chains for the enzyme. When acting on alkali treated paramylon, recombinant *Tr*GH16 activity increased rapidly from 6.4 IU/L to 39.1 IU/L, in the second 24 hours of induction, due to the efficient secretion revealed by protein concentration in the supernatant. From time point 72 to 120 h, concentration of total protein kept stable relatively and the increasing of enzyme activity slowed down, achieving the highest activity of 57.8 IU/L after inducing for 120 h. The data suggested that trend of enzyme activity increasement identically when using laminarin as substrate, and represented higher activities. At time point of 120 h, recombinant *Tr*GH16 showed the highest activity of 77.0 IU/L with 0.26 g/L protein in the supernatant.

Revealed by the results above, a putative endo- β -1,3-glucanase *Tr*GH16 was successfully cloned and expressed by recombinant *P. pastoris* using α -factor as leading signal. The glycosylic hydrolase domain of this enzyme contained a conserved E*X*D*X*(*X*)E motif of GH16 members, and the catalytic importance of this conserved motif has been reported by mutational analyses of laminarinase (Krah et al. 1998). Even when the supernatant was used directly as crude enzyme, it efficiently catalysed the hydrolysis of both substrates in this work. In comparison to alkali treated paramylon, larminarin was hydrolysed at a relatively higher rate. Although the genes encoding endo- β -1,3-glucanase in GH 16 have been studied from a variety of hosts for decades, this work expressed *Tr*GH16 (GenBank: EGR45305.1) in *P. pastoris* for the first time and confirmed its activity towards the unique substrate paramylon.

3.2.3 Cloning and expression of TrGH55 (EGR47521.1)

The gene of *Tr*GH55a had an open reading frame of 2277 bp coding for 759 amino acid sequence (Fig.39). Predicting by SignalP 3.0, a N-terminal signal peptide of 17 residues existed and the cleavage site lay between positions 17 (a) and 18 (t). Additionally, the processing site for Kex2 (Lys-Arg) lay at positions 33-34, which would cleave the proprotein after preproprotein entering the secretory pathway leading by the signal peptide.

131 acgtcaacge teeergege ttegeecetg acetegatgg caactteaac tateaggtet accagacegt caaceeggga gaeggagetg etetgeageg ageeattace agegatggea geggetegeg h v n a p r g f a p d l d g n f n y q v y q t v n p g d g a a l g r a i t s d g s g s 261 ccatteccag tggttegeat egeagecaag agtegtgtae atceegeeeg gaacatacae tetetegeag aegetgegat teaacaeega caeegtgttg atgggegatg ceaecaaeee geecateate 391 aaggeegeeg eeggettete gggegaeeag ateetegtea geggeeagga eeeaacaee aaegaaaagg gggagetgte etttgeegtg teeeteaga aegteateet egaeaegaee geeateeeg k a a a g f s g d q i l v s g q d p t t n e k g e l s f a v s l k n v i l d t t a i p 521 geggedaeca gtteaeegee etetggtggg gagtegeecea ggeegeteag etgeagaaeg teaagateae eatggettee teceagaaeg geaaeggtea eaegggeate eggatgggee geggetegae 651 geteggeeta geegaegtee gegtegageg tggeeagaat ggeatetggg tegaeggeea eeageagget geetteeaa acatetaett etteeagaae acegteggea tgeteateag eggeggeaae 781 acctttagca tettetegte caegtttgae aegtgeggaa egggeattte eaacaeggge ggtgeteett ggategeeet ggtagaegee aagtetatea atteeggggt taegtttaeg aegaegggat 911 tteetteett ettgategag aacetgacea aggataegae gteteeggte gtattgete gtgagtaaa ettggttggg getteegae gtatteett ggaaataegg ttgggagtaaa 1041 ccccatctac ggcgaggtga gateteaaaa cacgagacet ggtgetetgg cccctggtgg gagatateea tatgteeege cgccgacata cgccgatgte ecegtetetg gettettgaa cgtcaaggac gryp 1171 ccagegeaga acggaaacag agtegteaag ggggacaaca cegtegaega ageageeaeg etgaatgeea teetggeeet egeegeeage eagaacaaga ttgeetaett eeetttgga aagtateggg pa q n g n r v v k g d n t v d e a a t l na i l a l a a s q n k i a y f p f g k y r 1301 togacagcac gototttatt cotocggggt cocgcatogt gggcgaggoc tggtcaacca toacoggcaa cggccagtto ttoaagaacg agaacagooc ggagcocggt gtgccgtog gcoggocogg v d s t l f i p p g s r i v g e a w s t i t g n g q f f k n e n s p q p v v a v g r p 1431 ggacgtegge gtegeacaga tecaggaeat gegetteace gttteegaeg tteteceegg egecategte atgeagttea acatggeegg eegeaaceea ggegaegtgg eeettegaa etegetegte g d v g v a q i q d m r f t v s d v l p g a i v m q f n m a g r n p g d v a l w n s l v 1561 accgtggggg gcacgcggg tgetteegee etgaceaeg ectgtaegaa eeegggeaae gagtgeaagg gegegtteet gggeateeae ateaecaaga actegtegge gtaeateeag aaegtetgga 1691 actgggtgge ggaccacate geogagaget teggeggegg etecteeate geoggeaagg geggegtget ggtegaage tecaaggega egtggetgga geoggeggeggetgta n w v a d h i a e s f g g g s s i a g k g g v l v e s s k a t w l v a l g s e h w w l 1821 ccageteaac etgegeaacg ceaacaacgt egtegtetee atgeteeagt eegageacaa etaegageag ggegeeaacg egeageaggt egtgeeegee eeetgggteg ceaacgtgga egeetgggge 1951 gateceaact tttcgtggtg cageggegge gacaageggt geaggatggg etttggeaac tacateaatg ggggateeaa catttataeg tatgeetetg ettetgagge gttttteage gggeeeggg 2081 agggctgttc tcagtttcag tgccagcaaa ccatgcactg gattgccagc acgcccagta atctccaagc ctttggactg tgctccaagg acccgttaa cacgttgcgg ctcggggacg gcacattaat q g c s q f q c q q t m h w i a s t p s n l q a f g l c s k d s v n t l r l g d g t l 2211 caacacggcc aacggcttca ctggctcgtg gccgggtggc ggtggtgatg tcggtcgtta taccgtctaa

Fig.39 Nucleotide sequence and amino acid sequence of *Tr*GH55.

In the coding region, the putative signal peptide was boxed, and the processing site for Kex2 endoprotease was indicated by arrow.

By aligning amino acid sequence with available protein sequences from the GenBank database, the protein had 80% sequence identity to the characterized

GH 55 enzyme from *Trichoderma harzianum* CECT 2413 naming *Bgn*13.1 (GenBank: ACE81432.1).

EGR47521	1	MLKLTTIFTLLLGAASATPTPSPAANDEGITKRATSFWYPNMDHVNAPRGFAPDLDGNFNYQVYQTVNPGDGAALQRAIT	80
ACE81432	1	MLKLTALVALLLGAASATPTPSPPASDEGITKRATSFYYPNMDHVNAPRGFAPDLDGDFNYPIYQTVNAGDGNALQNAIT	80
EGR47521	81	SDGSG-SRHSQWFASQPRVVYIPPGTYTLSQTLRFNTDTVLMGDATNPPIIKAAAGFSGDQILVSGQDPTTNEKGELSFA	159
ACE81432	81	TDGKGGSRHPQWFASQPRVVYIPPGTYTISKTLRFNTDTILMGDPTNPPIIKAAAGFSGDQTLISAQDPSTNEKGELSFA	160
		R3 R4	
EGR47521	160	VSLKNVILDTTAIPGGNQFTALWWGVAQAQLQNVKITMASSQNGNGHTGIRMGRGSTLGLADVRVERGQNGIWVDGHQQ	239
ACE81432	161	VAIKNVVLDTTAIPGGNSFTALWWGVAQAAHLQNVRITMSSSSGGNGHTGIRMGRCSTLGLADVRVERGQNGIWIDGHQQ	240
EGR47521	240	AAFHNIYFFQNTVGMLISGGNTFSIFSSTFDTCGTGISNTGGAPWIALVDAKSINSGVTFTTTGFPSFLIENLTKDTTSP	319
ACE81432	241	ASFHNIYFFQNTIGMLISSGNTFSIFSSTFDTCGTAFPTLAGSPWIALIDAKSINSGVTFTTNQFPSFMIENLTKDNGTP	320
EGR47521	320	VVIARGSTLVGASTHIDTYSFGNTVGRNPIYGEVRSQNTRPGALAPGGRYPYVPPPTYADVPVSGFLNVKDPAQNGNRVV	399
ACE81432	321	VVVVRGSTLVGASSHVNTYSYGNTVGRNPTYGDVTSSNTRPSALAPGGRYPYVAPPTYGDLPISSFLNVKDPAQNGNRQV	400
EGR47521	400	KGDNTVDEAATLNAILALAASQNKIAYFPFGKYRVDSTLFIPPGSRIVGEAWSTITGNGQFFKNENSPQPVVAVGRPGDV	479
ACE81432	401	KGDNTINEADTLNAILELAASQNKVAYFPFGKYRVDSTLFIPKGSRIVGEAWATITGNGNFFKNENSPQPVVSVGRAGDV	480
EGR47521	480	GVAQIQDMRFTVSDVLPGAIVMQFNMAGRNPGDVALWNSLVTVGGTRGASALTNACTNPGNECKGAFLGIHITKNSSAYI	559
ACE81432	481	GIAQLQDLRVTTNDVLPGAILVQFNMAGNNPGDVALWNSLVTVGGTRGAQALANACTNNSNECKGAFIGIHVAKGSSPYI	560
		R2 R1	
EGR47521	560	QNVWN-WVADHIAESFGGGSSIAGKGGVLVESSKATWLYALGSEHWWLYQLNLRNANNVVVSMLQ4	634
ACE81432	561	QNVWELGLRDHIAENFSGGTSHRRERWNFGPIRRNATCLYPIGSGHWWLYQLNLHNAANVVVSLLQAETNYHQGANT	637
EGR47521	635	QQVVPAPWVANVDAWGDPNFSWCSGGDKRCRMGFGNYINGGSNIYTYASASWAFFSGPGQGCSQFQCQQTMHWIASTPSN	714
ACE81432	638	QQIPPAPWVANVGTWGDPDFSWCNGGDKRCRMGPANFINGGSNIYTYASAAWAFFSGPGQGCAQFECQQTIHWIASTPSN	717
EGR47521	715	LQAFGLCSKDSVNTLRLGDGTLINTANGFTGSWPGGGGDVGRYTV 759	
ACE81432	718	LQAFGLCSKDSVNTLRLGDGTFINTQNGYTGGWTPGGGDVARYTT 762	

Fig.40 Alignment of the *Tr*GH55 sequence.

EGR47521: *Tr*GH55from *T. reesei*;

ACE81432: Bgn13.1 from T. harzianum CECT 2413.

Strictly conserved residues were high-lighted in red, conserved catalytic residue R1 (Glu) was marked with "#". Strictly conserved residues related in catalytic reaction R2 (Asp), R3 (Ser) and R4 (Gln) are framed.

The residue R1 (Glu) was concluded as a catalytic site (Ishida et al. 2009). Residue R2 (Asp) had a primary role in positioning the substrate, R3 (Ser) and R4 (Gln) were reported both important, but not essential catalytic related residues in the latest research (Bianchetti et al. 2015). The Bgn13.1 was confirmed to be an endo- β -1,3-glucanase by experimental evidence for the first time (La-Cruz et al. 1995). It contributed enzymatic activity on periodate oxidized laminarin, which could not be hydrolyzed by exo- β -1,3-glucanase, but showed no ability of hydrolyze laminaribiose. Based on the high identity with Bgn13.1, *Tr*GH55 in this work was predicted as an endolytic mode β -1,3-glucanase.

Gene of putative endo- β -1,3-glucanase *Tr*GH55 was cloned from cDNA of *T. reesei.* A PCR product of 2223 bp was obtained, which contains a 2199 nucleotide ORF terminated by a TAG stop codon, encoding 732 amino acids of mature protein with a predicted molecular mass of 77.94 kDa. In Fig.35A, lane 1, the bright band between 2000 and 2500 bp showed the product of gene amplification. The faint bands in this lane were probably nonspecific PCR products, nevertheless their existence had no influence on isolation of the target band from agarose gel. The purified product was then ligated into pJET 1.2 cloning vector for further plasmid construction.

Gene of *Tr*GH55 was insert to the vector pAaHBgl by ligation between sites *Xho* I and *Not* I, and named as pAaHBgl-*Tr*GH55 in the following work. Two fragments with the sizes of 7395 bp and 2223 bp should be produced by digestion with *Xho* I and *Not* I when the gene of interested was inserted correctly.

110



Fig.41 Analysis of pAaHBgl-TrGH55 construction.

M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1 to 3, fragments of recombinant plasmids digested by *Xho* I and *Not* I.

As shown in Fig.41 from lane 1 to 3, two fragments were generated after restrict digestion of endonuclease, with lengths between 2000 to 2500 bp and 8000 to 6000 bp in each lane, respectively. The results revealed by DNA electrophoresis meet the requirements mentioned above, suggesting the recombinant plasmid construction was conducted correctly.

The recombinant pAaHBgI-*Tr*GH55 shown in Fig.41 Lane 1 was selected afterward for transformation into *P. pastoris* strain GS115. Recombinants appeared on the selective plate were picked and cultivated for recombinant protein secretion in small scale cultivation using 24-well plate. After methanol induction for 120 hours, supernatants were used as crude enzyme samples for protein and endo- β -1,3-glucanase activity assay. The Fig.42 below was the SDS-PAGE of secreted protein from the recombinant strains.



Fig.42 SDS-PAGE analysis secreting expression of recombinant *Tr*GH55.

Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); lane 1-6, 15 μ L culture supernatant of recombinant *P. pastoris* GS115 strains after methanol induction for 120 hours in small scale using 24 square-well plate (section 2.4.4.1).

The existence of an evident protein band in lane 3 (highlighted by red line) with a molecular mass between 70 and 100 kDa, which met the theoretical mass of 77.94 kDa, confirmed the secretion of recombinant *Tr*GH55 in *P. pastoris*. Faint protein bands with the same molecular mass in lane 5 and 6 revealed the potential existences of target enzyme. However, it was not difficult to estimate the bands from the gel. For the bands in lane 5 and 6, it was hard to confirm efficient expressions of the potential enzyme from respective recombinants. To investigate the *Tr*GH55 activity in these samples, cleavage assays against laminarin and alkali treated paramylon were performed and enzyme activities were determined by MBTH method (section 2.4.5.2).



Fig.43 Enzyme activity assay from culture supernatants of selected transformants (*Tr*GH55). Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction in small scale using 24 square-well plate (section 2.4.4.1) were used as crude enzymes for endo- β -1,3-glucanase activity assay. 90 µL enzyme sample was mixed with 90 µL laminarin (2.0 g/L, pH=6.5, indicated by black column) and alkali treated paramylon (2.0 g/L, pH=6.5, indicated by gray column), respectively, and incubated at 37 °C for 60 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

The only sample (Lane.3 in Fig.42) containing an evident protein band with the right molecular mass had an activity of 28.9 IU/L on alkali treated paramylon and 43.2 IU/L on laminarin. The other samples selected all have activity about 10 IU/L on ether alkali treated paramylon or laminarin, indicating the recombinant endo- β -1,3-glucanase *Tr*GH55 was not or inefficiently expressed. Combining the results from SDS-PAGE, the recombinant *P. pastoris* GS115 strain NO.3 successfully secreted *Tr*GH55 as main extracellular protein. And the recombinant enzyme had activity on both laminarin and alkali treated paramylon, providing reducing ends from the substrates. Therefore, this

recombinant strain was selected for further research and named as *P. pastoris* GS115-*Tr*GH55 in this work. Recombinant *Tr*GH55 expression was performed in 250 mL baffled flask. Supernatant collected in every 24 h since methanol inducing starts was assayed for endo- β -1,3-glucanase activity using laminarin and alkali treated paramylon as substrate respectively.



Fig.44 Recombinant *Tr*GH55 expression in *P. pastoris*. Supernatants of selected recombinant P.pastoris GS115 transformants after methanol induction using 24 shake flask (section 2.4.4.2) were used as crude enzymes for endo- β -1,3-glucanase activity assay. 90 µL enzyme sample was mixed with 90 µL laminarin (2.0 g/L, pH=6.5) and alkali treated paramylon (2.0 g/L, pH=6.5), respectively, and incubated at 37 °C for 60 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

During the methanol induction, endo- β -1,3-glucanase activity detected by determining reducing ends generated from laminarin increased from none to 125.1 IU/L in 120 hours. Comparing to 42.2 IU/L detected from 24-well plate

cultivation, the activity of recombinant enzyme achieved a threefold enhancement. A similar trend appeared when acting on alkali treated paramylon as substrate. After 120 hours inducing, the enzyme activity was 81.2 IU/L with a total protein concentration of 0.26 g/L in the supernatant. As discussed before, the recombinant *Tr*GH55 showed higher activity toward laminarin, as the well dissolved laminarin was probably more accessible to enzymes than paramylon.

According to the results of enzyme assay and protein electrophoresis, the conclusion can be made, that a putative endo- β -1,3-glucanase from *T.reesei* namely *Tr*GH55 was successfully cloned and secreted by recombinant *P.pastoris* using α -factor as leading signal. The recombinant *Tr*GH55a is predicted to be a GH55 family enzyme, because of the high amino acid sequence similarity with GH55 family member *Bgn*13.1 from *T. harzianum*. The enzyme *Bgn*13.1 expressed in strawberry was confirmed as an endo- β -1,3-glucanase by hydrolyzing laminaring. The recombinant enzyme had the highest activity of 2.4 mmol/L revealing by glucose equivalent produced after acting on laminarin for 20 hours (Mercado et al. 2015). While in this work, the *Tr*GH55 had the reducing sugar equivalent of 937.5 µmol/L. Although the activity was lower than *Bgn*13.1, the *Tr*GH55 was secreted by *P. pastoris* for the first time since genome sequence of *T. reesei* was published in 2008 (Martinez et al. 2008), and the activity could be enhanced by gene codon and express conditions optimization in further work.

3.2.4 Cloning and expression of *Tr*GH64 (EGR45483.1)

Nucleotide and amino acid sequences of *Tr*GH64 from *T. reesei* are shown in Fig.45. The open reading frame had a size of 1350 bp terminating by stop codon TGA, which corresponds to a translation product of 449 amino acids. Predicting by SignalP 3.0, a N-terminal signal peptide existed and the cleavage site lay between positions 16 (a) and 17 (a). Additionally, basic residues Thr (t) and Arg (r) at positions 23-24 were predicted as propetide cleavage site by ProP1.0 Server, indicating the mature protein started from residue at position 33 (v).

Fig.45 Nucleotide sequence and amino acid sequence of TrGH64.

In the coding region, the putative signal peptide was boxed, and the predicted propeptide cleavage site was indicated by an arrow.

By aligning amino acid sequence with available protein sequences from the GenBank database, the protein showed 25% similarity to laminarinpentaoseproducing β -1,3-glucanase naming *LPHase* from *Streptomyces matensis* (GenBank: BAA34349.1) (as showing in Fig.46). The *LPHase* was confirmed to be a GH family 64 member, endolyticly cleaving β -1,3-glucan chain into specific pentasaccharide oligomers in a direct displacement mechanism. In the catalytic domain, the conserved Glu acted as a porton donor to the glycosidic oxygen, and the conserved Asp facilitated a base assisted nucleophilic attack by a water molecule from the opposite side of the sugar ring, while the side chains of Thr, Asn and Trp acted for polysaccharide chain binding (Wu et al. 2009). When a polysaccharide chain of β -1,3-glucan diffused into the catalytic domain, the reducing end would be positioned bound in the region by Trp(R3), Asn(R4) and Thr(R5). Hydrolysis of the β -1,3-glycosidic bond from the end of the chain was performed by Glu(R1) and Asp(R2), generating pentasaccharide as product.

EGR45483	1	MRTFTLLTALMGAVSAAPSLITRVATPFSVASPGGIDDISVTGHNILNGTYHSPATTGPNKFANQANPKSGHLPIKLVNS	80
BAA34349	1	MLRTLRRRVTA VALGLATALGGGW LAAGVPSPA HAAVPAT IPLTITNN	48
EGR45483	81	FGGGH-VNAYIAGLD-SENRVVFVKGDGSLLYPSSGGSEIPVPVSDEIKISLPAKGKSLTVTVPIVITSARIYFAEGTLP	158
BAA34349	49	SGRAEQIHIVNLGTELSSGRQGWADASGAF-HPWPAGGNPPTPAPDA-SIPGPAPGRSTTIQIPKF-SGRIVFSYGRKM	124
		R1 R2 # P5 P3P4 #	
EGR45483	159	FFIVKIPTGDGLVQPSPANLKDPSAGINWGFVELSYTEGKAVWANISYVDFVGMILSMSLSSTDGSTQVTRGLHSYALGE	238
BAA34349	125	EF-RLTTG-GLVQPAVQNPTDPNRDILFNWSEYTLNDS-GLWINSTQVDMFSAPYTVGVRRGDGTTLSTGKLRPGGYNG	200
EGR45483	239	ICKDLAKQTAKDGFPWANLCIRNSEGRLLRALAPGDYSDIDKNGFK-NYWKSYVDKVWDHYSSNTLTINTQSSAGKVEC-	316
BAA34349	201	VFNALRGQSGGWANLIQTRSDGTVLRALSP-LYGVETGALPASVMDDYINRVWNKYTGTDLIVTPFADRPDVRYT	274
EGR45483	317	-KVKDGKLQ-CDGDNR-SYSKPSANDIWGCNSGPFAIKAGDNELHKAVVPRLCAAFVRATLLIPGGDVQPGVSRSKYY	391
BAA34349	275	GRVSGGVLRFTDGSGAVVTTFQKPDASSVFGCHR-LLDAPNDQVRGPISRTLCAGFNRTTLL-ANPHQPDRSAAGFY	349
EGR45483	392	KTNPANHYSRLVHHYEVDGRGYAFPYDDVNPNDKENASGTLATGTPEKLTVYIGAPPP 449	
BAA34349	350	QEPVTNHYARIIHAHMADGKAYGFAFDDVGHHES-LVHDGDPRGASLTLD-PFD 401	

Fig.46 Alignment of the *Tr*GH64 sequence.

EGR45483: TrGH64 from T. reesei;

BAA34349: LPHase from S. matensis.

The conserved catalytic residues are R1 and R2, substrate binding residues were maked as R3, R4, and R5.

Gene of putative endo-β-1,3-glucanase *Tr*GH64 was cloned by a PCR-based strategy using cDNA as template. A PCR product of 1316 bp was obtained containing a 1299 nucleotide ORF terminated by a TAG stop codon, which encoding 432 amino acids of mature protein with a predicted molecular mass of 46.35 kDa. As shown in Fig.35A, lane 5, the bright band with a size between 1000 and 1500 bp was obtained as the main product. The purified PCR product was then ligated into pJET 1.2 cloning vector for further plasmid construction.

Recombinant plasmid pAaHBgI-*Tr*GH64 construction was performed as described in (section 2.4.2). Gene of *Tr*GH64 was ligated into the vector pAaHBgI between sites *Xho* I and *Not* I. Two fragments with the sizes of 7395 bp and 1316 bp should be produced by endonucleases digestion of *Xho* I and *Not* I when the gene of interested was inserted correctly.



Fig.47 Analysis of pAaHBgI-*Tr*GH64 construction. Recombinant vector pAaHBgI-*Tr*GH64 screening by digestion with restriction endonucleases. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1-3, results of the recombinant plasmids digested by *Xho* I and *Not* I.

As shown in Fig.47 Lane 1 to 3, two evident fragments were obtained by *Xho* I and *Not* I digestion, the band between the 1000 and 1500 bp was the gene

coding *Tr*GH64 and the band with the size between 6000 and 8000 bp was linearized vector pAaHBgl. The recombinant vector shown in lane 1 in Fig.47 was selected as expression vector for further transformation into *P.pastoris* strain GS115. Performed as mentioned in the methods 2.4.4.1, transformants were picked from selective agar plate and induced by methanol for recombinant protein expression in 24-well plates. Supernatants from the small-scale cultivation were used as crude enzyme samples for SDS-PAGE and activity assay. Only one positive transformant was obtained demonstrating by the SDS-PAGE result of secreted proteins. In Fig.48 lane 5, a distinct band (highlighted with red underline) shows a secreted protein with a molecular mass between 40 and 55 kDa, which meets the predicted molecular mass of target enzyme *Tr*GH64 (46.35 kDa). No band with similar molecular mass appeared in other samples.



Fig.48 SDS-PAGE analysis secreting expression of recombinant *Tr*GH64. Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); lane 1-6, 15 μ L culture supernatant of six recombinant *P. pastoris* GS115 strains after methanol induction for 120 hours in small scale using 24 square-well plate (section 2.4.4.1).

Results in Fig.49 further confirmed the success of cloning and expression of recombinant *Tr*GH64 in *P. pastoris*. To measure the enzyme activity, alkali

treated paramylon and laminarin were used as substrate for recombinant enzyme hydrolyze assay, respectively. Enzyme activity was analyzed by determining reducing ends using MBTH method.



Fig.49 Enzyme activity assay from culture supernatants of selected transformants (*Tr*GH64). Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction in small scale using 24 square-well plate (section 2.4.4.1) were used as crude enzymes for endo- β -1,3-glucanase activity assay. 90 µL enzyme sample was mixed with 90 µL laminarin (2.0 g/L, pH=6.5, indicated by black column) and alkali treated paramylon (2.0 g/L, pH=6.5, indicated by gray column), respectively, and incubated at 37 °C for 60 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

Suggested by hydrolytic activity assay toward these two substrates, only the supernatant of transformant NO.5 (Fig.48 lane 5) had definite higher recombinant *Tr*GH64 activity than the other samples. As shown in Fig.49, recombinant *Tr*GH64 demonstrated activity of 33.8 IU/L towards alkali treated paramylon, and 40.1 IU/L when acting on laminarin. While the other samples

containing only background proteins of *P. pastoris* in SDS-PAGE, these samples all showed activities lower than 5.0 IU/L towards either substrate. Based on these results, it's confirmed that the active recombinant endo- β -1,3-glucanase *Tr*GH64 was secreted. This recombinant strain was named as *P. pastoris* GS115- *Tr*GH64 and cultivated in 250 mL baffled flask for further research.

During methanol induction, supernatant was taken every 24 hours and used as crude enzyme for hydrolytic assay. The total protein concentration and enzyme activity results are shown in Fig.50 below.



Fig.50 Recombinant *Tr*GH64 expression in *P. pastoris*. Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction using 24 shake flask (section 2.4.4.2) were used as crude enzymes for endo- β -1,3-glucanase activity assay. 90 µL enzyme sample was mixed with 90 µL laminarin (2.0 g/L, pH=6.5) and alkali treated paramylon (2.0 g/L, pH=6.5), respectively, and incubated at 37 °C for 60 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

Since the methanol inducing started, proteins in the supernatant accumulated and achieved a concentration of 0.34 g/L in 120 hours. Consequently, activity of endo- β -1,3-glucanase increased along the protein accumulation. When use laminarin as substrate, the enzyme activity rose rapidly to 100.7 IU/L in the first 72 hours, and increased steadily to 112.4 IU/L at time point 120h. The hydrolysis action of *Tr*GH64 towards β -1,3-glucan started by positioning a free chain of the polysaccharide and breaking the glycosidic bond by interactions of residues including Glu and Asp in the catalytic domain, afterwards. Therefore, enzyme activities toward alkali treated paramylon were relatively lower due to the difficult of accessing between enzyme and substrate. After methanol induction for 120 hours, the recombinant *Tr*GH64 had an activity of 79.0 IU/L on paramylon, which is about 30% lower than acting on laminarin.

Although 296 proteins were predicted relating to glycoside hydrolase family 64 enzymes by now according to sequence similarity (Davies and Sinnott 2008). The GH family β -1,3-glucanase was firstly confirmed to hydrolysis substrate by inverting mechanism in 2001 (Nishimura et al. 2001). The crystal structure was resolved in 2009 for the first time (Wu et al. 2009), and essential residues related to catalytic activity were firstly reported in 2011 (Shrestha et al. 2011). Including the *LPHase* from *S. matensis*, only four GH family enzymes were characterized by now. The other three characterized enzymes, *glu*B from *Lysobacter enzymogenes* Strain N4-7 (Palumbo et al. 2003), *LPHase* from *Streptomyces matensis* DIC-108 (Palumbo et al. 2003), and an unnamed glucanase from *Oerskovia xanthineolytica* (Shen et al. 1991), were all expressed in *E. coli.* Based on the experimental evidences and current knowledges, this work conducted the successful recombinant expression of *Tr*GH64 from *T. reesei* in *P. pastoris* for the first time.

2.3.5 Cloning and expression of *Tr*GH81 (EGR49603.1)

Nucleotide and amino acid sequences of *Tr*GH81 (GenBank: EGR49603.1) are shown in Fig.51. The open reading frame was 2592 bp and terminated by TAA stop codon, corresponding to a translation product of 863 aa. Predicting by SignalP 3.0, a N-terminal signal peptide existed and the cleavage site was between positions 18 (a) and 19 (I). No propetide cleavage site was predicted by ProP1.0 Server downstream of the signal peptide. Therefore, the putative mature enzyme from residue Leu (I) to Ser (s) from position 19 (I) to 863 (s) was cloned.

1 atgctggcct ctcttgtgaa cctcgtcatc gcctcggcgc ttggcgttgc ggctctgccc aatggcgcaa agctcagcaa cattgccgga cgcggtctgt tcgccccgat ctcgacgag aacccggcgc m l a s l v n l v i a s a l g v a a l p n g a k l s n i a g r g l f a p i s t s n p a 131 gcatetegg eggcaccag gccacegge cogatggae gcceqtage tegttitte cagecegtage etceptic cegaceaet eetgytage etcytage gecaceggeae 261 cgcctcgggc ccctttccct acgagtcgca gctcgacggc ctgggcatca actttggcgt gagcaacagc cgccagttcg acggcacgtc catcaagcag ccgacggaga acgactggcg cgcgggcttc t a s g p f p y e s q l d g l g i n f g v s n s r q f d g t s i k q p t q n d w r a g f 391 geogageaet egggegeett tyceaaecae aaggegaegg eetttgaeae geaeteggte aeggtgeagt aetteeaggg eggegegage atgaeetege egetgattee eggategeeg taeattaege 521 tgcagtacca ggctgccacg ccgctgctga cgtcgaggaa cggggggcatt gcgtcgttta acggacagaa tctctcgaat ggacagagtg tcactgctac tggeacctcg ttcaccgtcg tcgacaccac 651 cggcaccace tacgtgatet acgecetete gtecatetee etgactgeca cggccaccaa cagegeecag ggcactatea aggecaccgg caegtacaac ggegtgetee geetegteag geteaceag 781 gccagccaca aggcgctcct cgaccagcac tacaccgtct accccacggg cgtcggcctc gactacagct tcaccaccac caccggcag ctcatcttca actacaacac cgtgggcgac ggcagccagc a s h k a l l d q h y t v y p t g v g l d y s f t t t t g t l i f n y n t v g d g s q 911 tgstgatgst gasstggscg sacsassgt tgstststgs gggsgssaas sagssggssa sttsgtsst gggstasst accassagg gatggatgta socsatsats ggsaassagt ggaagstsst 1041 gtaccagetg agrageatea ectggaacee tectegegeg etegaetee eggegetete eagggtetee agraegaat tggtetgete eggaatetea etecteetg geceaeaga 1171 ttttactact ggggaggcag cetegeceget caggetegte ttgeceteat tgecegagget gteggeegea eegaceteat eeecaeggtg acaaaetaee teaagaeeag etteeagaae tggtteage f y y w g g s l a a q a r l a l i a e a v g r t d l i p t v t n y l k t s f q n w f t 1301 ccagcaccgg cgcgtctccc gcgtacgaaa cctcttgggg cggcgtcatc gacaaggccg gcgccaccaa ctcgggcatc gactttggca acggctacta caacgaccac cacttccact acggctactt 1431 titgtacgtg gocgocgta titgccaagta cgacgocaac tggctggcg agcataagga cittataac tggttggcc gagacatat caaccegtg cocaacgac cetactice cgtcaccego fl v v a a v i a k v d a n v l a g h k d f i n v f a r d i i n p s p n d p v f p v t r 1561 tgccgcgact ggttcgccgg ccactcgtgg gcctcgggca tcgccaacgg ggccggcagc cgagaccagg agtcgaccgg cgaggccgtc aacggctact acggcgcgct gctcgggcc acggtcggc c r d w f a g h s w a s g i a n g a g s r d g e s t g e a v n g v v g a l l w a t v a 1691 tgtcgcagga ctacgtcaac tacgcccggc tgctcgtcgc cacggagcag caggggggcc aggtctactg gcacctgtac ccgcagcaga gccagacaga ccccaacaac ccgtacccgg agccggcggt 1821 gcgcaacctg gtgaccatgg gcaacgtcga ggactggcag tcgggcgcct ggctgttctg gggcaaccag aagagcgaga ttgcggccat ccagatgctg cccgtgacgc ccatcaacga ggtgctgtac qnq 1951 gacgccagt gggtcaacaa tgtgtggtg tatgcccaga agagattt cgaccogt a stogccagt actggcgtc gtcatgat gcggctact ccaacgccaa cctcagat gctgccgct 2081 ggagegeeaa eetgaceae etgageteeg gaaacaeett eteeaaegag etettettea teggeaeeeg geeeaaeete ageggeeage eetaetgegg eteeaaette eegeagaaee egtaeggeaa w s a nltt w g s g n t f s n e l f f i g t r p n p s g q p i c g s n f p q n p y g 2211 cttcaagatc cagteggeca cgaegggeca gtgggtegte gegtegaegg egtegteeaa eetegtggee tegggetege aggeeagege eggegtette ateagetegt acaegeecaa egegggeaac 2341 ctcaagctga cgagcaacaa ccagttcgtg acggcggacc agtcgggcaa etttgcgetg caggcggccc gcgccacgge etcgteetgg gaggtgttta ecatecgeca aaaggtegge gcegeggegg 2471 gcgtgtacac gatcaagget gggagcaacg gcaagtgggt gacgetegg teggaegggt egttgatcaa caatggggeg acggaggett eggetgetgg gttcaagttt gtecagtegt aa

Fig.51 Nucleotide sequence and amino acid sequence of *Tr*GH81.

In the coding region, the putative signal peptide was boxed.

Comparison of the amino acid sequences of *Tr*GH81 with characterized protein *Rm*Lam81A from *Rhizomucor mieher* (GeneBank: AGV00786.1) resulted a similarity of 24%. The enzyme *Rm*Lam81A was an endo-β-1,3-glucannase of GH81 family. For the first time, crystal structure of *Rm*Lam81A was reported consisting three distinct domains: domain B stabilized the structure of the protein, as well as the core cleft formed by domain A and C (Zhou et al. 2013). As shown in Fig 52, strictly conserved residues in the sugar-binding motif were framed in blue. Among them, Glu (R1) and Glu (R3) were predicted catalytic residues. R1 and R3 were likely a proton donor and a basic catalyst respectively, and residues Ser (R2), Tyr (R4) and Try (R5) contributed to position and ionization state of catalytic residues. For instance, residue R2 stabilized the position of R3 by forming a hydrogen bond to the backbone N atom (Zhou et al. 2013). These residues were highly conserved among GH family 81 emzymes.

EGR49603	1	MLASLVNLVIASALGVAALPNGAKLSNIAGRGLFAPISTSNPAGISGGTTATGADGAPVSSFFAGLKPPFPTNSWWASYA	80
AGV00786	1	MRFQVIVAAATITMITSYIPGVASQSTSDGDDLFVPVSNFDPKSIFPEIK	76
EGR49603	81	ATPGNG-TASGPFPY-ESQLDGLGINFGVSNSRQFDGTSIKQPTQNDWRAGFAEHSGAFANHKA	142
AGV00786	77	YPSADNLAPTTPDPYTLRLLDGYGGNPGLTIRQPSAKVLGSYPPTNDVPYTDAGYMINSVVVDLRLTSSEWSDVVPDRQV	156
EGR49603	143	TAFDTHSVTVQYFQGGASMTSPLIPGSPYITLQYQAATPLLTSRNGGIASFNGQNLSNGQSVTATGTSFTVVDTTGPLIPGSPYITLQYQAATPLLTSRNGGIASFNGQNLSNGQSVTATGTSFTVVDTTG	218
AGV00786	157	TDWDHLSANLRLSTPQDSNSYIDFPIVR@MAYITANYNNLTPQFLSQHAIISVEADEKKSDDNTSTFSGRKFKITMNDDP	236
EGR49603	219	- TTYVIYALSSISLTATATNSAQGTIKATGTYNGVLRLVRLTQASHKALLDQHYTVYPTGVGLDYSFTTTTGTLI-FNYN	296
AGV00786	237	TSTFIIVSLG-DKPLELRKQDNSNLVASKPYTGVIRVAKLPAPEFETLLDASRAVWPTGGDISARSDDNNGASVTIKWK	314
EGR49603	297	TVGDGSQLLMLTWP)HRLSLQGANQPATSSLGYLTTKGWMYPIIGNQWKLL-YQLSSITWNP-PRALDSSCSSAVIQGLQ	374
AGV00786	315	TNSNEAPLLTY AV ANOLTSIDDSNYKRTIMTLQSATKGPMT ALVGNEWTLRETELSPVEWLPLQAAPNPTTINEIMTEIN R4 #	394
EGR49603	375	YEIGLIANSTPPVPNERYYWGGSLAAQARLALIAEAVGRTDLIPTVTNYLKTSFQNWFTPSTGASP-AYETSWGGV	449
AGV00786	395	KDI-ASNYTQETAKEDNYFSGKGLQKFAMLALILNKSDQTQLRNPELAQIALDKLKAAFLPYLQNEQADPFRYDTLYKGI	473
EGR49603	450	IDKAG ATNSGILFGNGYYNDIOFHY GYFLYVAAVIAKYDANWLAQ-HKDFINWFARDIINPSPNDPYFPVTRC	521
AGV00786	474	VAKAGLPTSMGGTDDLSAEFGHSYYSDIOUND OUTPAAFRN R1R2 R3	553
EGR49603	522	RDWFAGHSWASGIANGA—GSRDQËSTGËAVNGYYGALLWATVALSQDYVNYARLLVATEQQGAQVYWHLYPQQSQTDPN	599
AGV00786	554	WDWFAGHSWAGGIKPDGALDGRDdESVHESVNFYWGAKLWGLATGNTPLTKLASLQLAVTKRTTYEYFWMLDG	627
EGR49603	600	NPYPEPAVRNLVTMGNVEDWQSGAWLFWGNQKSEIAAIQMIPVTP-INEVLYDAQWVNNVWSYAQNEIVDPSIADDWRSV	678
AGV00786	628	KNRPENIVRNKV-IGIYFEQKIDYTTYFGRFLEYIHGIQQIPMIPELMEYIRTPEFVSQEWDEKLGAIA-PTVQSFWAGV	705
EGR49603	679	MIAAYSNANPQTAAAWSANLTTWGSGNTFSNELFFIGTRPNPSGQPICGSNFPQNPYGNFKIQSATTGQWVVASTASSNL	758
AGV00786	706	LYLNYAIINPAEAYPALRKVQ-MDDGQTRSYSLYLTATRPH-FFRRSLLAALARHGSTRRPSL	766
EGR49603	759	VASOSQASAGVFISSYTPNAGNLKLTSNNQFVTADQSGNFALQAARATASSWEVFTIRQKVGAAAGVYTIKAGSNGKWVT	838
AGV00786	767	PSSGDDDK-HEDGFLLRFRRLNP-FNLKHRIY-	796
EGR49603	839	LASDGSLINNGATEASAAGFKFVQS 863	
AGV00786			

Fig.52 Alignment of the *Tr*GH81 sequence.

EGR49603: TrGH81from T. reesei;

AGO00786: RmLam81A from R. miehei;

Strictly conserved residues were highlighted in red; sugar-binding related residues were framed in blue; and catalytic related residues were marked with "#".

The gene of putative endo- β -1,3-glucannase *Tr*GH81 was cloned by a PCR-based strategy using *T.reesei* cDNA as template. A PCR product of 2576 bp was obtained, containing a mature enzyme (851 aa) nucleotide ORF of 2556

bp. The predicted molecular mass of *Tr*GH81 is 91.24 kDa. As shown in Fig.35A, lane 6, the only bright band with a size around 2.5 kb was obtained, and ligated into pJET 1.2 cloning vector for plasmid construction.

The gene was ligated into the vector pAaHBgl by *Xho* I and *Not* I to construct the recombinant plasmids pAaHBgl-*Tr*GH81. Two fragments with sizes of 7395 bp and 2576 bp should arise after digestion with *Xho* I and *Not* I when plasmid construction was correct.



Fig.53 Analysis of pAaHBgl-TrGH81 construction.

Recombinant vector pAaHBgI-*Tr*GH81 digestion with restriction endonucleases. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1-3, results of three recombinant plasmids digested by *Xho* I and *Not* I.

In Fig.53, only lane 3 showed two bands in right size representing two fragments cleaved by *Xho* I and *Not* I from correctly constructed recombinant plasmid. In lane 1, the first faint band about 10 kb was probably uncut original vector pAaHBgI, the second bright band was the longer fragment linearized by restriction endonucleases from pAaHBgI, and the third band with the size about 1,000 bp was the gene inserted between *Xho* I and *Not* I. The uncut original vector existed from the first step of construction and was transformed into *E.coli*

causing a false positive clone. In lane 2, the smaller band was about 2.5 kb, which met the size of target gene (2576 bp), but the other band about 3.0 kb might be a linearized cloning vector pJET 1.2. The bands in lane 2 indicated that circular cloning vector pJET 1.2 with target gene were transformed into *E. coli* strain during the construction procedure. Therefore, the recombinant vector showing in Fig.53 Lane 3 was selected as positive expression vector for transformation into *P. pastoris* strain GS115. Colonies appeared on the selective plates were picked and cultivated in small scale for recombinant protein expression screening.

SDS-PAGE analysis of the supernatants after methanol induction demonstrate that a predominant protein with the molecular mass between 70 and 100 kDa (Fig.54, Lane 2 and 4 (highlighted with red underline)) was expressed, respectively, which meets the predicted molecular mass of 91.24 kDa, indicating a secretion of recombinant *Tr*GH81. Enzyme activities of these two samples were also observed when act on both alkali treated paramylon and laminarin (in Fig.55).



Fig.54 SDS-PAGE analysis secreting expression of recombinant *Tr*GH81. Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); lane 1-6, 15 μL culture supernatant of six recombinant *P*. *pastoris* GS115 strains after methanol induction for 120 hours in small scale using 24 square-well plate (section 2.4.4.1).



Fig.55 Enzyme activity assay from culture supernatants of selected transformants (*Tr*GH81). Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction in small scale using 24 square-well plate (section 2.4.4.1) were used as crude enzymes for endo- β -1,3-glucanase activity assay. 90 µL enzyme sample was mixed with 90 µL laminarin (2.0 g/L, pH=6.5, indicated by black column) and alkali treated paramylon (2.0 g/L, pH=6.5, indicated by gray column), respectively, and incubated at 37 °C for 60 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

The secreted proteins in Lane 2 and 4 of Fig.54 were active on alkali treated paramylon and laminarin. The crude enzyme sample NO.2 had activity of 17.2 IU/L toward alkali treated paramylon and 24.8 IU/L on laminarin, which were slightly lower comparing with sample NO.4, whose activites on these substrates were 17.9 IU/L and 26.5 IU/L respectively. Activities of sample NO.2 and NO.4 were obviously higher than those of the other four samples selected, which had

no predominant protein as showing in Fig.55. This suggested that the major secreted protein contributed cleavage activites against both substrates. Based on these results, the expression and enzymatic activity of recombinant *Tr*GH81 as an endo-β1,3-glucanase was preliminarily confirmed. And the recombinant strain NO.4 was named as *P. pastoris* GS115-*Tr*GH81 for shake flask cultivation.

During the methanol induction, supernatant was taken at each 24 h and used as crude enzyme for hydrolytic assay. The total protein concentration and enzyme activity results are showing in Fig.56 below.



Fig.56 Recombinant *Tr*GH81 expression in *P. pastoris*. Supernatants of selected recombinant *P.pastoris* GS115 transformants after methanol induction using 24 shake flask (section 2.4.4.2) were used as crude enzymes for endo- β -1,3-glucanase activity assay. 90 µL enzyme sample was mixed with 90 µL laminarin (2.0 g/L, pH=6.5) and alkali treated paramylon (2.0 g/L, pH=6.5), respectively, and incubated at 37 °C for 60 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

During the methanol induction, endo- β -1,3-glucanase activity detected by determining reducing ends generated from laminarin increased from none to 118.3 IU/L at 120 h, while the total protein concentration in the supernatant was 0.26 g/L. Comparing to 26.5 IU/L detected from 24-well plate cultivation, the activity of recombinant enzyme achieved a fourfold increasing. As being observed in other recombinant enzyme assay, similar trend appeared also when alkali treated paramylon was used as substrate, but with lower activity. After induction for 120 hours, the enzyme activity of recombinant *Tr*GH81 was 81.5 IU/L in the supernatant. This was a fourfold increasing by comparison with 17.9 IU/L in multi-well plate cultivation.

Hundreds of proteins were classified in glycoside hydrolase family 81, which were widely distributed in fungi, bacteria and plants (Davies and Sinnott 2008). In an inverting hydrolytic mechanism, all GH family enzymes specificly broke β -1,3-glycosidic bond by endolytic mode (McGrath and Wilson 2006). However, limited members of this family have been characterized, such as *Eng*A from *Aspergillus nidulans* FGSC A4 (Mouyna et al. 2002), *Lam*81 from *Thermobifida fusca* (McGrath and Wilson 2006), a β -glucan elicitor binding protein expressed in tobacco suspension cultured cells and in *E. coli* for elicitor activity assay (Umemoto et al. 1997), and *Eng* from *Pneumocystis carinii* expressed in *E. coli* for determining immunoreactivity (Kutty et al. 2015). Nevertheless, as a putative endo- β -1,3-glucanase, *Tr*GH81 was initially secreted in *P. pastoris* and be confirmed by both protein electrophoresis and enzyme assay, especially displayed activity towards alkali treated paramylon.

2.3.6 Cloning and expression of *Tr*GH17 (EGR46171.1)

Nucleotide and amino acid sequences of *Tr*GH17 (GenBank: EGR46171.1) are shown in Fig.57. The open reading frame had a size of 1206 bp, corresponds to a translation product with 401 amino acids. Predicting by SignalP 3.0, a N-terminal signal peptide existed and the cleavage site was between positions 19 (a) and 20 (g). No propetide cleavage site was predicted by ProP1.0 Server at the downstream of signal peptide.

1 atgaagetgt etaegaeeat egegaetgge getgeeettg tggeeggtgt ttetgetgga eagaaeaaet acetegget eaaeteeggee aaeaeeetee eegaegagte egeeaagte gagaaggaet g v s a g q n n y l g f n s g n t l 131 teettgeega gtteteeaeg geeeagaage tegteggege eeeeggeaee tteaaegeeg teegteteta caecaaeate eaggeetaet eeaaggaeae teetategag geetteteeg eegeeateaa a or k apgt fna vrlytni qayskd tpi 261 gaccaagacg tacattetee teggtgtetg ggeeteggge aetgataaca tegacaatga getggetgee etgaegegeeg etgteaagea gtaeggeaag gaeetgaegg aeetgateat tggtgtetee tkt yillgv wasg t d n i d n e l a a l s a a v k q y g k d l t d l i i g v s 391 atcggcagcg aggatetgta cegtgaeteg cagaetggte gaaceaacaa ggeeggtgte ggeaaeggee ecaaggaggt eeteggette ateaaegaet aeaagaagae etttgeeaa aetgetetgg i g s e d l y r d s q t g r t n k a g v g n g p k e v l g f i n d y k k t f a n t a l 521 ccaatgteee categgeeae gtegaeaeet gggatgeetg ggteaaegge aceaaeaage eegteetega egeegtegae tggateggtg ttgaegagta eeettetae gagaeaggea agggeaaega vnq tnk p v l 651 cattageaac geeggeaage tettegacae ggetttegag aceaegetgg gegetgeeaa tggeaageee gtetgggtga eggagaeegg etggeeeetg aceggeeeeg actgggaega ggeeaageee a g taf gaa ng kp vw gwpl tgp 781 agegteaaga acgeeeagaa gtaetggeag gaeattgget geaagagget etteaaeag taeeeeaeat tetggtaeaa eetgegtgae teeaaeegg eeaaeeaggt eaagteegge ateageeaga s v k n a q k y w q d i g c k r l f n k y p t f w y n l r d s n p a n q v k f g i s q 911 geetetete caeceetteg ttegacetga eetgeeeaa ggaggagaee aceaeeteeg eggeeaagee caecegeeae aceetegtea aggettetgg aacaeeegaa getgteaagt etggeteega s l s s t p s f d l t c p k e e t t t s a a k p t a t t l v k a s g t p e a v k s g s 1041 gggcaccgct aacgeeggag getecaagte eageaacage gatgetteea ceaceteate aggeeetgee geggeeacta ecaeeggege eggtgeegtg accaaggtet eeggegetge eettgeeggt 1171 gtggccatgg ttgccggtct cctggccctg ttctaa

Fig.57 Nucleotide sequence and amino acid sequence of TrGH17.

In the coding region, the putative signal peptide was boxed.

By aligning amino acid sequence with available protein sequences from the GenBank database, *Tr*GH17 showed high similarity to the GH family 17 members, such as 67% indent to exo- β -1,3-glucanase from *Pochonia chlamydosporia* 170 (GenBank: XP_018147378.1) (Wang et al. 2016), and 69 % indent to 1,3-beta-glucosidase from *Colletotrichum incanum* (GenBank: KZL82930.1) (Hacquard et al. 2016). When aligned with the characterized exo- β -1,3-glucanase naming *Bgl*2 from *Saccharomyces cerevisiae* (GenBank: AAA34648.1), as shown in Fig.58, they shared the GH family 17 signature

fragment [LIVMKS]-x-[LIVMFYWA](3)-[STAG]-E-[STACVI]-G-[WY]-P-[STN]-x-[SAGQ] and the two strictly conserved Glu residues (R1 and R2) (Klebl and Tanner 1989).

```
EGR46171
1
MKLSTTIATGAALVAGVSAGQNNYLG—PNSGNTLPDESAKFEKDFLAEFSTAQKLVGAPGTFNAVRLY-TNIQAYSKD
76

AAA34648
1
MRFSTTLAT-AATALFFTASQVSAIGELAFNLG—VKNNDGTCKSTSDYETELQALKSY
56

R1
R1
#

EGR46171
77
TPIEAFSAAIKTKTYILLGV—WASGTDNIDNELAALSAAVKQYGKDLTDLIIGV-SIGSEDLYRDSQTGR
144

AAA34648
57
TS—TVKVYAASDCNTLQNLGPAAEAEGFTIFVGVWPTDDSNYAAEKAALQTVLPKIKESTVAGFLVGSEALYRNDLTA-
133

EGR46171
145
TNKAGVGNGPKEVLGFINDYKKTFANTALANVFIGHVDTWDAWVNGTNKPVLDAVDWIGVDEYPFYETGKGNDISNAG-K
223

AAA34648
134
—SQLSDKINDVRSVVADISDS-DGKSYSGKQVGTVDSWNVLVAGYNSAVIEASDFVMANAFSYWQ—GQTMQNASYS
206

R2
#
#
#
#

EGR46171
224
LFDT—AFETTLGAANGKPV—WVTETGWPLTGPDWDEAKPSVKNAQKYWQDIGCKR—LFNKYPTFWYNLRD
290

AAA34648
207
FFDDIMQALQVIQSTKGSTDITFWYGETGWPTDGTNFESSYPSVDNAKQFWKEGICSMRAWGVNVIVFEAFDEDWKPNTS
286

EGR46171
291
SNPANQVKFGISQSLSSTPSFDLTCPKEETTISAAKPTAT
330
313
```

Fig.58 Alignment of the *Tr*GH17 sequence.

EGR46171: TrGH17 from T. reesei, AA1-330;

AAA34648: Bgl2 from S. cerevisiae.

Strictly conserved catalytic Glu residues (R1 and R2) were marked with "#". Glycoside hydrolase family 17 signature fragment were framed.

The gene of putative exo- β -1,3-glucanase *Tr*GH17 was cloned by a PCRbased strategy using cDNA as template. A PCR product of 1190 bp was obtained containing a 1170 nucleotide ORF, which encoding 381 amino acids of mature protein with a predicted molecular mass of 41.06 kD. Fig.59A shows the agarose gel analysis of PCR product from cDNA of *T. reesei* by PCR using *Tr*GH17 gene specific primers. There was one bright band between 1000 and 1500 bp as the main product, which met the right size according to the target gene. Meanwhile, some strain bands existed, which might due to non-specific amplifications. However, these products had no influence on isolating the target gene from agarose gel for cloning vector construction.



Fig.59 Analysis of *Tr*GH17 gene application and pAaHBgl-*Tr*GH17 construction.

A: Analysis of genes of glucanases obtained from cDNA of *T. reesei* QM6a by PCR using specific primers. M: 1 kb DNA ladder; lane 1: Gene of glucanase *Tr*GH17 (1190 bp);

B: Recombinant vector screening by digestion with restriction endonucleases. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1 to 4, results of four recombinant pAaHBgl-*Tr*GH17 plasmids digested by *Xho* I and *Not* I.

Recombinant plasmid pAaHBgl-*Tr*GH64 construction was performed as described in (section 2.4.2). Gene of *Tr*GH64 was ligated into the vector pAaHBgl between sites *Xho* I and *Not* I. To screen the positive constructions, *Xho* I and *Not* I were used to digest the recombinant plasmids. Two fragments with the sizes of 7395 bp and 1190 bp should be produced when the gene was inserted correctly. The digestion products were visualized by agarose gel

electrophoresis (see Fig.59B). All four plasmids tested had two fragments in right size, and the bands larger than 10 kb were probably un-cut circular vectors.

Afterwards, the recombinant vector shown in Fig.59B, lane 1 was selected as positive expression vector for further transformation into *P. pastoris* strain GS115. Performed as mentioned in section 2.4.4.1, transformants were picked from selective agar plate and induced by methanol for recombinant protein expression in 24 square-well plates. Supernatants from the small-scale cultivation were used as crude enzyme samples for SDS-PAGE and activity assay. Two positive recombinant strains were obtained demonstrating by the SDS-PAGE of secreted proteins. In Fig.60, lane 2 and 4 respectively, both contains a distinct band (highlighted with red underline) indicating a secreted protein with molecular mass about 40 kDa, which meets the predicted molecular mass of *Tr*GH17 (41.06 kDa). No band with similar molecular mass appears in other samples (lane 1,3,5 and 6).



Fig.60 SDS-PAGE analysis secreting expression of recombinant *Tr*GH17. Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); lane 1-6, 15 μ L culture supernatant of six recombinant *P. pastoris* GS115 strains after methanol induction for 120 hours in small scale using 24 square-well plate (section 2.4.4.1). Activities of these crude enzyme samples were measure by D-Glucose assay kit as described in section 2.4.5.3. The exo- β -1,3-glucanase enzyme activity was converted in term of IU/L, where one unit of enzyme activity (IU) was defined as the amount of enzyme required to liberate 1 µmol reducing sugar equivalent in 1 minute under the assay condition. The results in Fig.61 confirmed the distinct proteins showing in Fig.60, lane 2 and 4 were active towards laminarin and alkali treated paramylon and released free glucose from the substrates as hydrolytic product.



Fig.61 Enzyme activity assay from culture supernatants of selected transformants (*Tr*GH17). Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction in small scale using 24 square-well plate (section 2.4.4.1) were tested for $exo-\beta-1,3$ -glucanase

activity. 100 μ L enzyme sample was mixed with 100 μ L using alkali treated paramylon (A, 2.0 g/L, pH=6.5, indicated by black column) and laminarin (B, 2.0 g/L, pH=6.5, indicated by gray column), respectively, and incubated at 37 °C for 15 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

After 120 hours methanol induction in 24-well plate, two of the selected recombinant strains (No.2 and 4) secreted putative exo-β-1,3-glucanase *Tr*GH17 as main product in the supernatant. The data of enzyme activity assay confirmed the conclusion by showing high activity on both substrates. With the total protein concentration of 0.65 g/L, crude enzyme No.2 showed activity of 447.4 IU/L on alkali treated paramylon and 1363.6 IU/L on laminarin, obtained the relative activity of 688.3 IU/g and 2097.8 IU/g, respectively. Moreover, the measured activity of crude enzyme No.4 were 511.4 IU/L on alkali treated paramylon, and 1491.5 IU/L on laminarin with protein concentration 0.7 mg/L corresponding to relative activities of 730.6 IU/g and 2130.7 IU/g, respectively. Based on result of SDS-PAGE and activity assay, the recombinant *P. pastoris* strain No.4 was selected for further work and named as *P. pastoris* GS115-*Tr*GH17.

Similarly, the *P.pastoris* GS115-*Tr*GH17 cultivation was performed in 250 mL shake flask and changed the medium to BMMY for methanol inducing. Supernatants of every 24 h were assailed for total protein concentration and exo-hydrolysis activity (section 2.4.5.3).



Fig.62 Recombinant *Tr*GH17 expression in *P. pastoris*. Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction using 24 shake flask (section 2.4.4.2) were used as crude enzymes for exo- β -1,3-glucanase activity assay. 100 µL enzyme sample was mixed with 100 µL laminarin (2.0 g/L, pH=6.5) and alkali treated paramylon (2.0 g/L, pH=6.5), respectively, and incubated at 37 °C for 15 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

Along the methanol induction, extracellular proteins accumulated steadily and reached a relatively stable concentration of about 0.75 g/L after 96 h of induction. As confirmed by the SDS-PAGE (Fig.60), recombinant *Tr*GH17 was the main protein in the supernatant. The putative exo- β -1,3-glucanase was supposed to release free glucose from the non-reducing end of the substrate as end product. The quantitative analysis of the enzymatic hydrolysate demonstrated that during the 120 hours of cultivation, enzyme activity increased along the induction time. Generally, the recombinant *Tr*GH17 hydrolyzed laminarin more efficiently and showed high enzyme activities. Along the extracellular protein secretion, hydrolytic activity of *Tr*GH17 towards
laminarin increased steadily, and achieved the highest value of 5859.3 IU/L after 120 hours. When using alkali treated paramylon as enzyme assay substrate, the increasing trend was also observed in general, but with lower activity at each time point. Moreover, sample collected at 96 h obtained the highest 2705.9 IU/L. When the crude enzyme collected at 120 h acted on paramylon, the activity of 2514.2 IU/L was lower than the activity at time point 96 h. Although the hydrolysis was conducted mainly by recombinant *Tr*GH17, exo- β -1,3-glucanases produced by *P. pastoris* for cell growth and metabolism might also participate at the substrate degradation.

Abundant enzymes were classified as GH family 17 members, and 83 enzymes were characterized as β -1,3-glucanases. Among these enzymes, only *Bgl2* from *S. cerevisiae* (GenBank: AAA34648.1) (Plotnikova et al. 2006) and a β -1,3-glucan transferase from *Candia albicans* were characterized as exo-glucanase (Tsai et al. 2011). In this work, *Tr*GH17 was cloned and expressed in *P. pastoris* for the first time. The activity of the recombinant enzyme on laminarin and alkali treated paramylon were confirmed by determination of glucose as hydrolysis product.

2.3.7 Cloning and expression of *Pp*GH5 (CAY69081.1)

Nucleotide and amino acid sequences of *Pp*GH5 (GenBank: CAY69081.1) are shown in Fig.63. The open reading frame had a size of 1245 bp terminated by TAA stop codon, corresponding to a translation product with 414 amino acids. Predicting by SignalP 3.0, a N-terminal signal peptide existed and the cleavage site lay between positions 16 (s) and 17 (i). And a processing site for Kex2-like endoprotease was found downstream, resulting the mature protein N-terminal from position 23 (d).

131 tggagcccta tatcacccct tcgctttttg aagctttcgg cgatgatgta ccagtagatg aatatcgtta cactgaacga ctaggtaaat cgctggcttt ggatagactt caacagcatt ggtcaacgtt pyitp slf eaf gdd v pvd eyr yter lgk sla ldrl 261 ctacgacgaa aaagattttc aagacatcgc agcttacgga cttaactttg ttcgtatacc aattggctac tgggccttcc aattgctgga tgatgatcct tatgttcaag gccaggaaga gtatcttgac f y d e k d f q d i a a y g l n f v r i p ig y w a f q l l d d p y v q g q e e y l d 391 aaggetttgg agtggagtag aaageatgga etgaaagttt ggattgaeet teatggaget eeaggatete agaaeggett tgaeaaetee ggtaagegag acagttggga ttteeaaaat ggtaaeaaeg pgs 521 ttcaagtaac tttggatgtt ctgaaatata tttcaaaaaa atatggaacc acggactact acgatgtggt cattggtatt caactcttaa acgaaccatt gggacctatt ttagacatgg ataatctaag l k y i s k k y g t t d y y d v vigi qll nep g p i 651 acagttetat geggatggtt atgatetagt tagagatgtt gggaacaact ttgttgtaat eeagatgea ttttaecagg egecagagta etggggggae gattteaeet eageggaagg ttaetggaae r q f y a d g y d l v r d v g n n f v v i h d a f y q a p e y w g d d f t s a e g y w n 781 gtggtgcttg atcaccacca ttatcaagtc ttcgatgcag atgaattgca aagaagtatc gatgaacata tagaagccgc ctgtgattgg ggtagagatg caaataaaga gtaccactgg aacctctgtg qrsi h y q 911 gtgaatggtc ggcagcactt actgattgta ctccttggtt aaatggtgtc ggaaaaggca cgagatatga aggtcaactt gataactccc cttggatcgg atcttgtgag aatagccagg atccttcgaa lng g q 1 tpw g k g t r y d n s p 1041 attgagetet gaaegtatet gtgagtaeag aaggtaegta gaageerage tagatgett ettaeaeggg aaaagegeag gttttattt etggtgttte aagaeagagg eeagtttgg gtgggattt k l s s e r i c e y r r y v e a q l d a f l h g k s a g f i f w c f k t e a s l e w d f 1171 aaaaggttgg ttaatgccgg tatcatgcct cagccattgg acgacagaca gtatccaaat caatgtgggt tctaa k r l v n a g i m p q p l d d r q y p n q c g f -

Fig.63 Nucleotide sequence and amino acid sequence of *Pp*GH5.

In the coding region, the putative signal peptide was boxed, and the predicted propeptide cleavage site was indicated by arrow.

A comparison of *Pp*GH5 amino acid sequence with known proteins in GenBank database indicated high sequence identity and similarity to exo- β -1,3-glucanses from glycoside hydrolase family 5 (Fig.64). These enzymes shared conserved the active relating residues R1 (Glu) and R2 (Glu) and several regions including the GH family 5 signature fragment in form of:

-[LIV]-[LIVMFYWGA](2)-[DNEQG]-[LIVMGST]-X-N-E-[PV]-[RHDNSTLIVFY]-(Xu et al. 2006). R1 and R2 served as the proton and nucleophile respectively cleaving the glycosidic bond in a double-displacement mechanism (Cutfield et al. 1999).

1	MN L-YLITLLFASLCSAITLP KRDIIWDYSSEKIMGVNLGGWLVLEPYITPSLFEAFG	57
1	ML[5]LCTLLTVSSVLATPVPARDPSSIQ[7]KRYYDYDHGS1gEPIRGVNIGGWLLLEPYITPSLFEAFRTNDDN	82
1	MN L-TLLLLALIfsPSLIFSLPTANKVKLV KKGLNWDYQNAKIHGVNLGGWFVLEPFITPSLFDIYSKPNDD	71
1	MK LLSLLTLALGALSSPIPSQGGAQIY KRSFNYQTDKLRGVNIGGWLVLEPYITPSLFEVFG	62
58	DD-VPVDEYRYTERLGKSLALDRLQQHWSTFYDEKDFQDIAAYGLNFVRIPIGYWAFQLLDDDPYVQG-QEEYLDKALEW	135
83	DEgIPVDEYHFCQYLGKDLAKSRLQSHWSTFYQEQDFANIASQGFNLVRIPIGYWAFQTLDDDPYVSG1QESYLDQAIGW	162
72	SQ-VPVDEYHFTQKLGKDAAQQVLEQHWKTWYKENDFKMMLKYGLNAVRIPIGYWAFKLLDYDPYVQG-QVKYLDRALDW	149
63	DN-IPVDEYHYHQYLGAELAQSRLQQHWGSWITEQDFESIKGTGLNFVRIPIGYWAFQKLDSDPYVQG-QVEYLDKAIQW	140
	R1 #	
136	SRKHGLKVWIDLHGAPGSQNGFDNSGKRDSWDFQNGNNVQVTLDVLKYISKKYGTTDYYDVVIGIQLLNEPLBPILDMDN	215
163	ARNNSLKVWVDLHGAAGSQNGFDNSGLRDSYKFLEDSNLAVTTNVLNYILKKYSAEEYLDTVIGIELINEPLSPVLDMDK	242
150	ARKYNLKVWIDLHGAPGSQNGFDNSGLRDSLGFQQGNNVNFTLEVLEIIGKKYGGPEYEDVVIGIELLNEPLGPSLDLNY	229
141	ARNSGLYVWIDLHGAPGSQNGFDNSGLRDSYEFQNGNNTQITLDVLQQIFDKYGSSDYDDVIIGLELLNEPLSPVLDMAK	220
216	L-RQFYADGYDLVRD-VGNN-FVVIHDAFyQAPEYWGDDFTSAEGYWNVVLDHHHYQVFDADELQRSIDEHIEAACDWGR	292
243	MkNDYLAPAYEYLRNnIKSDQVIIIHDAF-QPYNYWDDFMTENDGYWGVTIDHHHYQVFASDQLERSIDEHIKVACEWGT	321
230	L-KEFFQQGYQNLRN-SGSVQAVIIQDAF-QPMGYW-DNFLTLDQYWNVVVDHHHYQVFSAGELQRSIDDHITVACNWGW	305
221	L-NEFWETAYWNLRN-SNSTQTVVIHDAF-TASGYFNDKFQLNQGYWGLVIDHHHYQVFSQQEVQRSIDEHVEVACQWGK	297
	R2 #	
293	DANKEYHWNLCGEWSAALTDCTPWLNGVGKGTRYEGQLDN SPWIGSCENSQDPSKLSSERICEYRRYVEAQLDAF1h	369
322	GVLNESHWTVCGEFAAALTDCTKWLNSVGFGARYDGSWVN[4]SSYIGSCANNDDIAYWSDERKENTRRYVEAQLDAF	400
306	DAKKEYHWNVAGEWSAALTDCARWLNGVGRGARFSGDFDN SPYFGSCDCYVNIATWPSEYRTNVRRYIEAQLDAF	380
298	DSKGENLWNLCGEWSAALTDCAKWLNGVGKGARYDQTFGN SQYTGSCTNSQDISTWSSDVKANYRRYIEAQLDAF	372
370	GKSAGFIFWCFKTEASLEWDFKRLVNAGIMPQPLDDRQYPNQCGF 414	
401	EMRGGWIIWCYKTESSLEWDAQRLMFNGLFPQPLTDRKYPNQCGTisn 448	
381	EQTGGWFFWNWKCENAIEWDLQGLITAGVFPYPFYNRQFPNQCGF 425	
373	EQRGGWVFWCWKTENAGEWDFQKLAYNGVFPQPLDDRQYPNQCGF 417	
	1 1 1 58 83 72 63 136 163 150 141 216 243 230 221 293 322 306 298 370 401 381 373	1 MN L-YLITLLFASLCSAITLP KRDIIMDYSSEKIMGVNLGGWLVLEPYITPSLFEAFRTNDDN 1 ML[5]LCTLLTVSSVLATPVPARDPSSIQ[7]KRYYDYDHGSIgEPIRGVNIGGWLLEPYITPSLFEAFRTNDDN 1 MN L-TLLLLALIFSPSLIFSLPTANKVKLV KKGLNWDYQNAKIHGVNIGGWLVLEPYITPSLFEAFRTNDDN 1 MN L-TLLLLALIFSPSLIFSLPTANKVKLV KKGLNWDYQNAKIHGVNIGGWLVLEPYITPSLFEAFRTNDDN 1 MK LLSLLTLALGALSSPIPSQGAQIY KRSFNYQTDKLRGVNIGGWLVLEPYITPSLFEVF6 28 DD-VPVDEYRYTERLGKSLALDRLQQHWSTFYDEKDFQDIAAYGLNFVRIPIGYWAFQLLDDDPYVQG-QEEYLDKALEW 29 DEYVTERLGKSLALDRLQQHWSTFYDEKDFQDFANIXASQGFNLVRIPIGYWAFQLLDDDPYVGG-QVEYLDKALEW 20 VPVDEYHFYQKLGADAAQQVLEQHWKTWYKENDFKMMLKYGLNAVRIPIGYWAFQLLDDDPYVGG-QVEYLDKALOW 21 D-IPVDEYHYHQYLGAELAQSRLQQHWGSWITEQDFESIKGTGLNFVRIPIGYWAFQLDDPYVGG-QVEYLDKALOW 23 DV-IPVDEYHYHQYLGAELAQSRLQQHWGSWITEQDFESIKGTGLNFVRIPIGYWAFQLDDPYVGG-QVEYLDKALOW 24 ARNISLKVWDLHGAAGSQNGFDNSGRDSGRDSGKPSQHQMINQVTLDVLKYISKKYGTDYVDV/TGIQLLNEPLSPLDMDK 25 ARKYNLKWUDLHGAAPGSQNGFDNSGRDSGLRDSGKFQMSUNQVTLDVLKYISKKYGTDYDVVTIGIELLNEPLSPLDMKK 26 L-RQFYADGYDLVRD-VGNN-FVVIHDAFQAPEYWGDDFTSAEGYMNVVLDHHYQVFDADELQRSIDEHIEAACDWGR 216 L-RQFYADGYDLVRD-VGNN-FVVIHDAF-QPYNWDDFMTENDGYWGVTIDHHHYQVFDADELQRSIDEHIEAACDWGR 216 L-RQFYADGYULWAD-YGNN-FVVIHDAF-QPYNWDDFMTENDGYWGVT

Fig.64 Alignment of the *Pp*GH5 sequence.

CAY69081: PpGH5 from P. pastoris;

AAA34599: EXG1 from S. cerevisiae;

CAA86951: SoEXG1 from Schwanniomyces occidentalis;

ACP74152: exo-β-1,3-glucanase from *Cyberlindnera saturnus*.

Strictly conserved residues were high-lighted in red, conserved catalytic residues R1 (Glu) and R2 (Glu) were marked with "#". Glycoside hydrolase family 5 signature fragment were boxed and strictly conserved residues in this region were high-lighted with blue background.

Gene of putative exo- β -1,3-glucanase *Pp*GH5 was cloned by a PCR-based strategy using genomic DNA as template. A PCR product of 1224 bp was obtained containing a 1197 nucleotide ORF encoding 398 amino acids of mature protein with a predicted molecular mass of 46.2 kDa. As shown in Fig.65A, lane 1, the only bright band with a size between 1.0 a 1.5 kb was the gene obtained. The bright band shown in Fig.65A, lane 1 was cleaned up and inserted into pJET 1.2 cloning vector for recombinant plasmid construction.



Fig.65 Analysis of *Pp*GH5 gene application and pAaHBgl-*Pp*GH5 construction.

A: Analysis of gene of *Pp*GH5 obtained from genomic DNA of *P. pastoris* GS115 by PCR using specific primers. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1: Gene of *Pp*GH5 (1224 bp);

B: Recombinant vector screening by digestion with restriction endonucleases. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1 to 4, results of four recombinant pAaHBgl-*Pp*GH5 plasmids digested by *Xho* I and *Not* I.

The gene was inserted into the vector pAaHBgI at sites *Xho* I and *Not* I afterward. Two fragments with the sizes about 7.5 kb and 1.2 kb should be

produced when the gene of interested was inserted correctly, when digest by *Xho* I and *Not* I. The screening results were showed in Fig.65B. All plasmids analyzed were rightly constructed, confirmed by the two right length cleavage products. The recombinant plasmid corresponding to Fig.65B, lane 4 was transformed into *P. pastoris* GS115 for protein expression.

Transformants were picked from selective agar plate and induced by methanol for recombinant protein expression in 24-well plates. Supernatants cultivation were used as crude enzyme samples for SDS-PAGE and activity assay. The only distinct protein band showing in Fig.66, lane 4 with a molecular mass about 50 kDa met the predicted *Pp*GH5 molecular weight (46.2 kDa). Results of protein electrophoresis suggested that lane 4 corresponds to recombinant *Pp*GH5 successfully expressed in *P. pastoris*. The activity assay performed afterwards (see Fig.67) confirmed this hypothesis and demonstrate that the recombinant *Pp*GH5 was an active enzyme on both laminarin and alkali treated paramylon.



Fig.66 SDS-PAGE analysis secreting expression of recombinant PpGH5. Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); lane 1-5, 10 µL culture supernatant of recombinant *P. pastoris* strains after methanol induction for 120 hours in small scale using 24 square-well plate (section 2.4.4.1).



Fig.67 Enzyme activity assay from culture supernatants of selected transformants (*Pp*GH5). Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction in small scale using 24 square-well plate (section 2.4.4.1) were tested for exo- β -1,3-glucanase activity. 100 µL enzyme sample was mixed with 100 µL using alkali treated paramylon (A, 2.0 g/L, pH=6.5, indicated by black column) and laminarin (B, 2.0 g/L, pH=6.5, indicated by gray column), respectively, and incubated at 37 °C for 15 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

The exo- β -1,3-glucanase hydrolysate was measured by D-Glucose assay kit and converted to enzyme activity (section 2.4.5.3). After methanol induction in 24-well plate for 120 hours, *Pp*GH5 was secreted as main protein by recombinant strain No.4 (see Fig.66, lane 4) and generated high activity toward both laminarin and alkali treated paramylon. Although the total protein concentration of the crude enzyme was 0.05 g/L, the activity on alkali treated paramylon was 937.5 IU/L, and 1811.1 IU/L on laminarin. The relative activities were 18.8 IU/mg and 36.2 IU/mg respectively, which were significantly higher than recombinant *Tr*GH17 mentioned in section 2.3.6. The enzyme activity results as well as the SDS-PAGE proved that the exo- β -1,3-glucanase *Pp*GH5 was successfully overexpressed by recombinant *P. pastoris* GS115-*Pp*GH5. The distinctive activity was also showed in the shake flask cultivation.



Fig.68 Recombinant *Pp*GH5 expression in *P. pastoris*. Supernatants of selected recombinant *P. pastoris* GS115 transformants after methanol induction using 24 shake flask (section 2.4.4.2) were used as crude enzymes for exo- β -1,3-glucanase activity assay. 100 µL enzyme sample was mixed with 100 µL laminarin (2.0 g/L, pH=6.5) and alkali treated paramylon (2.0 g/L,

pH=6.5), respectively, and incubated at 37 °C for 15 min. Substrate preparation and enzyme activity assay were described in section 2.4.5.

*Pp*GH5 was secreted as the main protein in the supernatant and contributed anomalous effective hydrolysis activity on both substrates. In general, the activity was over 5 times higher comparing to the data from multi-well plate cultivation, due to the more effective protein secretion in shake flask. When laminarin was used as substrate, activity of *Pp*GH5 increased dramatically to 8522.7 IU/L in the first 72 hours. The increasing behaved steady afterwards, and achieved the highest activity of 9779.8 IU/L at the time point of 120. The enzyme hydrolyzed alkali treated paramylon less effectively. After 24 hours inducing, activity on this recalcitrant substrate reached 2471.6 IU/L. Since then, the enhancement of enzyme activity achieved 5134.9 IU/L at 120 h in a steady pace.

However, the overexpression of this exo- β -1,3-glucanase from *P. pastoris* performed in this work was not the first implement. In 2006, the enzyme *Pp*GH5 (EXG1 in the literature) was cloned and expressed in *P. pastoris* X-33 and showed highest activity toward laminarin was 192 IU/mg (Xu et al. 2006). As unpurified enzyme was used in this work, the relative activity was about 39.1 IU/mg which revealed a great potential for improving. Despite the un-initiative expression, this work was the first time confirming *Pp*GH5's activity on alkali treated paramylon, which met the original purpose of searching paramylon degrading enzymes for future biorefinery research.

145

3.3 Cloning and expression of recombinant lytic polysaccharide monooxygenase in *P. pastoris*

Trail of the newly discovered LPMOs deconstructing paramylon granule were conducted in this part. Two LPMOs, *Tr*AA9 and from *T. reesei* and *Ao*AA11 from *A. oryzae* were cloned and expressed in *P. pastoris*. Moreover, their synergistic actions with β -1,3-glucanase were investigated by determining soluble products generated from paramylon granule.

3.3.1 Extracellular hydrolytic enzyme assay

Biomass of *T. reesei* and *A. oryzae* obtained from pre-cultivation was harvest and transferred to medium with 2.5 g/L paramylon granule as the only carbon source. These experiments were performed aiming at investigation whether these microbes could express paramylon hydrolytic enzymes when the substrate was only available as highly recalcitrant form. The supernatants were used as crude enzymes for activity assay using alkali treated paramylon and laminarin as substrates, respectively. The results were showed in the following figures (Fig.69 and Fig.70).



Fig.69 Extracellular hydrolytic enzyme assay of *T. reesei* induced by paramylon. During cultivation time, supernatants were taken as crude

enzyme solutions for activity assay (section 2.5.1). 100 μ L enzyme solution was incubated in 1 mL of 1 g/L alkali treated paramylon (pH 6,5) and 1 mL of 1 g/L laminarin (pH 6,5) for 16 hours at 37 °C, respectively. The reducing sugar equivalents were measured by 3,5-dinitrosalicylic acid (DNS) method (section 2.3.5.1).



Fig.70 Extracellular hydrolytic enzyme assay of *A. oryzae* induced by paramylon. During cultivation time, supernatants were taken as crude enzyme solutions for activity assay (section 2.5.1). 100 µL enzyme solution was incubated in 1 mL of 1 g/L alkali treated paramylon (pH 6,5) and 1 mL of 1 g/L laminarin (pH 6,5) for 16 hours at 37 °C, respectively. The reducing sugar equivalents were measured by 3,5-dinitrosalicylic acid (DNS) method (section 2.3.5.1).

At the time point of 0 h, crude enzyme samples from *T. reesei* and *A. oryzas* both generated reducing sugars from laminarin and alkali treated paramylon as reducing sugar equivalent of around 0.1 mmol/L. These reducing products were probably glucose residual from the step of medium changing. During 144 hours of cultivation, paramylon granule induced both strains to secret hydrolytic

enzymes which could degrade β -1,3-glucan and generating increasing reducing groups. Enzymes from both *T. reesei* and *A. oryzae* supernatants had stronger activity on laminarin than alkali treated paramylon. After 144 hours cultivation, hydrolytic enzymes secreted by *T. reesei* generated reducing sugar equivalent of 0.8 mmol/L from laminarin in the assay, which was a significantly higher than 0.49 mmol/L from alkali treated paramylon. The results from *A. oryzae* crude enzyme assay showed the same trend. From laminarin and alkali treated paramylon, 0.45 and 0.30 mmol/L of reducing sugar equivalents were hydrolyzed, respectively. The cause of this behavior might be the higher solubility of laminarin than alkali treated paramylon, exposing more accessible hydrolysis sites for glycosidic hydrolyse enzymes. In addition, enzyme mixture produced by *T. reesei* seemed to have better hydrolytic ability on both substrates than enzymes from *A. oryzae*, which may indicate the former strain has better capacity for breaking barrier of recalcitrant substrate.

These results might conform to the hypothesis that some enzymes deconstructed the crystal form of paramylon providing free polysaccharide chain ends for glycoside hydrolyses, when the recalcitrant paramylon was the exclusive carbon source in the medium. Based on current knowledges, the key enzymes should be LPMOs (Vaaje et al. 2010), although no research confirms the existence of LPMO which act on β -1,3-glucan specifically has been reported unti now.

3.3.2 Cloning and expression of *Tr*AA9 (EGR52697.1)

The nucleotide and amino acid sequences of *Tr*AA9 (also names *Cel*61A; GenBank: EGR52697.1) are showing in Fig.22 with a nucleotide sequence of 1035 bp terminated by TAG stop codon, which corresponds to a translation product of 343 amino acids. Analysis of the amino acid sequence of this LPMO candidate showed that a N-terminal signal peptide existed which cleavage site lay between positions 21 (g) and 22 (h) (arrowed in black). The mature protein

contained one catalytic domain (form aa 22 (h) to 256 (n)) belonging to AA9 and one C-terminal carbohydrate binding domain (from aa 308 (a) to 343 (n)) with a linker (from aa 257 (y) to 307 (p), framed in blue) between.

1 atgatecaga agetticeaa eeteetige acegeactgg eggiggetae tggegitgt ggaeatggae atattaatga eatigteate aaeggggigg ggiateagge etatgateet acaaegttie kls nllv talavatgvv ghghin divingv wyqaydp 131 catacgagte aaaceeecce atagtagtgg getggaegge tgeegaeett gaeaaegget tegtteaee egaegeatae eaaaaeeettg acateatetg eeaagaatg eeaagggea pyesnppivv gwt aadl dng fvs pday qnp dii chknatnakg 261 cgcgtctgtc aaggccggag acactattct cttccagtgg gtgccagttc catggccgca ccctggtccc attgtcgact acctggccaa ctgcaatggt gactgcggag acgttgacaa gacgacgctt h a s v k a g d t i l f q w v p v p w p h p g p i v d y l a n c n g d c e t v d k t t l 391 gagttettea agategatgg egttggtete eteageggeg gggateeggg eacetgggee teagaegtge tgateteeaa caacaacaee tgggtegtea agateeega caatettgeg eeageaatt g v g l lsg gdp gtwa s d v lisnnt i p 521 acgtgeteeg eeacgagate ategegttae acagegeegg geaggeaaae ggegeteaga actaeceece gtgetteaae attgeegtet eaggeteggg ttetetgeag eecageggeg ttetagggae y v l r h e i i a l h s a g q a n g a q n y p q c f n i a v s g s g s l q p s g v l g 651 cgacetetat caegegaegg accetggtgt teteateaac atetacaeca gecegeteaa etacateate cetggaeeta eegtggtate aggeetgeea aegagtgttg eeeaggggag eteegeege hat dpg vlin iyt spl nyii pgp 781 acggccaccg ccagcgccac tgttcctgga ggcggtagcg gcccgaccag cagaaccacg acaacggcga ggacgacgca ggcctcaagc aggcccacct tacgcctc cgcaaccacg tcggcacctg t a t a s a t v p g g g s g p t s r t t t t a r t t q a s s r p s s t p p a t t s a p 911 ctggeggeee aacceagaet etgtaeggee agtgtggtgg cageggttae agegggeeta etegatgege geegeeagee aettgeteta eettgaaeee etaetaegee eagtgeetta aetag

Fig.71 Nucleotide sequence and amino acid sequence of TrAA9.

As described in section 2.5.2, the PCR based strategy was used for cloning this putative LPMO. The mature protein was expressed with a 6×histidine tag on the C-terminus in *P. pastoris*. As one intron exited in encoding region, the cDNA from *T.reesei* was used as PCR template. A PCR product of 1010 bp was obtained containing a 1002 nucleotide ORF terminated by a TAA stop codon, which encoding 333 amino acids of mature protein with a predicted molecular mass of 34.26 kD. As shown in Fig.72A, lane 1, the bright band about 1000 bp was the target product. Some faint bands smaller than 500 bp were unspecific PCR products having no influence to the following gel cutting. The target shown in Fig.72A, lane 1 was than cleaned up and inserted into pJET 1.2 cloning vector for plasmid construction.



Fig.72 Analysis of *Tr*AA9 gene application and pAaHBgl-*Tr*AA9 construction.

A: Analysis of gene of *Tr*AA9 obtained from cDNA of *T. reesei* by PCR using specific primers. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1: Gene of *Tr*AA9 (EGR52697.1, 1010 bp);

B: Recombinant vector screening by digestion with restriction endonucleases. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1 to 4, results of four recombinant pAaHBgI-*Tr*AA9 plasmids digested by *Xho* I and *Not* I.

The gene was inserted into the vector pAaHBgl between sites *Xho* I and *Not* I. To screen the positive constructions, *Xho* I and *Not* I were used to digest the recombinant plasmids pAaHBgl-*Tr*AA9. Two fragments with the sizes of 7.5 kb and 1.2 kb should be produced when the gene of interest was inserted correctly. The result was showed in Fig.72B. Three plasmids analyzed were all rightly constructed, due to the two right length cleavage products (lane 1-3). The result showing in lane 4 indicated the cloning vector pJET 1.2 with target gene was transformed into *E. coli*. The 3 kb fragment had the size of linearized pJET 1.2 vector, and the shorter fragment shared the same size of target gene as other samples. The recombinant plasmid corresponding to Fig.72B, lane 1 was

selected to transform *P. pastoris* GS115 for protein expression. The selection and small scale cultivation were performed as described in section 2.4.4.1. After methanol induction for 120 hours, proteins in supernatants were analyzed by SDS-PAGE (Fig.73).



Fig.73 SDS-PAGE analysis secreting expression of recombinant *Tr*AA9.

Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); Lane 1-13, 15 μ L culture supernatant of six recombinant *P.pastoris* GS115 strains after methanol induction for 120 hours in small scale using 24 square-well plate (section 2.4.4.1).

Lane 5 and 12 revealed positive results out of 13 assayed samples. The recombinant *Tr*AA9 protein was detected at a molecular mass of about 60 kDa, which was higher than its calculated mass of 35 kDa due to over glycosylation. This phenomenon of molecular mass increasing was also reported when the enzyme was expressed in *P. pastoris* system by other researches, and the activity on phosphoric acid swollen cellulose (PASC) was not influenced generating mainly C1-oxidized cello-oligosaccharides, small amount of C4- and double (C1- and C4-) oxidized products (Tanghe et al. 2015).

The result indicated a successful *Tr*AA9 expression in *P. pastoris* using alpha-mating factor from *S. cerevisiae* (α -MF) as secretion signal. The recombinant *P. pastoris* GS115, that expressed enzyme *Tr*AA9 corresponding

to lane 5, was selected for further research. Although the LPMO *Tr*AA9 was confirmed active on carboxymethylcellulose (CMC), barly-glucan, microcrystalline cellulose (Avicel) and phosphoric acid swollen cellulose (PASC) (Karlsson et al. 2001), there was still no clear evidence whether *Tr*AA9 could deconstruct highly crystallized paramylon. The first priority of this part of work is not to preempt heterogenous expression of this enzyme, but to investigate *Tr*AA9's activity on native paramylon by experimental evidence.

3.3.3 Cloning and expression of AoAA11 (BAE61530.1)

The nucleotide and amino acid sequences of *Ao*AA11, which was the only characterized LPMO of the newly discovered AA11 family from *A. oryzae* namely *Ao*AA11 (GenBank: BAE61530.1) (Hemsworth et al. 2014a), were showed in Fig.74. The nucleotide sequence is 1266 bp long with TAA as stop codon, encoding a 421 amino acids translation product. In this work, the *Ao*AA11 mature protein of 220 residues (from aa 20 (h) to 235 (c) in Fig.74, framed in blue) was selected as the target domain for cloning and expression following the method in section 2.4.3 and 2.4.4.

The genomic DNA of *A. oryzae* was isolated as template for obtaining target gene by PCR using gene specific primers. A PCR product of 696 bp was obtained containing a 669 bp nucleotide ORF encoding the mature protein and a 6×histidine tag at the C-terminus. The predicted molecular mass of *Ao*AA11 was 23.88 kD. As shown in Fig.75A, lane 1, the only bright band represented the gene amplified from genomic DNA of *A. oryzae*. The band was cleaned up and inserted into pJET 1.2 cloning vector for expression vector construction.

1	atgtttagca	aggettteet	ttccgctgct	ctgctcggcg	ctgccgccgt	tgagggtcac	atgatgatgg	cgcagcccgt	tccttacggc	aaggacactc	tcaacaactc	tccacttgcg	gccgatggca
	m f s	k a f	lsaa	1 l g	a a a	v e g h	mmm	aqp	v р у д	k d t	lnn	s p l a	ad g
131	gtgatttccc	gtgcaagttg	aggtccaaca	cttaccaggt	caccgaagag	aacactgccg	ccatcggtca	atcgatgcct	ctgtctttca	ttggtagcgc	tgttcacggc	ggcggatctt	gccaggtcag
	s d f	pckl	rsn	t y q	vtee	n t a	a i g	qsmp	lsf	igs	a v h g	g g s	c q v
261	tctgaccacc	gaccgtgagc	ccacgaagga	ctccaagtgg	atagtcatca	agtcgatcga	aggtggatgt	cctgccaacg	ttgatggtaa	tctttctggt	ggccccacct	ccacgggagc	ttccaagttc
	sltt	dre	ptk	dskw	i v i	k s i	e g g c	pan	vdg	nlsg	gpt	stg	askf
391	acctacacca	tccctgaagg	tattgagcct	ggcaaataca	ccctcgcctg	gacttggttc	aaccgtatcg	gaaaccgtga	gatgtacatg	aactgtgccc	ctcttactgt	caccggtagt	tcttcgaagc
	t y t	і ре	gıep	g k y	tla	wtwf	nrı	gnr	e m y m	nca	plt	vtgs	s s k
521	gtgacgaagt	tcccaaggag	aagacggttg	agaagcgctc	tgctaacttc	cctcccatgt	tcgtcgccaa	tgtgaacggc	tgcaccacca	aggaaggtgt	tgatattcgt	ttccccaatc	ccggttccat
	rde	у рке	k t v	ekr	sant	ppm	f va	n v n g	ctt	k e g	vdir	fpn	pgs
651	cgttgagtac	gctggtgata	agagcaacct	tgcggctgag	ggcagccagg	cctgcactgg	cacccccaca	ttcggtggcg	atggtaacac	cgccggttcc	agtggctcat	ctggcagttc	ttctggaagc
	1 v е у	a g a	K S N	laae	g s q	a c t	gτpτ	гgg	a g n	tags	sgs	sgs	ssgs
781	tcttctggcg	gctccagctc	ttcggctgcc	ggctccggtg	ctactgcacc	tcccgcgcca	gcggtttcct	cgactttggt	ccccaagcct	tcccagtcct	ctgctcctgg	tgtctttgtc	cccaccggct
	ssg	g s s	5 5 d d	g s g	did	ррар	d V S	SLI	v р к р	sqs	s a p	gviv	pug
911	ctcccgccca	gcctacccac	accagegeee	cctcgggtgg	ctctagctcc	ggctctggtt	ccagctctgg	ttccaactct	ggctctagct	ccggctccag	ctcttcctct	agctcttcct	ccagctctgg
	spa	dbru	L S d	psg	g s s s	gsg	5 5 5	gsns	g s s	sgs	5 5 5 5	5 5 5	5 5 5
1041	tgctctgacc	ggatcttgca	gctcggaggg	aacctggaac	tgcattggtg	gatcttcctt	ccagcgctgt	gccaacggac	agtggaccgc	agtgcagcag	atggccactg	gtactgaatg	tactgctggt
	yait	y s c	556	y c w n	CIY	y s s	IQIC	any	y w u	a v y y	mau	yıe	Clay
1171	caggceteca	acctcaagat	caaggccacc	aacctcaagc	cccgcatgct	ccacgagatg	cgtcacagga	agcgcaacta	ccacaaccac	gcttaa			

Fig.74 Nucleotide sequence and amino acid sequence of AoAA11.





A: Analysis of gene of *Ao*AA11 obtained from genomic DNA of *A. oryzae* by PCR using specific primers. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1: Gene of LPMO *Ao*AA11 (696 bp);

B: Recombinant vector screening by digestion with restriction endonucleases. M: 1 kb DNA ladder (Plasmid Factory, Germany); lane 1: Results of constructed pAaHBgl-*Ao*AA11 plasmids digested by *Xho* I and *Not* I. In Fig.75B, only one plasmid (lane 1) was constructed correctly, confirmed by the length of cleavage products, which were 7.5 kb and 700 bp. In lane 3 and 4, these two plasmids were vector pAaHBgl. There was only one band in lane 2, which may due to low loading dosage causing an invisible short fragment. Therefore, the recombinant plasmid corresponding to Fig.75B, lane 1 was chosen for *P. pastoris* GS115 transformation. Selection and cultivation of recombinant *P. pastoris* were conducted as mentioned before. After 120 hours methanol induction, proteins in supernatants were analyzed by SDS-PAGE.



Fig.76 SDS-PAGE analysis secreting expression of recombinant AoAA11.

Lane M, PageRuler Prestained protein ladder (Part No.26616, Thermo scientific, USA); Lane 1-11, 15 μ L culture supernatant of six recombinant *P. pastoris* GS115 strains after methanol induction for 120 hours in small scale using 24 square-well plate (section 2.4.4.1).

By comparing proteins in supernatant from 11 transformers, recombinant *Ao*AA11 was successfully over-expressed by three of them (Lane 2, 4 and 5), as a soluble protein. And molecular mass of the target protein was about 24 kDa revealing by the SDS-PAGE, which met the predicted 23.88 kDa. As reported, *Ao*AA11 was cloned and expressed via periplasmic secretion in *E*.

coli (Hemsworth et al. 2014b). The *E. coli* expressed *Ao*AA11 was active on chitin, generating aldonic acid oligosaccharides with even-numbered degrees of polymerization by oxidation at C1, but no activity on other substrates, such as cellulose and starch. The following work further confirmed this conclusion.

3.3.4 Synergistic Action of LPMO and β-1,3-Glucanase

LPMOs had a major significance in biomass conversion, due to the ability to boost the activity of classical glucoside hydrolyses. In nature, the most known renewable material of the second-generation bio-ethanol, cellulose, was degraded by the cooperation of enzymes, including various hydrolytic enzymes in different glycoside hydrolase families: endo-glucanase, which created free chain ends (Ljungdahl 2008); cellubiohydrolase cleaving cellobiose from free chain ends of substrate (Sanchez 2009), and β -glucosidase which produced glucose by hydrolyzing cellobiose (Pothiraj et al. 2006). In addition to these major glycoside hydrolases, the non-hydrolytic auxiliary enzyme, lytic polysaccharide monooxygenase, was confirmed as a significant member to improve the substrate conversion to biorefinery products (Courtade et al. 2016). Mechanism of this boosting effect was assumed as introduction of new chain breaks in crystalline substrate, providing free substrates for hydrolytic enzymes and enhancing biomass degradation (Dimarogona et al. 2013). Meanwhile, this interesting finding offered profound curiosity, whether TrAA9 or AoAA11 could improve recalcitrant paramylon deconstruction when incorporate with β -1,3glucanases? In this work, trials of synergistic action between LPMO and β -1,3glucanase were performed. Commercial enzyme (laminarinase), endo- and exo- β -1,3-glucanase expressed in this work were synergistically acted on paramylon granules with TrAA9 or AoAA11 respectively for 72 hours, followed by oligosaccharides analysis.

The qualitative analysis of soluble products generated by LPMOs were commonly performed by two methods. The first method used was matrix assisted laser desorption/ionization-time-of-flight mass spectrometry (MALDI-TOF MS) which detects oligosaccharides in different degree of polymerization (Vaaje Kolstad et al. 2010). And the other way was high-performance anionexchange chromatography coupled with pulsed amperometric detection (HPAEC-PAD) (Forsberg et al. 2011). In this method, the compatibility of electrochemical detection with gradient elution, coupled with the high selectivity of the anion-exchange stationary phases, allowed mixtures of mono-, oligo-and polysaccharides to be separated with high resolution.

In this part of work, a convenient and sensitive colorimetric method, phenolsulfuric acid method, was applied to determine the oligosaccharides generated by synergistic action of LPMOs and β -1,3-glucanases from insoluble paramylon granule. The concentrated sulfuric acid broke glycosidic bond of polysaccharides, oligosaccharides and disaccharides to monomers, then dehydrated the product to hydroxymethylfurfural (HMF). HMF reacted with phenol generating a yellow-gold colored product which could be measured at 490 nm (DuBois et al. 1956). A microtiter plate format of the method had been developed for pure sugar, oligosaccharide or polysaccharide solutions in a sensitive linear range of substrate (1-150 nmol) (Masuko et al. 2005). However, in enzymatic reaction mixture, the protein concentration may significantly affect the absorption measurements (Agbenorhevi and Kontogiorgos 2010). The various amounts of ammonium sulphate, sulphur and carbondioxide produced by adding sulphuric acid to proteins would interfere with the overall chemistry of the reactions.

Taking this aspect into consideration, dialysis membranes and vivaspins were used for reducing the impact of proteins. Small proteins were removed from the enzyme samples by dialysis using a tubular membrane. Samples taken at incubation time points of 0 and 72 h were centrifuged using ultrafiltration spin, which only allowed potentially generated oligosaccharides with low polymerize degree to pass the membrane. The collected filtrate was analyzed then by phenol-sulfuric acid method. For confirming the ability of *Tr*AA9 and *Ao*AA11 to improve the activity of β -1,3-glucanase on paramylon granule, laminarinase (100 mg/L) collaborated with *Tr*AA9 and *Ao*AA11 respectively. 100 mg/L BSA was mixed with laminarinase as negative control. Paramylon granule was added to all enzyme combinations to a final concentration of 1.0 g/L. The detailed procedure was described in section 2.5.3.

Revealing by glucose equivalent, the carbohydrate content produced by combinations of laminarinase with BSA and *Tr*AA9 both increased. As a negative control, BSA was incapable of boosting the activity of laminarinase. The unexpected increasing of glucose equivalent by 47.6 % may due to protein denaturation after incubation at 37 °C for 72 hours, causing impurities in the filtrate and affecting the assay. The result of enzyme combination of laminarinase and *Ao*AA11 even showed a decreasing of 19.0 % in carbohydrate assay, which further demonstrated that the measurement could be easily affected and negating *Ao*AA11's oxidative ability on crystalline paramylon.



Fig.77 Synergistic action of laminarinase and LPMOs. Experimental conditions were described in section 2.5.3.

Besides that, cooperation of laminarinase and *Tr*AA9 generated a significant carbohydrate accumulation. The ratio of glucose equivalents increased up to 97.1% to 17 mg/L after 72 h synergistic action between LPMO and β -1,3-glucanase, which was most likely a confirming of *Tr*AA9's activity towards paramylon granule. Considering that the measurement of glucose equivalent might be inaccurate, the result from the enzymatic reaction was not sufficient to draw the conclusion, that *Tr*AA9 actively stimulated the hydrolytic reaction of laminarinase. But the trend of sharp increasing could be regarded as a sign to reconsider the substrate specificity of *Tr*AA9.

The facilitating ability of *Tr*AA9 boosting oligosaccharide accumulation from paramylon granule when acting with recombinant endo- and exo- β -1,3-glucanase were also investigated.



Fig.78 Synergistic action of *Tr*AA9 and recombinant β -1,3-glucanases. Experimental conditions were described in section 2.5.3.

As showing in Fig.78, combination of *Tr*AA9 (LPMO) and *Tr*GH64 (endo- β -1,3-glucanase) generated a 27.3% ratio of glucose equivalent increasement. Meanwhile, combination of *Tr*AA9 (LPMO) and *Pp*GH5 (exo- β -1,3-glucanase) also slightly enhanced paramylon deconstruction, by additional 10.7 % glucose equivalent after synergistic action for 72 hours. However, it was suspectable that these proportions were approaching the error range. Combination of *Tr*AA9 and BSA acted as negative control, resulting a glucose equivalent decreasing of 19.1%. Nevertheless, as a positive control, combination of *Tr*AA9 and laminarinase generated the highest conversion, resulting a 104.3 % increasement of glucose equivalent. It was consistent with the result in Fig.77, and provoked a keen curiosity over the substrate specificity of LPMOs.

TrAA9 belonged to a cellulose degrading LPMO family AA9, while AoAA11 was a chitin degrading LPMO family AA11 member. The substrate specificity of different families was believed maintaining unity, until researchers discovered that some AA10 members could also degrade cellulose (Forsberg et al. 2011). Actually, the mechanism of LPMOs' substrate specificity remained unconfirmed. Many LPMOs interacted with crystalline substrates at relatively flat surfaces. AA9 enzymes achieved this binding through interactions with aromatic residues in the CBM-like motif (Harris et al. 2010a). For AA10 enzymes, the mechanism of substrate binding was revealed through polar interactions with hydrophilic residues (Aachmann et al. 2012). Since then, researches generally agreed that the specificity of LPMOs were defined by the ability of enzymes to bind substrates. In 2014, structures of cellulose-degrading AA10s were published, revealing a clear configuration difference in active sites, as ScAA10_C and ScAA10_B resembled active site structure of AA10 and AA9, respectively (Forsberg et al. 2014b). Furthermore, the cellulose-active ScAA10 (Book et al. 2014) was found binding chitin without degrading function (Forsberg et al. 2014a). Based on this, it was proposed that enzyme functionality of LPMOs and substrate binding affinity were hardly correlated, but that the substrate specificity depended on active center configuration. More interestingly, EPR spectrum of AoAA11 appeared to resemble a group of cellulose-active LPMOs, but had experimentally measured chitinolytic active

(Hemsworth et al. 2014b). Up to date, the substrate specificity of LPMOs were mainly based on experimental evidences without generally confirmed criteria, leaving a challenge of further structural and mutagenesis studies.

After being neglected for decades, LPMO, the oxidative enzyme that degrade recalcitrant polysaccharides became supreme since its "discovery" in 2010. The abilities of LPMOs to enhance the biomass conversion and lower the cost of deconstruction enzyme cocktails made it a milestone in biorefinery field by deconstructing recalcitrant materials, such as cellulose, hemicellulose, chitin, and lignin without power consuming physical pretreatments, and synergistically acting with hydrolases. The combination of LPMO and cellulase resulted in improvement of glucose yields between two- to eight-fold from cellulose (Langston et al. 2011). The study of LPMOs continued to progress rapidly, and kept going deeper in biology and application. But up to date, there was still no LPMO being confirmed having a substrate specificity for β -1,3-glucans, such as paramylon. Based on the desire of utilize crystalline paramylon as raw material for biorefinery production, the challenge is not whether LPMOs should be used in paramylon conversion, but rather which one can deconstruct the substrate? Although the results from this work were hardly reliable to answer this problem, the basic knowledge and recombinant LPMO expression lay the foundation for further research in this field.

4. Conclusion and Outlook

As the first part of this work, investigation and partial purification of *E. gracilis* extracellular proteins were conducted. Extract prepared from E. gracilis cultivation in late stationary phase rapidly hydrolyzes alkali treated or 2x lyophilized paramylon. The 50x concentrated extract showed the highest hydrolase activity when incubated at pH 5.5 and 37 °C for 16 hours. After partial purification for enzymes responsible for paramylon degradation, one protein potentially belonging to glycoside hydrolase family 22 was found. The original mRNA template was isolated for cDNA library synthesis, and roled as template in degenerate PCR. However, the full amino acid or gene sequence of the target protein are yet to obtain. Paramylon as a storage polysaccharide synthesized by *E. gracilis* is an important renewable raw material for bioenergy producing in the future. Although structural changes of paramylon caused by alkali treatment of native granule increased the hydrolyze activity of cell free extract, no reaction product was obtained from untreated paramylon. So far, however, an exo- and one endo-β-1,3-glucanase were reported (Barras and Stone 1969; Takeda et al. 2015), variety and property of enzymes in the cell free extract responsible of paramylon degradation still need further research. Whatsmore, no heterogeneous enzyme has been reported capable of hydrolyzing paramylon yet, thus leaves numerous choices of paramylon depolymerizer from biomass conversion microbes, such as T. reeei, Bacillus subtilis, etc..

The bioconversion of paramylon into simple fermentable sugars requires synergistic interaction of at least two types of glycoside hydrolases: endo- and exo- β -1,3-glucanase. Functionally, the endo- β -1,3-glucanase disrupts the substrate at random internal sites in the polysaccharide, generating chain ends for exo- β -1,3-glucanase. In second part of this work, six β -1,3-glucanases

including four endo- β -1,3-glucanases and two exo- β -1,3-glucanases were selected. The endo- β -1,3-glucanases are from GH family 16, 55, 64, and 81, and the exo- β -1,3-glucanases are from GH family 5 and 17. These enzymes were successfully cloned and expressed in *P. pastoris*, and confirmed their activities on laminarin, and more importantly, the alkali alkali treated paramylon. Aside from the exo- β -1,3-glucanase *Pp*GH5, all enzymes were expressed in *P. pastoris* for the first time.

P.pastoris was reported having many advantages over other expression systems, including simple molecular manipulation, promising expression, fast growth, ability for posttranslational modifications, and docility for large scale fermentation (Li et al. 2007). Recombinant protein expression in *P. pastoris* can be improved using various strategies, as protein expression is influenced by multiple parameters, such as divergence of codon usage, cultivation conditions including temperature, pH, composition of the culture medium, inducer concentration, and inducing duration. Codon usage optimization was believed as a main factor affecting the successful expression of foreign proteins in different hosts (Saravana Perumal et al. 2016). Codon optimization process involves retranslating the nucleotide sequence of the wild type gene by replacing codons having low frequencies in the expression host with more frequently used ones. Rare codons decrease mRNA stability and translation rate, while high G+C contents could cause reduced translation rate or failed expression (Sinclair and Choy 2002). In P. pastoris, codon optimization technique has been widely used (Akbarzadeh et al. 2014), resulting in significant improvement of enzyme expression in *P. pastoris* (Sun et al. 2016). For further work, strategy of gene condon optimized could be conducted primarily to improve recombinant protein productivity, as in this work all enzymes from *T.reesei* were expressed using wild type genes. Besides this, cultivation parameters could be optimized by designing experiments using response surface methodology. The methodology has been widely used for studying the interactions of several parameters during biotechnological processes (Muntari et al. 2012). Recombinant enzymes expressed in this work were studied preliminarily using supernatant as crude enzymes. With integrated hexa-histidine tag, purifications could be conducted easily for characterizations. For instance, the effects of temperature, pH, substrate concentration, metal ions and inhibitors on enzyme activity.

In this work, endo- and exo- β -1,3-glucanases were expressed and confirmed their activities on alkali treated paramylon. However, glycoside hydrolases can hardly act on native paramylon granule. As the last part of this work, two lytic polysaccharide monooxygenases from T. reesei and A. oryzae were expressed in P. pastoris. The investigation of synergistic action between LPMOs and glycoside hydrolases toward native paramylon were performed. Firstly. inductions of *T. reesei* and *A. oryzae* were performed using paramylon granule as carbon source. The increasing hydrolytic activities on both laminarin and alkali treated paramylon along the cultivating duration indicated that these two microbes expressed enzymes, which were able to deconstruct paramylon. From both strains, total glucanase activity toward laminarin were significantly higher than acting on alkali treated paramylon due to the difference of substrate solubility. And on both substrates, enzymes secreted by T. reesei demonstrated better hydrolytic ability than A. oryzae. Laminarin was easier to access, resulting in reducing sugar equivalent of 0.8 mmol/L for T. reesei enzymes and 0.45 mmol/L for A. oryzae enzymes. While using alkali treated paramylon as enzyme substrate, the result were 0.49 mmol/L for T. reesei enzymes and 0.3 mmol/L for A. oryzae enzymes. In the secreted enzyme cocktail, LPMO is supposed to be the key role which breaks the barrier of paramylon's highlycrystallization. For further confirming the hypothesis, two LPMOs, TrAA9 and from T. reesei and AoAA11 from A. oryzae were cloned and expressed in P. pastoris. Results from SDS-PAGE revealed that recombinant TrAA9 and AoAA11 were both successfully expressed in P. pastoris. In the experiment of synergistic action of LPMOs and β -1,3-glucanases, *Tr*AA9 acted as improving the activity of laminarinase and recombinant β -1,3-glucanases. Particularly, the combination of *Tr*AA9 and laminarinase yielded an increasing of about 100.0% glucose equivalent after incubation for 72 hours. On the contrary, the result of *Ao*AA11 acting on paramylon granule with laminarinase revealed a decreasing trend. This result confirmed *Ao*AA11's chitin specificity, and on the other hand, indicated the unreliability of phenol sulfuric acid method. In future, accurate analysis method such as MALDI-TOF MS and HPAEC-PAD should be involved for confirming *Tr*AA9's enhancing function to β -1,3-glucanases. Additionally, enzymes are definitely expressed in *E. gracilis* to deconstruct paramylon for autotrophic nutrition. The intracellular proteins could be resources for screening such enzymes. Along with the studied LPMO, for instance *Tr*AA9 in this work, these enzymes need further research of deconstructing abilities on native paramylon granule in order to achieve the goal of low-cost and efficient degradation.

References

- Aachmann FL, Sorlie M, Skjak BG, Eijsink VGH, Vaaje KG. (2012): NMR structure of a lytic polysaccharide monooxygenase provides insight into copper binding, protein dynamics, and substrate interactions. In *Proceedings of the national academy of sciences of the united states of america* 109 (46), 18779–18784. DOI: 10.1073/pnas.1208822109.
- Aam BB, Heggset EB, Norberg AL, Sorlie M, Varum KM, Eijsink VGH. (2010): Production of chitooligosaccharides and their potential applications in medicine. In *Marine drugs* 8 (5), 1482–1517. DOI: 10.3390/md8051482.
- Adamian L, Naveed H, Liang J. (2011): Lipid-binding surfaces of membrane proteins: evidence from evolutionary and structural analysis. In *Biochimica et biophysica acta* 1808 (4), 1092–1102. DOI: 10.1016/j.bbamem.2010.12.008.
- Aditiya HB, Mahlia T, Chong WT, Nur H, Sebayang AH. (2016): Second generation bioethanol production: A critical review. In *Renewable and sustainable energy reviews* 66, 631–653. DOI: 10.1016/j.rser.2016.07.015.
- Agbenorhevi JK, Kontogiorgos V (2010): Polysaccharide determination in protein/polysaccharide mixtures for phase-diagram construction. In *Carbohydrate polymers* 81 (4), 849–854. DOI: 10.1016/j.carbpol.2010.03.056.
- Ahmad M, Hirz M, Pichler H, Schwab H. (2014): Protein expression in *Pichia pastoris*: recent achievements and perspectives for heterologous protein production. In *Applied microbiology and biotechnology* 98 (12), 5301–5317. DOI: 10.1007/s00253-014-5732-5.
- Akbarzadeh A, Ranaei Siadat SO, Motallebi M, Zamani MR, Barshan Tashnizi M, Moshtaghi S. (2014): Characterization and high level expression of acidic endoglucanase in *Pichia pastoris*. In *Applied biochemistry and biotechnology* 172 (4), 2253–2265. DOI: 10.1007/s12010-013-0672-6.
- Al AQ, Nixon BT, Fortwendel JR. (2016): The Enzymatic Conversion of Major Algal and Cyanobacterial Carbohydrates to Bioethanol. In *Frontiers in energy research.* 4 (3), 1844. DOI: 10.3389/fenrg.2016.00036.
- Almeida JRM de, Moraes LMP de, Torres FAG. (2005): Molecular characterization of the 3-phosphoglycerate kinase gene (PGK1) from the methylotrophic yeast *Pichia pastoris*. In *Yeast* 22 (9), 725–737. DOI: 10.1002/yea.1243.
- Anthon GE, Barrett DM. (2002): Determination of reducing sugars with 3-methyl-2benzothiazolinonehydrazone. In *Analytical biochemistry* 305 (2), 287–289. DOI: 10.1006/abio.2002.5644.

Barras DR, Stone BA. (1969): β-1,3-Glucan hydrolases from Euglena gracilis. In

Enzymology 191 (2), 329-353. DOI: 10.1016/0005-2744(69)90252-6.

- Barsanti L, Bastianini A, Passarelli V, Tredici MR, Gualtieri P. (2000): Fatty acid pattern in wild type and WZSL mutant of *Euglena gracilis*: Effects of carbon sources and growth conditions. In *Journal of applied phycology* 12 (3/5), 515–520. DOI: 10.1023/A:1008187514624.
- Barsanti L, Vismara R, Passarelli V, Gualtieri P. (2001): Paramylon (beta-1,3-glucan) content in wild type and WZSL mutant of *Euglena gracilis*. Effects of growth conditions. In *Journal of applied phycology* 13 (1), 59–65. DOI: 10.1023/A:1008105416065.
- Beeson WT, Phillips CM, Cate JHD, Marletta MA. (2012): Oxidative cleavage of cellulose by fungal copper-dependent polysaccharide monooxygenases. In *Journal of the american chemical society* 134 (2), 890–892. DOI: 10.1021/ja210657t.
- Bianchetti CM, Takasuka TE, Deutsch S, Udell HS, Yik EJ, Bergeman LF, Fox BG. (2015): Active site and laminarin binding in glycoside hydrolase family 55. In *Journal* of biological chemistry. DOI: 10.1074/jbc.M114.623579.
- Book AJ, Yennamalli RM, Takasuka TE, Currie CR, Phillips GN, JR, Fox BG. (2014):
 Evolution of substrate specificity in bacterial AA10 lytic polysaccharide
 monooxygenases. In *Biotechnology for biofuels* 7, 109. DOI: 10.1186/1754-6834-7-109.
- Bũmer D, Preisfeld A, Ruppel HG. (2001): Isolation and characterization of paramylon synthase from *Euglena gracilis*. In *Journal of phycology* 37 (1), 38–46. DOI: 10.1046/j.1529-8817.2001.037001038.x.
- Cereghino JL, Cregg JM. (2000): Heterologous protein expression in the methylotrophic yeast *Pichia pastoris*. In *Federation of european microbiological societies microbiology reviews* 24 (1), 45–66.
- Cereghino GPL, Cereghino JL, Sunga AJ, Johnson MA, Lim M, Gleeson MA, Cregg JM. (2001): New selectable marker/ auxotrophic host strain combinations for molecular genetic manipulation of *Pichia pastoris*. In *Gene* 263 (1), 159–169.
- Cereghino GPL, Godfrey L, La CB, Johnson S, Khuongsathiene S, Tolstorukov I, Yan M, Cereghino JL, Veenhuis M, Subramani S, Cregg JM. (2006): Mxr1p, a key regulator of the methanol utilization pathway and peroxisomal genes in *Pichia pastoris*. In *Molecular and cellular biology* 26 (3), 883–897. DOI: 10.1128/MCB.26.3.883-897.2006.
- Chaturvedi V, Verma P. (2013): An overview of key pretreatment processes employed for bioconversion of lignocellulosic biomass into biofuels and value added products. In *Biotechology* 3 (5), 415–431. DOI: 10.1007/s13205-013-0167-8.
- Chen CY, Zhao XQ, Yen HW, Ho SH, Cheng CL, Lee DJ, Bai FW, Chang JS. (2013): Microalgae-based carbohydrates for biofuel production. In *Biochemical engineering Journal* 78, 1–10. DOI: 10.1016/j.bej.2013.03.006.

- Chomczynski P, Sacchi N. (1987): Single-step method of RNA isolation by acid guanidinium thiocyanate-phenol-chloroform extraction. In *Analytical biochemistry* 162 (1), 156–159. DOI: 10.1006/abio.1987.9999.
- Chuen IS, Lynch HC. (1998): Production of a yeast cell wall degrading enzyme, β -1,3-glucanase by recombinant *Bacillus subtilis*. In *Biochemical society transactions* 26 (2), 175. DOI: 10.1042/bst026s175.
- Chung CT; Niemela SL; Miller RH. (1989): One-step preparation of competent Escherichia coli: transformation and storage of bacterial cells in the same solution. In Proceedings of the national academy of sciences of the united states of america 86 (7), 2172–2175. Available online at https://www.ncbi.nlm.nih.gov/pmc/articles/PMC286873/.
- Clarke AE, Stone BA. (1960): Structure of the paramylon from *Euglena gracilis*. In *Biochimica et biophysica acta* 44, 161–163.
- Cos O, Ramon R, Montesinos JL, Valero F. (2006): Operational strategies, monitoring and control of heterologous protein production in the methylotrophic yeast *Pichia pastoris* under different promoters: a review. In *Microbial cell factories* 5,17. DOI: 10.1186/1475-2859-5-17.
- Courtade G, Wimmer R, Rohr AK, Preims M, Felice AKG, Dimarogona M, Vaaje KG, Sorlie M, Sandgren M, Ludwig R, Eijsink VGH, Aachmann FL. (2016): Interactions of a fungal lytic polysaccharide monooxygenase with beta-glucan substrates and cellobiose dehydrogenase. In *Proceedings of the national academy of sciences of the united states of america* 113 (21), 5922–5927. DOI: 10.1073/pnas.1602566113.
- Cregg JM, Barringer KJ, Hessler AY, Madden KR. (1985): *Pichia pastoris* as a host system for transformations. In *Molecular and cellular biology* 5 (12), 3376–3385.
- Cregg JM, Cereghino JL, Shi J, Higgins DR. (2000): Recombinant protein expression in *Pichia pastoris*. In *Molecular biotechnology* 16 (1), 23–52. DOI: 10.1385/MB:16:1:23.
- Cregg JM, Madden KR, Barringer KJ, Thill GP, Stillman CA. (1989): Functional characterization of the two alcohol oxidase genes from the yeast *Pichia pastoris*. In *Molecular and cellular biology* 9 (3), 1316–1323.
- Cregg J, Tolstorukov I. (2012): *P. pastoris* ADH promoter and use thereof to direct expression of proteins. United States Patent. Patent number: US 8222386.
- Cutfield SM, Davies GJ, Murshudov G, Anderson BF, Moody PC, Sullivan PA, Cutfield JF. (1999): The structure of the exo-beta-(1,3)-glucanase from *Candida albicans* in native and bound forms: relationship between a pocket and groove in family 5 glycosyl hydrolases. In *Journal of molecular biology* 294 (3), 771–783. DOI: 10.1006/jmbi.1999.3287.
- Daly R, Hearn MTW. (2005): Expression of heterologous proteins in *Pichia pastoris*. In *Journal of molecular recognition* 18 (2), 119–138. DOI: 10.1002/jmr.687.
- Daroch M, Geng S, Wang GY. (2013): Recent advances in liquid biofuel production from

algal feedstocks. In *Applied Energy* 102, 1371-1381. DOI: 10.1016/j.apenergy.2012.07.031.

- Davies G, Henrissat B. (1995): Structures and mechanisms of glycosyl hydrolases. In *Structure* 3 (9), 853–859. DOI: 10.1016/S0969-2126(01)00220-9.
- Davies GJ, Sinnott ML. (2008): Sorting the diverse: The sequence-based classifications of carbohydrate-active enzymes. In *Biochemical journal* 30 (4), 26–32. DOI: 10.1042/BJ20080382.
- Dean N. (1999): Asparagine-linked glycosylation in the yeast Golgi. In *Biochimica et biophysica acta* 1426 (2), 309–322.
- Delic M, Mattanovich D, Gasser B. (2013): Repressible promoters: a novel tool to generate conditional mutants in *Pichia pastoris*. In *Microbial cell factories* 12, 6. DOI: 10.1186/1475-2859-12-6.
- Dellacorte C. (1994): Isolation of nucleic acids from the sea anemone *Condylactis* gigantea (Cnidaria: Anthozoa). In *Tissue and cell* 26 (4), 613–619.
- Dimarogona M, Topakas E, Christakopoulos P. (2013): Recalcitrant polysaccharide degradation by novel oxidative biocatalysts. In *Applied microbiology and biotechnology* 97 (19), 8455–8465. DOI: 10.1007/s00253-013-5197-y.
- Drocourt D, Calmels T, Reynes JP, Baron M, Tiraby G. (1990): Cassettes of the Streptoalloteichus hindustanus ble gene for transformation of lower and higher eukaryotes to phleomycin resistance. In *Nucleic acids research* 18 (13), 4009.
- DuBois M, Gilles KA, Hamilton JK, Rebers PA, Smith F. (1956): Colorimetric Method for Determination of Sugars and Related Substances. In *Analytic chemistry* 28 (3), 350– 356. DOI: 10.1021/ac60111a017.
- Duman JG, Miele RG, Liang H, Grella DK, Sim KL, Castellino FJ, Bretthauer RK. (1998):
 O-Mannosylation of *Pichia pastoris* cellular and recombinant proteins. In *Biotechnology and applied biochemistry* 28 (1), 39–45.
- Ehrenberg CG. (1830): Organisation, systematik und geographisches verhältniss der infusionsthierchen. Berlin: Druckerei der Königlichen akademie der wissenschaften, 1795–1876.
- Ellis SB, Brust PF, Koutz PJ, Waters AF, Harpold MM, Gingeras TR. (1985): Isolation of alcohol oxidase and two other methanol regulatable genes from the yeast *Pichia pastoris*. In *Molecular and cellular biology* 5 (5), 1111–1121.
- Erickson HP. (2009): Size and shape of protein molecules at the nanometer level determined by sedimentation, gel filtration, and electron microscopy. In *Biological procedures online* 11, 32–51. DOI: 10.1007/s12575-009-9008-x.
- Eriksson KE, Pettersson B, Westermark U. (1974): Oxidation: An important enzyme reaction in fungal degradation of cellulose. In *Federation of european biochemical societies letters* 49 (2), 282–285. DOI: 10.1016/0014-5793(74)80531-4.

- Eugene. (2015): The evolution and diversity of life. *In Schoolbag.info online*. Website: http://schoolbag.info/biology/living/140.html
- Evangelista V, Passarelli V, Barsanti L, Gualtieri P. (2003): Fluorescence Behavior of Euglena Photoreceptor. In Photochemistry and photobiology 78 (1), 93–97. DOI: 10.1562/0031-8655(2003)0780093FBOEP2.0.CO2
- Fibriansah G, Masuda S, Koizumi N, Nakamura S, Kumasaka T. (2007): The 1.3 A crystal structure of a novel endo-beta-1,3-glucanase of glycoside hydrolase family 16 from alkaliphilic *Nocardiopsis* sp. strain F96. In *Proteins* 69 (3), 683–690. DOI: 10.1002/prot.21589.
- Forsberg Z, Mackenzie AK, Sorlie M, Rohr AK, Helland R, Arvai AS, Vaaje-Kolstad G, Eijsink VGH. (2014a): Structural and functional characterization of a conserved pair of bacterial cellulose-oxidizing lytic polysaccharide monooxygenases. In *Proceedings of the national academy of sciences of the united states of america* 111 (23), 8446–8451. DOI: 10.1073/pnas.1402771111.
- Forsberg Z, Nelson CE, Dalhus B, Mekasha S, Loose JSM, Crouch LI, Rohr AK, Gardner JG, Eijsink VGH, Vaaje-Kolstad G. (2016): Structural and Functional Analysis of a Lytic Polysaccharide Monooxygenase Important for Efficient Utilization of Chitin in *Cellvibrio japonicus*. In *Journal of biological chemistry* 291 (14), 7300–7312. DOI: 10.1074/jbc.M115.700161.
- Forsberg Z, Rohr AK, Mekasha S, Andersson KK, Eijsink VGH, Vaaje-Kolstad G, Sorlie M. (2014b): Comparative study of two chitin-active and two cellulose-active AA10type lytic polysaccharide monooxygenases. In *Biochemistry* 53 (10), 1647–1656. DOI: 10.1021/bi5000433.
- Forsberg Z, Vaaje-Kolstad G, Westereng B, Bunaes AC, Stenstrom Y, MacKenzie A, Sorlie M, Horn SJ, Eijsink VGH. (2011): Cleavage of cellulose by a CBM33 protein. In *Protein science* 20 (9), 1479–1483. DOI: 10.1002/pro.689.
- Gellissen G. (2000): Heterologous protein production in methylotrophic yeasts. In *Applied microbiology and biotechnology* 54 (6), 741–750.
- Giczey G, Kerenyi Z, Fulop L, Hornok L. (2001): Expression of *cmg1*, an Exo-β-1,3-Glucanase Gene from *Coniothyrium minitans*, Increases during Sclerotial Parasitism. In *Applied and environmental microbiology* 67 (2), 865–871. DOI: 10.1128/AEM.67.2.865-871.2001.
- Gomez LD, Whitehead C, Barakate A, Halpin C, McQueen-Mason SJ. (2010): Automated saccharification assay for determination of digestibility in plant materials. In *Biotechnology for biofuels* 3, 23. DOI: 10.1186/1754-6834-3-23.
- Gottlieb J. (1850): Ueber eine neue, mit Stärkmehl isomere Substanz; von Professor *J.Gottlieb.* In *European journal of organic chemistry* 75 (1), 51–61. DOI: 10.1002/jlac.18500750105.

Gudmundsson M, Kim S, Wu M, Ishida T, Momeni MH, Vaaje-Kolstad G, Lundberg D,

Royant A, Stahlberg J, Eijsink VGH, Beckham GT, Sandgren M. (2014): Structural and electronic snapshots during the transition from a Cu(II) to Cu(I) metal center of a lytic polysaccharide monooxygenase by X-ray photoreduction. In *Journal of biological chemistry* 289 (27), 18782–18792. DOI: 10.1074/jbc.M114.563494.

- Hacquard S, Kracher B, Hiruma K, Munch PC, Garrido-Oter R, Thon MR, Weimann A, Damm U, Dallery JF, Hainaut M, Henrissat B, Lespinet O, Sacristan S, van Loren Themaat E ver, Kemen E, McHardy AC, Schulze-Lefert P, O'Connell RJ. (2016): Survival trade-offs in plant roots during colonization by closely related beneficial and pathogenic fungi. In *Nature communications* 7, 11362. DOI: 10.1038/ncomms11362.
- Harris PV, Welner D, McFarland KC, Re E, Navarro PJ, Brown K, Salbo R, Ding H, Vlasenko E, Merino S, Xu F, Cherry J, Larsen S, Lo LL. (2010): Stimulation of lignocellulosic biomass hydrolysis by proteins of glycoside hydrolase family 61: structure and function of a large, enigmatic family. In *Biochemistry* 49 (15), 3305– 3316. DOI: 10.1021/bi100009p.
- Harris PV, Xu F, Kreel NE, Kang C, Fukuyama S. (2014): New enzyme insights drive advances in commercial ethanol production. In *Current opinion in chemical biology* 19, 162–170. DOI: 10.1016/j.cbpa.2014.02.015.
- Hartl L, Gastebois A, Aimanianda V, Latgé JP. (2011): Characterization of the GPIanchored endo β-1,3-glucanase Eng2 of *Aspergillus fumigatus*. In *Fungal genetics and biology* 48 (2), 185–191. DOI: 10.1016/j.fgb.2010.06.011.
- Harun R, Yip JWS, Thiruvenkadam S, Ghani WAWAK, Cherrington T, Danquah MK.
 (2014): Algal biomass conversion to bioethanol a step-by-step assessment. In *Biotechnology journal* 9 (1), 73–86. DOI: 10.1002/biot.201200353.
- Hauser TR, Cummins RL. (1964): Increasing Sensitivity of 3-Methyl-2-Benzothiazolone
 Hydrozone Test for Analysis of Aliphatic Aldehydes in Air. In *Analytic chemistry* 36 (3), 679–681. DOI: 10.1021/ac60209a067.
- Hemsworth GR, Henrissat B, Davies GJ, Walton PH. (2014): Discovery and characterization of a new family of lytic polysaccharide monooxygenases. In *Nature chemical biology* 10 (2), 122–126. DOI: 10.1038/nchembio.1417.
- Hemsworth GR, Johnston EM, Davies GJ, Walton PH. (2015): Lytic Polysaccharide Monooxygenases in Biomass Conversion. In *Trends in biotechnology* 33 (12), 747– 761. DOI: 10.1016/j.tibtech.2015.09.006.
- Higgins DR, Cregg JM. (1998): Introduction to *Pichia pastoris*. In *Methods in molecular biology* 103, 1–15. DOI: 10.1385/0-89603-421-6:1.
- Holt SC, Stern AI. (1970): The effect of 3-(3,4-dichlorophenyl)-1,1-dimethylurea on chloroplast development and maintenance in *Euglena gracilis*. I. Ultrastructural characterization of light-grown cells by the techniques of thin sectioning and freeze-etching. In *Plant physiology* 45 (4), 475–483.
- Hong TY, Cheng CW, Huang JW, Meng M. (2002): Isolation and biochemical

characterization of an endo-1,3-beta-glucanase from *Streptomyces sioyaensis* containing a C-terminal family 6 carbohydrate-binding module that binds to 1,3-beta-glucan. In *Microbiology* 148 (4), 1151–1159.

- Horn SJ, Vaaje-Kolstad G, Westereng B, Eijsink VG. (2012): Novel enzymes for the degradation of cellulose. In *Biotechnology for biofuels* 5 (1), 45. DOI: 10.1186/1754-6834-5-45.
- Inan M, Meagher MM. (2001): Non-repressing carbon sources for alcohol oxidase (AOX1) promoter of *Pichia pastoris*. In *Journal of bioscience and bioengineering* 92 (6), 585–589. DOI:10.1016/S1389-1723(01)80321-2
- Isaksen T, Westereng B, Aachmann FL, Agger JW, Kracher D, Kittl R, Ludwig R, Haltrich D, Eijsink VGH, Horn SJ. (2014): A C4-oxidizing lytic polysaccharide monooxygenase cleaving both cellulose and cello-oligosaccharides. In *Journal of biological chemistry* 289 (5), 2632–2642. DOI: 10.1074/jbc.M113.530196.
- Ishida T, Fushinobu S, Kawai R, Kitaoka M, Igarashi K, Samejima M. (2009): Crystal structure of glycoside hydrolase family 55 beta-1,3-glucanase from the basidiomycete *Phanerochaete chrysosporium*. In *Journal of biological chemistry* 284 (15), 10100–10109. DOI: 10.1074/jbc.M808122200.
- Jae KP, Kenji M, Ikuo F, Yukikazu Y, Tsuyoshi N, Makoto K, Hideyuki M. (1997): Purification and Characterization of the Chitinase (ChiA) from *Enterobacter* sp. In *Bioscience, biotechnology, and biochemistry* 61 (4), 684–689.
- Jeng WY, Wang NC, Lin CT, Shyur LF, Wang AH. (2011): Crystal structures of the laminarinase catalytic domain from *Thermotoga maritima* MSB8 in complex with inhibitors: essential residues for β-1,3- and β-1,4-glucan selection. In *Journal of biological chemistry* 286 (52), 45030–45040. DOI: 10.1074/jbc.M111.271213.
- Jenkins J, Lo LL, Harris G, Pickersgill R. (1995): Beta-glucosidase, beta-galactosidase, family A cellulases, family F xylanases and two barley glycanases form a superfamily of enzymes with 8-fold beta/alpha architecture and with two conserved glutamates near the carboxy-terminal ends of beta-strands four and seven. In *Federation of european biochemical societies letters* 362 (3), 281–285.
- Jungo C, Rérat C, Marison IW, Stockar UV. (2006): Quantitative characterization of the regulation of the synthesis of alcohol oxidase and of the expression of recombinant avidin in a *Pichia pastoris* Mut+ strain. In *Enzyme and microbial technology* 39 (4), 936–944. DOI: 10.1016/j.enzmictec.2006.01.027.
- Karlsson J, Saloheimo M, Siika-Aho M, Tenkanen M, Penttila M, Tjerneld F. (2001): Homologous expression and characterization of Cel61A (EG IV) of *Trichoderma* reesei. In European journal of biochemistry 268 (24), 6498–6507.
- Kim SK. (2015): Springer Handbook of Marine Biotechnology. Springer Berlin Heidelberg.
- Kimura M, Kamakura T, Zhou TQ, Kaneko I, Yamaguchi I. (1994): Cloning of the blasticidin S deaminase gene (BSD) from *Aspergillus terreus* and its use as a

selectable marker for *Schizosaccharomyces pombe* and *Pyricularia oryzae*. In *Molecular and general genetics* 242 (2), 121–129. DOI: 10.1007/BF00391004.

- Kiss JZ, Roberts EM, Brown RM, Triemer RE. (1988): X-ray and dissolution studies of paramylon storage granules from *Euglena*. In *Protoplasma* 146 (2), 150–156. DOI: 10.1007/BF01405924.
- Kiss JZ, Vasconcelos AC, Triemer RE. (1987): Structure of the *Euglenoid* Storage Carbohydrate, Paramylon. In *American journal of botany* 74 (6), 877. DOI: 10.2307/2443868.
- Kitaoka M, Matsuoka Y, Mori K, Nishimoto M, Hayashi K. (2012): Characterization of a bacterial laminaribiose phosphorylase. In *Bioscience, biotechnology, and biochemistry* 76 (2), 343–348. DOI: 10.1271/bbb.110772.
- Klebl F, Tanner W. (1989): Molecular cloning of a cell wall exo-beta-1,3-glucanase from *Saccharomyces cerevisiae*. In *Journal of bacteriology* 171 (11), 6259–6264.
- Klis FM. (1994): Review: cell wall assembly in yeast. In *Yeast* 10 (7), 851–869. DOI: 10.1002/yea.320100702.
- Koichi O, Hideo N, Masahiro O. (1969): A yeast capable of utilizing methanol. In Agricultural and biological chemistry 33 (10), 1519–1520. DOI: 10.1080/00021369.1969.10859497
- Koizumi N, Sakagami H, Utsumi A, Fujinaga S, Takeda M, Asano K, Sugawara I, Ichikawa S, Kondo H, Mori S. (1993): Anti-HIV (human immunodeficiency virus) activity of sulfated paramylon. In *Antiviral research* 21 (1), 1–14.
- Kondo Y, Kato A, Hojo H, Nozoe S, Takeuchi M, Ochi K. (1992): Cytokine-related immunopotentiating activities of paramylon, a beta-(1-3)-D-glucan from *Euglena* gracilis. In Journal of pharmacobio-dynamics 15 (11), 617–621.
- Koshland DE. (1953): Stereochemistry and the mechanism of enzymatic reactions. In *Biological reviews* 28 (4), 416–436. DOI: 10.1111/j.1469-185X.1953.tb01386.x.
- Krah M, Misselwitz R, Politz O, Thomsen KK, Welfle H, Borriss R. (1998): The laminarinase from thermophilic eubacterium *Rhodothermus marinus* conformation, stability, and identification of active site carboxylic residues by site-directed mutagenesis. In *European journal of biochemistry* 257 (1), 101–111.
- Kranthi BV, Kumar HRV, Rangarajan PN. (2010): Identification of Mxr1p-binding sites in the promoters of genes encoding dihydroxyacetone synthase and peroxin 8 of the methylotrophic yeast *Pichia pastoris*. In *Yeast* 27 (9), 705–711. DOI: 10.1002/yea.1766.
- Kranthi BV, Kumar R, Kumar NV, Rao DN, Rangarajan PN. (2009): Identification of key DNA elements involved in promoter recognition by Mxr1p, a master regulator of methanol utilization pathway in *Pichia pastoris*. In *Biochimica et biophysica acta* 1789 (6-8), 460–468. DOI: 10.1016/j.bbagrm.2009.05.004.

- Kreger DR, Meeuse BJD. (1952): X-ray diagrams of *Euglena*-paramylon, of the acidinsoluble glucan of yeast cell walls and of laminarin. In *Biochimica et biophysica acta* 9 (6), 699–700.
- Kumar NV, Rangarajan PN. (2012): The zinc finger proteins Mxr1p and repressor of phosphoenolpyruvate carboxykinase (ROP) have the same DNA binding specificity but regulate methanol metabolism antagonistically in *Pichia pastoris*. In *Journal of biological chemistry* 287 (41), 34465–34473. DOI: 10.1074/jbc.M112.365304.
- Kunst, A., Draeger, B. & Ziegenhorn, J. (1984): Methods of Enzymatic Analysis 3rd ed. 6, 163-172, Verlag Chemie, Weinheim, Deerfield Beach/Florida, Basel
- Kutty G, Davis AS, Ma L, Taubenberger JK, Kovacs JA. (2015): *Pneumocystis* encodes a functional endo-β-1,3-glucanase that is expressed exclusively in cysts. In *Journal of infectious diseases* 211 (5), 719–728. DOI: 10.1093/infdis/jiu517.
- Kyoto iGEM (2011). Kyoto digestion DNSassay.

Retrieved from http://2011.igem.org/File:Kyoto-digestion-DNSassay1.jpg

- La-Cruz, Pintor-Toro JA, Benítez T, Llobell A, Romero LC. (1995): A novel endo-beta-1,3-glucanase, BGN13.1, involved in the mycoparasitism of *Trichoderma harzianum*. In *Journal of bacteriology* 177 (23), 6937–6945.
- Lambertz C, Garvey M, Klinger J, Heesel D, Klose H, Fischer R, Commandeur U. (2014): Challenges and advances in the heterologous expression of cellulolytic enzymes: a review. In *Biotechnology for biofuels* 7 (1), 135. DOI: 10.1186/s13068-014-0135-5.
- Langston JA, Shaghasi T, Abbate E, Xu F, Vlasenko E, Sweeney MD. (2011): Oxidoreductive cellulose depolymerization by the enzymes cellobiose dehydrogenase and glycoside hydrolase 61. In *Applied and environmental microbiology* 77 (19), 7007–7015. DOI: 10.1128/AEM.05815-11.
- Larriba G, Andaluz E, Cueva R, Basco RD. (1995): Molecular biology of yeast exoglucanases. In *Federation of european microbiological societies microbiology letters* 125 (2-3), 121–126.
- Lee RA, Lavoie JM. (2013): From first- to third-generation biofuels. Challenges of producing a commodity from a biomass of increasing complexity. In *Animal frontiers* 3 (2), 6–11. DOI: 10.2527/af.2013-0010.
- Leedale GF. (1967): Euglenoid Flagellates: Prentice Hall, Englewood Cliffs, New Jersey.
- Leggio LL, Simmons TJ, Poulsen J-CN, Frandsen KEH, Hemsworth GR, Stringer MA, Freiesleben PV, Tovborg M, Johansen KS, Maria L de, Harris PV, Soong C-L, Dupree P, Tryfona T, Lenfant N, Henrissat B, Davies GJ, Walton PH. (2015): Structure and boosting activity of a starch-degrading lytic polysaccharide monooxygenase. In *Nature communications* 6, 5961. DOI: 10.1038/ncomms6961.
- Levasseur A, Drula E, Lombard V, Coutinho PM, Henrissat B. (2013): Expansion of the enzymatic repertoire of the CAZy database to integrate auxiliary redox enzymes. In
Biotechnology for biofuels 6 (1), 41. DOI: 10.1186/1754-6834-6-41.

- Li P, Anumanthan A, Gao XG, Ilangovan K, Suzara VV, Düzgüneş N, Renugopalakrishnan V. (2007): Expression of Recombinant Proteins in *Pichia Pastoris*. In *Appl Biochem Biotechnol* 142 (2), 105–124. DOI: 10.1007/s12010-007-0003-x.
- Liang S, Zou C, Lin Y, Zhang X, Ye Y. (2013): Identification and characterization of P GCW14. A novel, strong constitutive promoter of *Pichia pastoris*. In *Biotechnology letters* 35 (11), 1865–1871. DOI: 10.1007/s10529-013-1265-8.
- Life technologies. (2014): Pichia Expression Kit: for expression of recombinant proteins in *Pichia pastoris*. Available online at https://tools.thermofisher.com/content/sfs/manuals/pich_man.pdf.
- Ljungdahl LG. (2008): The cellulase/hemicellulase system of the anaerobic fungus Orpinomyces PC-2 and aspects of its applied use. In Annals of the New York academy of sciences 1125, 308–321. DOI: 10.1196/annals.1419.030.
- Lombard V, Golaconda Ramulu H, Drula E, Coutinho PM, Henrissat B. (2014): The carbohydrate-active enzymes database (CAZy) in 2013. In *Nucleic acids research* 42 (Database issue), D490-5. DOI: 10.1093/nar/gkt1178.
- Ma Y, Liu X, Yin Y, Zou C, Wang W, Zou S, Hong J, Zhang M. (2015): Expression optimization and biochemical properties of two glycosyl hydrolase family 3 betaglucosidases. In *Journal of biotechnology* 206, 79–88. DOI: 10.1016/j.jbiotec.2015.04.016.
- Marchessault RH, Deslandes Y. (1979): Fine structure of (1,3)-β-d-glucans: curdlan and paramylon. In *Carbohydrate research* 75, 231–242. DOI: 10.1016/S0008-6215(00)84642-X.
- Marin B, Palm A, Klingberg M, Melkonian M. (2003): Phylogeny and Taxonomic Revision of Plastid-Containing Euglenophytes based on SSU rDNA Sequence Comparisons and Synapomorphic Signatures in the SSU rRNA Secondary Structure. In *Protist* 154 (1), 99–145. DOI: 10.1078/143446103764928521.
- Martín-Cuadrado A, Fontaine T, Esteban P, Del Dedo JE, Medina-Redondo M, Del Rey F, Latgé JP, Aldana CRV. (2008): Characterization of the endo-beta-1,3-glucanase activity of *S. cerevisiae* Eng2 and other members of the GH81 family. In *Fungal genetics and biology* 45 (4), 542–553. DOI: 10.1016/j.fgb.2007.09.001.
- Martinez AT. (2016): How to break down crystalline cellulose. In *Science* 352 (6289), 1050–1051. DOI: 10.1126/science.aaf8920.
- Martinez D, Berka RM, Henrissat B, Saloheimo M, Arvas M, Baker SE, Chapman J, Chertkov O, Coutinho PM, Cullen D, Danchin EGJ, Grigoriev IV, Harris P, Jackson M, Kubicek CP, Han CS, Ho I, Larrondo LF, Leon AL de, Magnuson JK, Merino S, Misra M, Nelson B, Putnam N, Robbertse B, Salamov AA, Schmoll M, Terry A, Thayer N, Westerholm-Parvinen A, Schoch CL, Yao J, Barabote R, Barbote R,

Nelson MA, Detter C, Bruce D, Kuske CR, Xie G, Richardson P, Rokhsar DS, Lucas SM, Rubin EM, Dunn-Coleman N, Ward M, Brettin TS. (2008): Genome sequencing and analysis of the biomass-degrading fungus *Trichoderma reesei* (syn. *Hypocrea jecorina*). In *Nature biotechnology* 26 (5), 553–560. DOI: 10.1038/nbt1403.

- Masuko T, Minami A, Iwasaki N, Majima T, Nishimura S, Lee YC. (2005): Carbohydrate analysis by a phenol-sulfuric acid method in microplate format. In *Analytical biochemistry* 339 (1), 69–72. DOI: 10.1016/j.ab.2004.12.001.
- Mba MF, Davies GJ, Drancourt M, Henrissat B. (2012): Genome analyses highlight the different biological roles of cellulases. In *Nature reviews. Microbiology* 10 (3), 227–234. DOI: 10.1038/nrmicro2729.
- McCarter JD, Withers SG. (1994): Mechanisms of enzymatic glycoside hydrolysis. In *Current opinion in structural biology* 4 (6), 885–892. DOI: 10.1016/0959-440X (94)90271-2.
- McGrath CE, Wilson DB. (2006): Characterization of a *Thermobifida fusca* beta-1,3glucanase (Lam81A) with a potential role in plant biomass degradation. In *Biochemistry* 45 (47), 14094–14100. DOI: 10.1021/bi061757r.
- Mercado JA, Barcelo M, Pliego C, Rey M, Caballero JL, Munoz-Blanco J, Ruano-Rosa D, Lopez-Herrera C, Los Santos B de, Romero-Munoz F, Pliego-Alfaro F. (2015):
 Expression of the beta-1,3-glucanase gene bgn13.1 from *Trichoderma harzianum* in strawberry increases tolerance to crown rot diseases but interferes with plant growth. In *Transgenic research* 24 (6), 979–989. DOI: 10.1007/s11248-015-9895-3.
- Mertz B, Gu X, Reilly PJ. (2009): Analysis of functional divergence within two structurally related glycoside hydrolase families. In *Biopolymers* 91 (6), 478–495. DOI: 10.1002/bip.21154.
- Monfils AK, Triemer RE, Bellairs EF. (2011): Characterization of paramylon morphological diversity in photosynthetic euglenoids (Euglenales, Euglenophyta). In *Phycologia* 50 (2), 156–169. DOI: 10.2216/09-112.1.
- Mouyna I, Sarfati J, Recco P, Fontaine T, Henrissatz B, Latge JP. (2002): Molecular characterization of a cell wall-associated beta (1-3) endoglucanase of *Aspergillus fumigatus*. In *Medical mycology* 40 (5), 455–464.
- Muntari B, Amid A, Mel M, Jami MS, Salleh HM. (2012): Recombinant bromelain production in *Escherichia coli*: process optimization in shake flask culture by response surface methodology. In *Applied and industrial microbiology express* 2, 12. DOI: 10.1186/2191-0855-2-12.
- Naik SN, Goud VV, Rout PK, Dalai AK. (2010): Production of first and second generation biofuels. In *Renewable and sustainable energy reviews* 14 (2), 578–597. DOI: 10.1016/j.rser.2009.10.003.
- Nakagawa YS, Kudo M, Loose JSM, Ishikawa T, Totani K, Eijsink VGH, Vaaje-Kolstad G. (2015): A small lytic polysaccharide monooxygenase from *Streptomyces griseus*

targeting alpha- and beta-chitin. In *Federation of european biochemical societies journal* 282 (6), 1065–1079. DOI: 10.1111/febs.13203.

- Nguyen THM, Vu VH. (2012): Bioethanol production from marine algae biomass: prospect and troubles. In *Journal of Vietnamese environment* 3 (1), 25-29. DOI: 10.13141/jve.vol3.no1.pp25-29.
- Nishimura T, Bignon C, Allouch J, Czjzek M, Darbon H, Watanabe T, Henrissat B.
 (2001): Streptomyces matensis laminaripentaose hydrolase is an 'inverting' β-1,3-glucanase. In Federation of european biochemical societies letters 499 (1), 187–190.
 DOI: 10.1016/S0014-5793(01)02551-0.
- Noraini MY, Ong HC, Badrul MJ, Chong WT. (2014): A review on potential enzymatic reaction for biofuel production from algae. In *Renewable and sustainable energy reviews* 39 (2014), 24-34. DOI: 10.1016/j.rser.2014.07.089.
- Ozimek P, Veenhuis M, Klei IJ. (2005): Alcohol oxidase: A complex peroxisomal, oligomeric flavoprotein. In *Federation of european biochemical societies yeast* research 5 (11), 975–983. DOI: 10.1016/j.femsyr.2005.06.005.
- Palumbo JD, Sullivan RF, Kobayashi DY. (2003): Molecular Characterization and Expression in Escherichia coli of Three -1,3-Glucanase Genes from Lysobacter enzymogenes Strain N4-7. In Journal of bacteriology 185 (15), 4362–4370. DOI: 10.1128/JB.185.15.4362-4370.2003.
- Papageorgiou AC, Li D. (2015): Expression, purification and crystallization of a family 55 β-1,3-glucanase from *Chaetomium thermophilum*. In *Structural biology communications* 71 (6), 680–683. DOI: 10.1107/S2053230X15006366.
- Patrick SM, Fazenda ML, McNeil B, Harvey LM. (2005): Heterologous protein production using the *Pichia pastoris* expression system. In *Yeast* 22 (4), 249–270. DOI: 10.1002/yea.1208.
- Patrick WM, Nakatani Y, Cutfield SM, Sharpe ML, Ramsay RJ, Cutfield JF. (2010): Carbohydrate binding sites in *Candida albicans* exo-β-1,3-glucanase and the role of the Phe-Phe 'clamp' at the active site entrance. In *Federation of european biochemical societies journal* 277 (21), 4549–4561. DOI: 10.1111/j.1742-4658.2010.07869.x.
- Paulova L, Hyka P, Branska B, Melzoch K, Kovar K. (2012): Use of a mixture of glucose and methanol as substrates for the production of recombinant trypsinogen in continuous cultures with *Pichia pastoris* Mut+. In *Journal of biotechnology* 157 (1), 180–188. DOI: 10.1016/j.jbiotec.2011.10.010.
- Phaff HJ. (1986): My life with yeasts. In *Annual review of microbiology* 40, 1–28. DOI: 10.1146/annurev.mi.40.100186.000245.
- Pitson SM, Seviour RJ, McDougall BM. (1993): Noncellulolytic fungal β-glucanases: Their physiology and regulation. In *Enzyme and microbial technology* 15 (3), 178–192. DOI: 10.1016/0141-0229(93)90136-P.

- Plotnikova TA, Selyakh IO, Kalebina TS, Kulaev IS. (2006): Bgl2p and Gas1p are the major glucan transferases forming the molecular ensemble of yeast cell wall. In *Biochemistry and biophysics* 409, 244–247.
- Pothiraj C, Kanmani P, Balaji P. (2006): Bioconversion of lignocellulose materials. In *Mycobiology* 34 (4), 159–165. DOI: 10.4489/MYCO.2006.34.4.159.
- Pourcq K, Schutter K, Callewaert N. (2010): Engineering of glycosylation in yeast and other fungi: current state and perspectives. In *Applied microbiology and biotechnology* 87 (5), 1617–1631. DOI: 10.1007/s00253-010-2721-1.
- Quinlan RJ, Sweeney MD, Leggio LL, Otten H, Poulsen J, Johansen KS, Krogh KBRM, Jorgensen CI, Tovborg M, Anthonsen A, Tryfona T, Walter CP, Dupree P, Xu F, Davies GJ, Walton PH. (2011): Insights into the oxidative degradation of cellulose by a copper metalloenzyme that exploits biomass components. In *Proceedings of the national academy of sciences of the united states of america* 108 (37), 15079– 15084. DOI: 10.1073/pnas.1105776108.
- Ramesh HPF, Tharanathan RN. (2003): Carbohydrates: the renewable raw materials of high biotechnological value. In *Critical reviews in biotechnology* 23 (2), 149–173. DOI: 10.1080/713609312.
- Reese ET, SIU RGH, Levison HS. (1950): The biological degradation of soluble cellulose derivatives and its relationship to the mechanism of cellulose hydrolysis. In *Journal* of bacteriology 59 (4), 485–497.
- Roa M, Blobel G. (1983): Biosynthesis of peroxisomal enzymes in the methylotrophic yeast *Hansenula polymorpha*. In *Proceedings of the national academy of sciences of the united states of america* 80 (22), 6872–6876.
- Roggenkamp R, Janowicz Z, Stanikowski B, Hollenberg CP. (1984): Biosynthesis and regulation of the peroxisomal methanol oxidase from the methylotrophic yeast *Hansenula polymorpha*. In *Molecular and general genetics* 194 (3), 489–493.
- Russo R, Barsanti L, Evangelista V, Frassanito AM, Longo V, Pucci L, Penno G, Gualtieri P. (2016): *Euglena gracilis* paramylon activates human lymphocytes by upregulating pro-inflammatory factors. In *Food science and nutrition. DOI:* 10.1002/fsn3.383.
- Sakai Y, Nakagawa T, Shimase M, Kato N. (1998): Regulation and physiological role of the DAS1 gene, encoding dihydroxyacetone synthase, in the methylotrophic yeast *Candida boidinii*. In *Journal of bacteriology* 180 (22), 5885–5890.
- Sakamoto Y, Nakade K, Konno N. (2011): Endo-β-1,3-glucanase GLU1, from the fruiting body of *Lentinula edodes*, belongs to a new glycoside hydrolase family. In *Applied and environmental microbiology* 77 (23), 8350–8354. DOI: 10.1128/AEM.05581-11.
- Sanchez C. (2009): Lignocellulosic residues: biodegradation and bioconversion by fungi. In *Biotechnology advances* 27 (2), 185–194. DOI: 10.1016/j.biotechadv.2008.11.001.
- Saravana PP, Wilson S, Santhiagu A. (2016): Optimization of codon usage of the envelope protein E2 gene from various genotypes of hepatitis C virus to predict the

expression level in *Pichia pastoris*. In *Genes and genomics* 38 (10), 977–984. DOI: 10.1007/s13258-016-0442-2.

- Schaechter M (2012): Eukaryotic microbes. In *Eukaryotic microbiology* 59 (6), 654. DOI: 10.1111/j.1550-7408.2012.00637.x
- Schutter K de, Lin YC, Tiels P, Hecke A, Glinka S, Weber-Lehmann J, Rouze P, Peer Y, Callewaert N. (2009): Genome sequence of the recombinant protein production host *Pichia pastoris*. In *Nature biotechnology* 27 (6), 561–566. DOI: 10.1038/nbt.1544.
- Schwarz T, Bartholmes P, Kaufmann M. (1995): Large-scale production of algal biomass for protein purification. In *Biotechnology and applied biochemistry* 22 (2), 179–190. DOI: 10.1111/j.1470-8744.1995.tb00345.x.
- Scorer CA, Clare JJ, McCombie WR, Romanos MA, Sreekrishna K. (1994): Rapid selection using G418 of high copy number transformants of *Pichia pastoris* for highlevel foreign gene expression. In *Biotechnology* 12 (2), 181–184.
- Shen S, Sulter G, Jeffries TW, Cregg JM. (1998): A strong nitrogen source-regulated promoter for controlled expression of foreign genes in the yeast *Pichia pastoris*. In *Gene* 216 (1), 93–102.
- Shen SH, Chretien P, Bastien L, Slilaty SN. (1991): Primary sequence of the glucanase gene from *Oerskovia xanthineolytica*. Expression and purification of the enzyme from *Escherichia coli*. In *Journal of biological chemistry* 266 (2), 1058–1063.
- Shrestha KL, Liu SW, Huang CP, Wu HM, Wang WC, Li YK. (2011): Characterization and identification of essential residues of the glycoside hydrolase family 64 laminaripentaose-producing-beta-1, 3-glucanase. In *Protein engineering, design and selection* 24 (8), 617–625. DOI: 10.1093/protein/gzr031.
- Sibirny AA, Ubiyvovk VM, Gonchar MV, Titorenko VI, Voronovsky AY, Kapultsevich YG, Bliznik KM. (1990): Reactions of direct formaldehyde oxidation to CO2 are nonessential for energy supply of yeast methylotrophic growth. In *Archives of microbiology* 154 (6), 566–575. DOI: 10.1007/BF00248838.
- Sinclair G, Choy FYM. (2002): Synonymous codon usage bias and the expression of human glucocerebrosidase in the methylotrophic yeast, *Pichia pastoris*. In *Protein expression and purification* 26 (1), 96–105.
- Sonck E, Stuyven E, Goddeeris B, Cox E. (2010): The effect of beta-glucans on porcine leukocytes. In *Veterinary immunology and immunopathology* 135 (3-4), 199–207. DOI: 10.1016/j.vetimm.2009.11.014.
- Sreekrishna K, Potenz RH, Cruze JA, McCombie WR, Parker KA, Nelles L, Mazzaferro PK, Holden KA, Harrison RG, Wood PJ. (1988): High level expression of heterologous proteins in methylotrophic yeast *Pichia pastoris*. In *Journal of basic microbiology* 28 (4), 265–278.
- Sun FF, Bai R, Yang H, Wang F, He J, Wang C, Tu M. (2016): Heterologous expression of codon optimized *Trichoderma reesei* Cel6A in *Pichia pastoris*. In *Enzyme and*

microbial technology 92, 107-116. DOI: 10.1016/j.enzmictec.2016.07.004.

- Takeda T, Nakano Y, Takahashi M, Konno N, Sakamoto Y, Arashida R, Marukawa Y, Yoshida E, Ishikawa T, Suzuki K. (2015): Identification and enzymatic characterization of an endo-1,3-beta-glucanase from *Euglena gracilis*. In *Phytochemistry* 116, 21–27. DOI: 10.1016/j.phytochem.2015.05.010.
- Takeyama H, Kanamaru A, Yoshino Y, Kakuta H, Kawamura Y, Matsunaga T. (1997): Production of antioxidant vitamins, β-carotene, vitamin C, and vitamin E, by two-step culture of *Euglena gracilis* Z. In *Biotechnology and bioengineering* 53 (2), 185–190. DOI: 10.1002/(SICI)1097-0290(19970120)53:2<185::AID-BIT8>3.0.CO;2-K.
- Tanghe M, Danneels B, Camattari A, Glieder A, Vandenberghe I, Devreese B, Stals I, Desmet T. (2015): Recombinant Expression of *Trichoderma reesei* Cel61A in *Pichia pastoris*: Optimizing Yield and N-terminal Processing. In *Molecular biotechnology* 57 (11-12), 1010–1017. DOI: 10.1007/s12033-015-9887-9.
- Thor D, Xiong S, Orazem CC, Kwan AC, Cregg JM, Cereghino JL, Cereghino GPL. (2005): Cloning and characterization of the *Pichia pastoris* MET2 gene as a selectable marker. In *Federation of european biochemical societies yeast research* 5 (10), 935–942. DOI: 10.1016/j.femsyr.2005.03.009.
- Torres M, Palomares O, Quiralte J, Pauli G, Rodríguez R, Villalba M, Wang W. (2015): An Enzymatically Active β-1,3-Glucanase from Ash Pollen with Allergenic Properties: A Particular Member in the Oleaceae Family. In *Public library of science ONE* 10 (7), e0133066. DOI: 10.1371/journal.pone.0133066.
- Tsai PW, Yang CY, Chang HT, Lan CY. (2011): Characterizing the role of cell-wall beta-1,3-exoglucanase Xog1p in *Candida albicans* adhesion by the human antimicrobial peptide LL-37. In *Public library of science ONE* 6 (6), e21394. DOI: 10.1371/journal.pone.0021394.
- Tschopp JF, Brust PF, Cregg JM, Stillman CA, Gingeras TR. (1987): Expression of the lacZ gene from two methanol-regulated promoters in *Pichia pastoris*. In *Nucleic acids research* 15 (9), 3859–3876.
- Umemoto N, Kakitani M, Iwamatsu A, Yoshikawa M, Yamaoka N, Ishida I. (1997): The structure and function of a soybean beta-glucan-elicitor-binding protein. In *Proceedings of the national academy of sciences of the united states of america* 94 (3), 1029–1034.
- Vaaje KG, Westereng B, Horn SJ, Liu Z, Zhai H, Sorlie M, Eijsink VGH. (2010): An oxidative enzyme boosting the enzymatic conversion of recalcitrant polysaccharides. In *Science* 330 (6001), 219–222. DOI: 10.1126/science.1192231.
- Vadhana AKP, Samuel P, Berin RM, Krishna J, Kamatchi K, Meenakshisundaram S. (2013): Improved secretion of *Candida antarctica* lipase B with its native signal peptide in *Pichia pastoris*. In *Enzyme and microbial technology* 52 (3), 177–183. DOI: 10.1016/j.enzmictec.2013.01.001.

- Van BJ, Vries RP. (2011): Fungal enzyme sets for plant polysaccharide degradation. In *Applied microbiology and biotechnology* 91 (6), 1477–1492. DOI: 10.1007/s00253-011-3473-2.
- Van K, Yurimoto H, Sakai Y, Veenhuis M. (2006): The significance of peroxisomes in methanol metabolism in methylotrophic yeast. In *Biochimica et biophysica acta* 1763 (12), 1453–1462. DOI: 10.1016/j.bbamcr.2006.07.016.
- Van VV, Beeson WT, Span EA, Farquhar ER, Marletta MA. (2014): A family of starchactive polysaccharide monooxygenases. In *Proceedings of the national academy of sciences of the united states of america* 111 (38), 13822–13827. DOI: 10.1073/pnas.1408090111.
- Varghese JN, Garrett TP, Colman PM, Chen L, Fincher GB. (1994): Three-dimensional structures of two plant beta-glucan endohydrolases with distinct substrate specificities. In *Proceedings of the national academy of sciences of the united states of america* 91 (7), 2785–2789.
- Viladot JL, Ramon E, Durany O, Planas A. (1998): Probing the mechanism of *Bacillus* 1,3-1,4-beta-D-glucan 4-glucanohydrolases by chemical rescue of inactive mutants at catalytically essential residues. In *Biochemistry* 37 (32), 11332–11342. DOI: 10.1021/bi980586q.
- Vogel K, Barber AA. (1968): Degradation of paramylon by *Euglena gracilis*. In *The Journal of protozoology* 15 (4), 657–662.
- Vuong TV, Wilson DB. (2010): Glycoside hydrolases: catalytic base/nucleophile diversity. In *Biotechnology and bioengineering* 107 (2), 195–205. DOI: 10.1002/bit.22838.
- Wang G, Liu Z, Lin R, Li E, Mao Z, Ling J, Yang Y, Yin WB, Xie B. (2016): Biosynthesis of Antibiotic Leucinostatins in Bio-control Fungus Purpureocillium lilacinum and Their Inhibition on Phytophthora Revealed by Genome Mining. In *Public library of science pathogens* 12 (7), e1005685. DOI: 10.1371/journal.ppat.1005685.
- Wang L, Behr SR, Newman RK, Newman C. (1997): Comparative cholesterol-lowering effects of barley β-glucan and barley oil in Golden Syrian hamsters. In *Nutrition research* 17 (1), 77–88. DOI: 10.1016/S0271-5317(96)00234-5.
- Waterham HR, Digan ME, Koutz PJ, Lair SV, Cregg JM. (1997): Isolation of the *Pichia pastoris* glyceraldehyde-3-phosphate dehydrogenase gene and regulation and use of its promoter. In *Gene* 186 (1), 37–44.
- Waters MG, Evans EA, Blobel G. (1988): Prepro-alpha-factor has a cleavable signal sequence. In *Journal of biological chemistry* 263 (13), 6209–6214.
- Weidner M, Taupp M, Hallam SJ. (2010): Expression of recombinant proteins in the methylotrophic yeast *Pichia pastoris*. In *Journal of visualized experiments* (36). DOI: 10.3791/1862.
- Westereng B, Cannella D, Wittrup AJ, Jorgensen H, Larsen AM, Eijsink VGH, Felby C. (2015): Enzymatic cellulose oxidation is linked to lignin by long-range electron

transfer. In Scientific reports 5, 18561. DOI: 10.1038/srep18561.

- Westereng B, Ishida T, Vaaje-Kolstad G, Wu M, Eijsink VGH, Igarashi K, Samejima M, Stahlberg J, Horn SJ, Sandgren M. (2011): The putative endoglucanase PcGH61D from *Phanerochaete chrysosporium* is a metal-dependent oxidative enzyme that cleaves cellulose. In *Public library of science ONE* 6 (11), e27807. DOI: 10.1371/journal.pone.0027807.
- Wojtkowiak A, Witek K, Hennig J, Jaskolski M. (2012): Two high-resolution structures of potato endo-1,3-β-glucanase reveal subdomain flexibility with implications for substrate binding. In *Biological crystallography* 68 (6), 713–723. DOI: 10.1107/S090744491200995X.
- Wu H, Li J, Dong D, Liu T, Zhang T, Zhang D, Liu W. (2009): Structure, mechanistic action, and essential residues of a GH-64 enzyme, laminaripentaose-producing beta-1,3-glucanase. In *Journal of biological chemistry* 284 (39), 26708–26715. DOI: 10.1074/jbc.M109.010983.
- Wu HM, Liu SW, Hsu MT, Hung CL, Lai CC, Cheng WC, Wang HJ, Li YK, Wang WC. (2015): Heterologous coexpression of *Vitreoscilla hemoglobin* and *Bacillus megaterium* glucanase in *Streptomyces lydicus* A02 enhanced its production of antifungal metabolites. In *Enzyme and microbial technology* 81, 80–87. DOI: 10.1016/j.enzmictec.2015.08.003.
- Wu M, Beckham GT, Larsson AM, Ishida T, Kim S, Payne CM, Himmel ME, Crowley MF, Horn SJ, Westereng B, Igarashi K, Samejima M, Stahlberg J, Eijsink VGH, Sandgren M. (2013): Crystal structure and computational characterization of the lytic polysaccharide monooxygenase GH61D from the Basidiomycota fungus *Phanerochaete chrysosporium*. In *Journal of biological chemistry* 288 (18), 12828– 12839. DOI: 10.1074/jbc.M113.459396.
- Wu S, Letchworth GJ. (2004): High efficiency transformation by electroporation of *Pichia pastoris* pretreated with lithium acetate and dithiothreitol. In *Biotechniques* 36 (1), 152–154.
- Xu Z, Shih MC, Poulton JE. (2006): An extracellular exo-beta-(1,3)-glucanase from *Pichia pastoris*: purification, characterization, molecular cloning, and functional expression. In *Protein expression and purification* 47 (1), 118–127. DOI: 10.1016/j.pep.2005.11.025.
- Yamamoto R, Nevins DJ. (1983): Degradation of a glucan containing β -(1,3) and β -(1,6) linkages by exo-(1,3)- β -d-glucanase. In *Carbohydrate research* 122 (2), 217–226. DOI: 10.1016/0008-6215(83)88333-5.
- Yurimoto H, Kato N, Sakai Y. (2005): Assimilation, dissimilation, and detoxification of formaldehyde, a central metabolic intermediate of methylotrophic metabolism. In *Chemical Record* 5 (6), 367–375. DOI: 10.1002/tcr.20056.

Yurimoto H, Oku M, Sakai Y. (2011): Yeast methylotrophy: metabolism, gene regulation

and peroxisome homeostasis. In *International journal of microbiology* 2011, e101298. DOI: 10.1155/2011/101298.

- Zhang HU, Shi WL, You JF, Di BM, Qin XM, Yu HU, Liu Q, Ryan PR, Yang ZM. (2015): Transgenic arabidopsis thaliana plants expressing a β -1,3-glucanase from sweet sorghum (*Sorghum bicolor* L.) show reduced callose deposition and increased tolerance to aluminium toxicity. In *Plant cell and environment* 38 (6), 1178–1188. DOI: 10.1111/pce.12472.
- Zhou P, Chen Z, Yan Q, Yang S, Hilgenfeld R, Jiang Z. (2013): The structure of a glycoside hydrolase family 81 endo-β-1,3-glucanase. In *Biological crystallography* 69 (10), 2027–2038. DOI: 10.1107/S090744491301799X.
- Zumstein H. (1900): Zur Morphologie und Physiologie der Euglena gracilis Klebs. In Jahrbücher für Wissenschaftliche Botanik 34, 149–198.

Appendices

Solutions and mediums

Tris-acetate-EDTA (TAE) stock solution (50×)		
Component	Concentration	
Tris base	242.0 g/L	
Glacial acetate acid	5.71% (v/v)	
EDTA	9.3 g/L	

Tris-glycine stock solution	(5×))
-----------------------------	------	---

Component	Concentration
Tris base	15.1 g/L
Glycine	94.0 g/L
SDS	50.0 g/L

TSS stock solution	(2×)
--------------------	------

Component	Concentration
PEG 8000	100 g/L
MgCl ₂ ×6H ₂ O	6 g/L
DMSO	1% (v/v)

Component	Concentration (1L)
Methanol	450 mL
Glacial acetic acid	100 mL
Coomassie brilliant blue G-250	3.0 g
Deionized H ₂ O	450 mL

Coomassie brilliant blue G-250 staining buffer

Silver staining fixer

Component	Concentration (1L)
Ethanol	400 mL
Acetic acid	100 mL
Deionized H ₂ O	500 mL

Silver staining sensitizing solution	
Component	Concentration (1L)
$Na_2S_2O_3$	0.2 g
Deionized H ₂ O	1 L

Silver nitrate solution	
Component	Concentration (1L)
AgNO ₃	1 g
Deionized H ₂ O	1 L
35% Formaldehyde	200 µL add before use

Silver staining visualizing solution

Component	Concentration (1L)
Na ₂ CO ₃	30 g
Deionized H ₂ O	1 L
35% Formaldehyde	500 µL add before use

Silver staining stop solution

Component	Concentration (1L)
Acetic acid	50 mL
Deionized H ₂ O	950 mL

LB medium	
Component	Concentration (1L)
Tryptone	10 g
Yeast extract	5 g
NaCl	5 g
Deionized H ₂ O	Add to 1 L

pH=7.4

YPD medium	
Component	Concentration (1L)
Peptone	20 g
Yeast extract	10 g
400 g/L glucose (sterile by 0.22 μm filter)	50 mL
Deionized H ₂ O	Add to 1 L

pH=6.5

Component	Concentration (1L)
Peptone	20 g
Yeast extract	10 g
1.0 mol/L Potassium phosphate buffer (pH 6.0)	50 mL
10× Yeast nitrogen base (YNB)	100 mL
500× Biotin	2 mL
10× Glycerol	100 mL for BMGY
10× Methanol	100 mL for BMMY
Deionized H ₂ O	Add to 1 L

BMGY and BMMY medium

pH=6.0

Declaration

I hereby declare that this thesis is a presentation of my original research work. Wherever contributions of other are involved, every effort is made to indicate this clearly, with due reference to the literature. Care has been taken to give credit to all sources that have given information in the form of images, discussions or materials.

Bielefeld,

.....

Yingfei Shi